

Broadband Continuous-time MASH Sigma-Delta ADCs

Citation for published version (APA): Zhang, C. (2021). *Broadband Continuous-time MASH Sigma-Delta ADCs*. [Phd Thesis 1 (Research TU/e / Graduation TU/e), Electrical Engineering]. Eindhoven University of Technology.

Document status and date: Published: 21/09/2021

Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

• The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.

Broadband Continuous-Time MASH Sigma-Delta ADCs

Chenming Zhang



Broadband Continuous-Time MASH Sigma-Delta ADCs

Chenming Zhang

This work was supported by the Dutch Technology Foundation (STW) under project 12433.

Cover design by Ms. Mercedes Benjaminse (Proefschriftmaken) and Chenming Zhang. The ownership of the right-bottom figure about the automotive radar belongs to En-Silica (www.ensilica.com), and a written permission for the usage has been acquired.

Zhang, C. Broadband Continuous-Time MASH Sigma-Delta ADCs Proefschrift Technische Universiteit Eindhoven, 2021. A catalogue record is available from the Eindhoven University of Technology Library ISBN: 978-90-386-5322-8 NUR 959 Trefw.: analog-to-digital converters, Sigma-Delta ADC, calibration, excess loop delay compensation, metastability.

> ©C. Zhang 2021 All rights reserved. Reproduction in whole or in part is prohibited without the written consent of the copyright owner.

Broadband Continuous-Time MASH Sigma-Delta ADCs

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Technische Universiteit Eindhoven, op gezag van de rector magnificus, prof.dr.ir. F.P.T. Baaijens, voor een commissie aangewezen door het College voor Promoties in het openbaar te verdedigen op dinsdag 21 september 2021 om 11.00 uur

door

Chenming Zhang

geboren te Yongqing, Hebei, China

Dit proefschrift is goedgekeurd door de promotoren en de samenstelling van de promotiecommissie is als volgt:

voorzitter:	Prof.dr.ir. S.M. Heemstra de Groot
1^e promotor:	Prof.dr.ir. L.J. Breems
2^e promotor:	Prof.dr.ir. A.H.M. van Roermund
Copromotor:	Dr. G.I. Radulov, PDEng., MSc.
leden:	Prof.DrIng. M. Ortmanns (Ulm University)
	Prof.dr. G.G.E. Gielen (KU Leuven)
	Prof.dr.ir. E. Cantatore
	Dr.ir. M. Bolatkale (Technische Universiteit Delft)

Het onderzoek of ontwerp dat in dit proefschrift wordt beschreven is uitgevoerd in overeenstemming met de TU/e Gedragscode Wetenschapsbeoefening.

Contents

	List	of symbols and abbreviations	9
1	Inti	roduction	13
	1.1	Background	13
	1.2	Problem definition	14
	1.3	Aim of the thesis	15
	1.4	Scope of the thesis	15
	1.5	Original contributions	16
	1.6	Thesis outline	16
Ι	Aı	chitecture: analysis, limitations and design aspects	19
2	\mathbf{CT}	$\Sigma\Delta$ analog-to-digital conversion for broadband applications	21
	2.1	Broadband applications	21
	2.2	Receiver architecture and target specifications for the ADC	24
	2.3	ADC overview	26
	2.4	CT $\Sigma\Delta$ ADCs compared to Nyquist ADCs	30
	2.5	Increasing the bandwidth of CT $\Sigma\Delta$ ADCs	32
	2.6	Conclusions	33
3	Arc	hitecture choices for broadband CT $\Sigma\Delta$ ADCs	35
	3.1	CT SDM general description and definition	35
	3.2	SDM parameter design choices and related loop properties $\ . \ . \ .$	38
	3.3	SDM topology design choices	43

	3.4	Benchmarking different $\Sigma\Delta$ ADCs	46
	3.5	Design-space reduction	49
	3.6	Conclusions	50
4	Cha	llenges of broadband CT $\Sigma\Delta$ ADCs	51
	4.1	Loop filter non-idealities	51
	4.2	Quantizer non-idealities	55
	4.3	DAC non-idealities	58
	4.4	Clock non-idealities	61
	4.5	System-level non-idealities	63
	4.6	Supply and reference non-idealities	64
	4.7	Limitations from CMOS technologies	65
	4.8	Design-space reduction	66
	4.9	Conclusions	68
5	Syn	thesis procedure for broadband CT $\Sigma\Delta$ ADC architectures	69
	5.1	Choice of main design parameters	69
	5.2	A generic synthesis flow for broadband CT MASH SDMs $\ . \ . \ . \ .$	77
	5.3	System-level optimization	82
	5.4	System-level verification and block-level specifications	82
	5.5	Conclusions	87
II	N	ovel design concepts	89
6	A d	igital calibration technique for broadband CT MASH $\Sigma\Delta$ ADCs	91
	6.1	Introduction	91
	6.2	Modelling of the loop filter non-idealities	93
	6.3	Mismatch between analog filter and digital NCF	94
	6.4	Analog RC calibration	95
	6.5	Digital calibration algorithm	95
	6.6	Simulation results	98
	6.7	Conclusions	102

7	Higl	n-speed ELD compensation	103
	7.1	Overview for high-speed ELD compensation techniques $\ . \ . \ . \ .$	103
	7.2	Current-mode multi-path ELD compensation	110
	7.3	Overcoming parasitic capacitance on summation node	118
	7.4	Conclusions	124
8	Con	parators for broadband CT $\Sigma\Delta$ ADCs	125
	8.1	Current-mode locally-time-interleaved multi-bit quantizer	125
	8.2	Comparator offset calibration	137
	8.3	Conclusions	149
9	DAG	C linearization techniques	151
	9.1	Introduction	151
	9.2	State-of-the-art implementations of DAC linearization techniques $~$	152
	9.3	Proposed implementation of DAC linearization techniques $\ \ . \ . \ .$	154
	9.4	Circuit implementation	159
	9.5	Simulation results	160
	9.6	Conclusions	162
10	Ove	rcoming metastability error limitations	165
	10.1	Introduction	165
	10.2	State of the art in metastability error modeling and mitigation	166
	10.3	Proposed modeling of metastability error mechanism $\ldots \ldots \ldots$	168
	10.4	Metastability error compensation	172
	10.5	Metastability shaping technique	185
	10.6	Conclusions	190
II	ΙΊ	Test chip design and validation	193
11	A 6	GHz 500 MHz BW 2-1-1 MASH CT $\Sigma\Delta$ ADC	195
	11.1	Circuit-level implementation	195

		017
	11.2 Simulation results	217
	11.3 Test chip validation	221
	11.4 Benchmarking	229
	11.5 Conclusions	233
	~	
12	Conclusions and recommendations	237
	12.1 Conclusions	237
	12.2 Recommendations for future work	239
	References	243
	List of publications	255
	Summary	257
	Samenvatting	259
	Acknowledgements	261
	A control to the first state of	201
	Biography	263

List of symbols and abbreviations

Symbol Description

Unit

3GPP	3rd Generation Partnership Project	
ADAS	advanced driver assistance systems	
ADC	Analog to digital converter	
BLMS	Block Least Mean Square	
BP	Band-pass	
BW	Bandwidth	
CDMA	Code-division multiple access	
CMFB	Common-mode feedback	
DEM	Dynamic element matching	
CIFF	Cascade-of-integrators, feedforward form	
CMFB	Common-mode feedback	
CT	Continuous-time	
CRFB	Cascade-of-resonators, feedback form	
CRFF	Cascade-of-resonators, feedforward form	
CLF	Connecting loop filters	
DAB	Digital audio broadcasting	
DAC	Digital to analog converter	
dBFS	dB full-scale	
DEM	dynamic element matching	
D-FF	D-flipflop	
DR	dynamic range	
DT	Discrete-time	
DWA	Data weighted averaging	
EDGE	Enhanced data rates for GSM evolution	
ENOB	Effective number of bits	bit
ELD	Excess loop delay	
ESD	electrostatic discharge	
FIR	Finite impulse response	
FM	Frequency modulation	
FMCW	Frequency-modulated continuous wave	
FoM	Figure of merit	
FoM_W	Waldon figure of merit	J/conversion-step

FoM_S	Schreier figure of merit	dB
F_s	Sampling frequency	Hz
f_T	Transit frequency of MOS transistors	
GBW	Gain bandwidth product	Hz
GPRS	General packet radio service	
GSM	Global system for mobile communications	
GRO	Gated-ring-oscillator	
HD2	2^{nd} order harmonic	
HD3	3^{rd} order harmonic	
IBN	In-band noise	
IC	Integrated circuit	
LDE	Layout-dependent effects	
IF	Intermediate frequency	
ILF	Internal loop filter	
IMD_2	2^{nd} order inter-modulation-distortion	
IMD_3	3^{rd} order inter-modulation-distortion	
INL	Integral non-linearity	
ISG	Inter-stage gain	
LDO	Low-dropout	
LP	Low-pass	
LSB	Least significant bit	
LTE	Long term evolution	
MASH	Multi-stAge noise SHaping	
MC	Monte-carlo	
MEC	metastability error compensation	
MOS	Metal-oxide-silicon	
MSA	Maximum stable input amplitude	
NCF	Noise cancellation filter	
NRZ	Non-return-to-zero	
NTF	Noise transfer function	
OBG	Out-of-band gain	
OSR	Over-sampling ratio	
OTA	Operational transconductance amplifier	
\mathbf{PC}	Personal computers	
PLL	phase-locked loop	
PVT	Process, voltage and temperature	
PSRR	power supply rejection ratio	
RZ	Return-to-zero	
SAR	Successive approximation register	
SC	Switched-capacitor	
SDM	Sigma-Delta modulator	
SFDR	Spurious free dynamic range	
S&H	sample-and-hold	
SMASH	Sturdy MASH	
SNDR	Signal to noise and distortion ratio	

SNR	Signal to noise ratio	
SoC	System on chip	
SQNR	Signal to quantization noise ratio	
STF	Signal transfer function	
TDC	Time-to-digital converter	
THD	Total harmonic distortion	
TI	time-interleaved	
UGBW	Unity-gain bandwidth	Hz
UMTS	Universal mobile telecommunications system	
UWB	Ultra-wide-band	
V2X	Vehicle-to-everything	
V_{CM}	Common-mode voltage	V
VCO	Voltage-controlled oscillator	

Chapter 1

Introduction

This chapter introduces the research context of the thesis. First, the background of this research is given, which is further elaborated in more detail in Chapter 2. Afterwards, the problem definition is described. Moreover, this chapter presents the aim and the scope of this thesis. Then the original contributions are listed. In the end, the outline of this thesis is shown.

1.1 Background

The development of the semiconductor industry has been making our lives safer, smarter, easier, more productive and more efficient, by enabling numerous new products and applications. For example, high-speed networks (e.g. 5G) and smart phones have changed our lifestyle, and we can work, study, go shopping, play games, etc. with the same smart phones. Today's smart phones have more computation power than the computers that navigated the landing on the moon in 1960's. Electrification and automation become the main trends of today's automotive industry, which cannot happen without the advanced driver assistance systems (ADAS), power management, microcontrollers, etc. provided by the semiconductor industry. Breathing machines (respirators), CT scanners, etc. have saved millions of lives in the on-going Covid-19 crisis. Personal computers (PC) and tablets enable working from home and remote studying. All of the aforementioned products, services, applications and many more cannot come true without the development and innovation of the semiconductor industry. That's why although the world's economy is heavily impacted by the Covid-19 crisis, the demand for the semiconductor products (chips) is even higher.

Analog-to-digital converters (ADCs) are essential building blocks in many semiconductor products. The basic function of an ADC is to convert an analog signal into a digital signal. Light (image), sound, electromagnetic waves, etc. in nature are all analog signals. In modern electrical devices, the data are represented, processed, computed and stored (in most cases) in the format of digital signals. Without ADCs (and digital-to-analog converters (DACs)), analog signals and digital signals are two separated domains. ADCs (and DACs) are the bridges to connect analog domain with digital domain, and hence they are widely used in many application categories of semiconductor products.

 $\Sigma\Delta$ ADCs make use of the available bandwidth (BW) offered by integrated circuit (IC) technologies, and trade time resolution for amplitude resolution in an efficient way by noise shaping. Continuous-time (CT) $\Sigma\Delta$ ADCs are widely used in telecommunication, automotive, consumer electronics, healthcare and other applications. Compared to other types of ADCs which have an input sample-and-hold (S&H), CT $\Sigma\Delta$ ADCs have advantages of resistive input, inherent anti-aliasing filtering and easy integration. Broadband CT $\Sigma\Delta$ ADCs are integrated in many telecommunication and automotive products, e.g. 5G base station transceivers [1], and car radar [2]. Since future applications require more data rate and higher performance, CT $\Sigma\Delta$ ADCs with broader signal bandwidth are needed [3]. Indeed, the achievable signal BW scales up with the CMOS technology nodes, and it is of great interest to explore the BW boundary of CT $\Sigma\Delta$ ADCs for a given CMOS technology and a certain resolution target.

1.2 Problem definition

 $CT \Sigma\Delta$ ADCs are widely used for medium-to-high resolution (> 50 dB signal to noise and distortion ratio (SNDR)) and low-to-medium bandwidth (≤ 100 MHz) ADC applications. For medium resolution applications (50 dB \leq SNDR \leq 70 dB), for the same SNDR, the reported broadest signal bandwidth of CT $\Sigma\Delta$ ADCs is less than the reported broadest signal bandwidth of pipelined ADCs and time-interleaved (TI) successive approximation register (SAR) ADCs [4] (see Chapter 2.3). CT $\Sigma\Delta$ ADCs have advantages in the system-level integration, however, the problem which motivates this research is that the current maximum achievable signal BW of CT $\Sigma\Delta$ ADCs is lower than the demanded signal BW of future automotive and telecommunication applications, e.g. car radar, base station. For CT $\Sigma\Delta$ ADCs, the signal BW is limited by the maximal achievable sampling rate and the over-sampling ratio (OSR). The minimum OSR is bounded by the order of the noise transfer function (NTF), quantizer resolution and the signal to quantization noise ratio (SQNR) target. The maximal achievable sampling rate is limited by the stability of the $\Sigma\Delta$ modulators (SDMs), the quantization gain and metastability error, and the maximum speed offered by the technology. The noise and linearity targets have impact on the maximal achievable sampling rate as well. In the end, the signal BW and resolution of the CT $\Sigma\Delta$ ADCs are bounded by power consumption and area.

The research questions that this thesis addresses are the following: How to maximize the achievable BW of a CT SDM in a given CMOS technology for a given DR? More specifically, what is the maximum achievable signal bandwidth of CT SDM in 40nm CMOS technology for a DR target of $60 \sim 70$ dB and a total power consumption of

less than 1 W? What SDM architectures should be chosen and what design techniques should be used? Are there new techniques that can be used to further increase the signal bandwidth? What are the fundamental limitations for the maximal achievable signal bandwidth for CT $\Sigma\Delta$ ADCs?

1.3 Aim of the thesis

The aim of this thesis is to explore the maximum achievable bandwidth of CT $\Sigma\Delta$ ADCs for a given CMOS technology and a certain resolution target. The motivation of this thesis is to bring the advantages of CT $\Sigma\Delta$ ADCs to a broader bandwidth than the state-of-the-art. More specifically, the target of this thesis is to explore the maximum achievable bandwidth of CT $\Sigma\Delta$ ADCs in a 40 nm CMOS technology for a dynamic range (DR) of 60 ~ 70 dB and total power consumption of less than 1 W. The aim of this thesis is further elaborated as follows:

- Explore architecture choices and their implementation limitations for broadband CT $\Sigma\Delta$ ADCs. Research on the maximal SQNR potential of single-loop and Multi-stAge noise SHaping (MASH) $\Sigma\Delta$ ADC architectures, and synthesize a CT $\Sigma\Delta$ ADC architecture for test chip implementation.
- Investigate the challenges in the circuit implementation of broadband CT $\Sigma\Delta$ ADCs. Propose novel concepts for system-level and circuit-level solutions to combat these challenges.
- Design a broadband CT $\Sigma\Delta$ ADC test chip, validate the study of this thesis and prove the proposed concepts.

1.4 Scope of the thesis

The scope of this thesis is elaborated as follows:

- Since the motivation of this thesis is to bring the advantages of CT $\Sigma\Delta$ ADCs to applications with a broader bandwidth, other types of ADCs are not investigated in detail. This thesis focuses on low-pass (LP) CT $\Sigma\Delta$ ADCs, and hence band-pass (BP) CT $\Sigma\Delta$ ADCs are not investigated in detail.
- This thesis focuses on the core circuity of the ΣΔ modulator. Supply regulation circuits, such as low-dropout (LDO) regulators, are out of the scope of this thesis. The hardware implementation of some digital circuitry, such as decimation filter and noise cancellation filter, is out of the scope of this thesis.

• The target applications of the broadband CT $\Sigma\Delta$ ADCs in this thesis are future automotive and telecommunication applications, e.g. car radar, base station. For these applications, system on chip (SoC) is the main trend for transceiver products. To benefit for the system integration, CMOS technology is considered in this thesis. The ADC test chip design is performed in a 40 nm CMOS technology in this thesis.

1.5 Original contributions

The original contributions of this thesis are as follows:

- Generic formulas to calculate the SQNR limits for both single-loop and MASH SDMs (Chapter 5.1)
- A digital calibration of the errors due to the limited DC gain and 2nd pole of the loop filter integrators for MASH SDMs (Chapter 6).
- A current-mode multi-path excess loop delay (ELD) compensation technique (Chapter 7.2).
- Techniques overcoming the parasitic capacitance on the summation node (Chapter 7.3).
- A current-mode locally-time-interleaved multi-bit quantizer (Chapter 8.1).
- A comparator offset calibration technique (Chapter 8.2).
- A new implementation of the DAC linearization techniques (Chapter 9).
- A metastability error compensation technique (Chapter 10.4).
- A metastability shaping technique (Chapter 10.5).
- A 6-GHz-sampling 3-bit 2-1-1 MASH CT SDM test chip with the largest signal BW reported so far of 500 MHz. The MASH ADC test chip achieves 58 dB DR in 500 MHz BW when sampled at 6 GHz and it consumes 574 mW. If sampled at 4 GHz, it achieves 65 dB DR in 300 MHz BW and it consumes 506 mW.

1.6 Thesis outline

This thesis is divided into three parts and 12 chapters. The outline of this thesis is described below.

- Part I discusses the analysis, limitations and design aspects in the $\Sigma\Delta$ ADC architecture design. It is further divided into 4 chapters.
- Chapter 2 elaborates on the potential of CT $\Sigma\Delta$ ADCs for broadband applications.
- Chapter 3 presents the architecture design choices for broadband CT $\Sigma\Delta$ ADCs. It is part of the top-down approach of the broadband CT $\Sigma\Delta$ ADC design.
- Chapter 4 discusses the challenges, non-idealities and limitations of the circuit implementation of broadband CT $\Sigma\Delta$ ADCs. It is part of the bottom-up approach of the broadband CT $\Sigma\Delta$ ADC design.
- Chapter 5 presents a synthesis flow for broadband CT MASH $\Sigma\Delta$ modulators, on the basis of an exemplary ADC architecture for the target application defined earlier.
- **Part II** presents several novel design concepts proposed in this thesis. It is further divided into 5 chapters.
- Chapter 6 proposes a digital automatic calibration method for broadband CT MASH $\Sigma\Delta$ ADCs. The main contribution of this method is the calibration of the errors due to the limited DC gain and 2nd pole of the loop filter integrators.
- Chapter 7 discusses high-speed ELD compensation techniques for broadband CT $\Sigma\Delta$ ADCs. This chapter proposes a current-mode multi-path ELD compensation technique and techniques to overcome the parasitic capacitance on the summation node.
- **Chapter 8** presents a current-mode locally-time-interleaved multi-bit quantizer and a comparator offset calibration technique.
- Chapter 9 proposes a new implementation of the DAC linearization techniques in broadband CT $\Sigma\Delta$ ADCs with the mapping engine in the digital domain.
- Chapter 10 studies the impact of metastability errors in broadband CT $\Sigma\Delta$ ADCs. This chapter proposes two system-level solutions to combat the metastability errors a metastability error compensation technique, and a metastability shaping technique.
- Part III presents the CT MASH $\Sigma\Delta$ ADC test chip design and validation. It contains 1 chapter.
- Chapter 11 presents the circuit-level implementation, simulation and validation results of a 6-GHz-sampling 500-MHz-BW 2-1-1 MASH CT $\Sigma\Delta$ ADC test chip.
- **Chapter 12** presents the conclusions of this thesis and recommendations for future work.

Part I

Architecture: analysis, limitations and design aspects

Chapter 2

${\bf CT} \ \Sigma \Delta \ {\bf analog-to-digital} \ {\bf conversion} \\ {\bf for \ broadband} \ {\bf applications}$

This chapter elaborates on the potential of $CT \Sigma \Delta$ ADCs for broadband applications. Firstly, the evolution of mobile telecommunication standards shows the needs of increasing signal bandwidth. ADCs with broader bandwidth benefit from the speed advantage offered by more advanced CMOS technologies. The chapter further presents a typical receiver architecture and gives target ADC specifications for such broadband applications. Afterwards, the chapter depicts an overview of different ADC architectures. Low-pass $CT \Sigma \Delta$ ADCs are further compared to Nyquist ADCs in the context of a typical receiver system. In the end, the motivation of this thesis is presented, and conclusions are drawn.

2.1 Broadband applications

We have been witnessing and experiencing the expeditious development of new technologies. One example is our mobile phones over the last two decades – higher download/upload speed, more Apps, more advanced functions. Figure 2.1 shows the evolution of the mobile telecommunication standards. The 1st generation of the mobile telecommunication standards uses analog radio signals, which was launched for commercial use starting from 1979 and it can offer a maximal user's download speed of about 2.4 kbit/s. The first commercial 2G network based on GSM standard was launched in 1991, which increased the user's download speed to 35 kbit/s. Starting from 2G, digital radio signals are used in the transmission. Two standards are used worldwide as the 3G mobile telecommunication standards – UMTS and CDMA2000. They were first offered in 2001 and 2002, and the maximal user's download speed are 384 kbit/s and 2 Mbit/s, respectively. 4G standard LTE-Advanced started in commercial use around 2010, and can offer 100 Mbit/s download speed. Starting from 2019, 5G networks were launched whose download speed is up to 1.8 Gbit/s. Figure



Figure 2.1: Mobile telecommunication standards evolution: user's download speed vs. year (based on [6]).

2.1 shows that the user's download speed increased exponentially in the last 30 years. If this trend continues, the next generation (6G) mobile telecommunication standard will appear around 2028 and it will offer a user's download speed of more than 30 Gbit/s. The exponential increasing trend resembles Moore's law [5].

The increasing download/upload speed is enabled by the breakthroughs in communication theory, hardware and software. According to the Shannon theory, the channel capacity C is defined by the signal BW, signal power P_S and noise power P_N [7]:

$$C = BW \cdot log_2 \frac{P_S + P_N}{P_N} \quad [\text{bit/s}]$$
(2.1)

To increase the channel capacity, the signal BW and/or signal to noise ratio (SNR) should be increased.

$$SNR = \frac{P_S}{P_N} \tag{2.2}$$

When the signal power is much larger than the noise power, the channel capacity is linearly proportional to the BW, while logarithmically proportional to the SNR. In more advanced CMOS technology with lower minimum transistor gate length, the supply voltage for the thin-oxide transistors becomes lower, while the maximum speed offered by the technology is higher. As the signal amplitude is limited by the supply voltage, the signal power P_S is lower. If the noise power P_N is limited by the thermal noise, the power consumption of the circuit increases by a factor of 4 to reduce the noise by 6 dB [8]. Thus in more advanced CMOS technology, it is usually more power efficient to increase the BW which benefits from the higher speed offered by the technology, to achieve a higher channel capacity C. This trend can be observed



Figure 2.2: Historical overview of the signal BW vs. SNDR plots of the state-of-theart ADCs with technology information (a) till 2005 (b) from 2006 till 2010 (c) from 2011 till 2015 (d) from 2016 till 2020 (based on [4]).

from Figure 2.2 as well – over the last 15 years, mainly the signal BW of the state-of-the-art ADCs increases, while their signal to noise and distortion ratio stays roughly in the same range. That's why the signal BW in the evolution of many applications is becoming broader and broader.

Actually there is no strict definition about broadband applications and narrow-band applications. They are relative. Firstly, broadband and narrow-band approaches can be defined based on the applications. For example, in the automotive frequency modulation (FM) radio applications, the narrow-band approach is to digitize only one or a few channels, and the corresponding signal bandwidth of the ADC is about a few hundred of kHz. Instead, a broadband approach is to digitize the whole band (e.g. FM and/or digital audio broadcasting (DAB) band), and the corresponding signal bandwidth of the ADC is a few tens of MHz [9].

Secondly, broadband and narrow-band should be defined according to the technology and the state-of-the-art at that time. Figure 2.2 shows the signal BW vs. SNDR plots of the state-of-the-art ADCs within four different time frames with technology information. Till the end of 2005, only a few published ADCs achieved an equivalent sampling jitter error better or equal to 1 ps [10], and most of the ADCs were implemented in > 100 nm CMOS technology. And for medium-to-high resolution ADCs with SNDR \geq 60 dB, the broadest signal BW is 90 MHz [11]. From 2006 till 2010, more ADCs outperformed 1 ps equivalent sampling jitter error, and about half of the ADCs were implemented in 55 - 90 nm CMOS technology. From 2011 till 2015, the best published ADC achieved an equivalent sampling jitter error very close to 0.1 ps [12], and 22 - 45 nm CMOS technology became popular for broadband ADCs. From 2016 till 2020, several published state-of-the-art ADCs already achieve an equivalent sampling jitter error better than 0.1 ps [13], and most of them were implemented in 7 - 45 nm CMOS technology. Moreover, for medium-to-high resolution ADCs with SNDR \geq 60 dB, the broadest signal BW till now is 1.6 GHz [14]. Figure 2.2 shows that state-of-the-art ADCs have been moving towards broader bandwidth in the last two decades, which benefits from higher speed offered by more advanced CMOS technology.

Broadband approaches have many advantages compared to narrow-band approaches. Firstly, the available timing resolution offered by the technology is used. Moreover, the channel selection happens in the digital domain, which benefits from all the advantages of digital processing including perfect matching, low noise, power efficiency and friendly scaling over the technologies.

On the other hand, broadband approaches have some disadvantages. Firstly, speed (or bandwidth) limitations of the building blocks and the integrity of the clock often put limitations on the broadband systems. Moreover, the influence of the interconnect loading and crosstalk can become significant, and the RC delay of the interconnect adds up in the total delay. Last but not least, the in-band and out-of-band interference often places stringent requirements on the dynamic range and linearity of the broadband systems. The challenges for broadband CT $\Sigma\Delta$ ADCs will be studied in detail in Chapter 4.

2.2 Receiver architecture and target specifications for the ADC

In communication systems, a receiver captures the signal when it has passed through the communication channel [15]. The receiver is an essential part in many broadband application systems, e.g. base station for telecommunication, automotive radar, automotive radio, etc [16, 2, 9]. Figure 2.3 shows a simplified functional block diagram of a typical direct-conversion receiver with ADCs [15]. After mixing down and amplified by the baseband filter, the IF signal is filtered. The frequency components at the aliasing frequency need to be suppressed by the anti-aliasing filter, and the out-of-band interference needs to be reduced by the interference filter. After that, an input buffer is used to drive the ADC. After the ADC, a decimation filter is generally needed to remove the out-of-band noise and interference in the digital domain.



Figure 2.3: Simplified block diagram of a direct-conversion receiver with ADCs [15].

In the recently launched 5G base station transceiver SoC products, ADCs with signal BW 100 - 200 MHz and SNDR 60 - 65 dB are needed in the receivers [1]. This thesis studies the generic design aspects of broadband ADCs on the basis of a target application domain, e.g. future communication and automotive applications. However, the novel concepts and techniques described in Part II are generic, and can be applied to many other broadband applications. In defining the target specifications, not only the application requirements have to be considered, but also what is practically achievable in the technology chosen for the implementation. In this thesis, the broadband ADC is designed in a 40 nm CMOS technology. Considering the speed limitation of this technology, the primary design target for this ADC is chosen to be signal bandwidth of 500 MHz, SNDR ≥ 60 dB and dynamic range ≥ 63 dB. Further practical system, circuit and technology limitations will be addressed in the following chapters.

Power efficiency is another important criterion. First of all, in some battery-powered (or even cell-battery powered) broadband applications, such as UWB car key applications, the power budget for the ADCs is very limited [17]. On the other hand, even for other broadband applications for which the supply of the power is not such a big concern, such as base station and automotive radar applications, the power consumption of the ADCs is limited by the cooling efficiency of the package. The more power consumption of the ADCs, the more heat is generated, and the more expensive packages with more advanced cooling system are needed.

In the ADC community, two types of figure of merit (FoM) are commonly used:

• Walden FoM [18]:

$$FoM_W = \frac{P}{2 \cdot BW \cdot 2^{\frac{SNDR-1.76}{6.02}}} \quad [J/conversion-step]$$
(2.3)

• Schreier FoM [19]:

$$FoM_S = DR + 10 \cdot \log_{10} \frac{BW}{P} \quad [dB]$$
(2.4)

25

Signal bandwidth	$500 \mathrm{~MHz}$	
Peak SNDR	$\geq 60 \text{ dB}$	
Dynamic range	$\geq 63 \text{ dB}$	
Power consumption	\leq 700 mW	
Walden FoM	\leq 1 pJ/conv. step	
Schreier FoM	$\geq 155 \text{ dB}$	

Table 2.1: ADC main target specifications

In equations (2.3) and (2.4), P is the power consumption of the ADC, BW is the (reported) bandwidth of the ADC, SNDR and DR are in dB. Although the Walden FoM and the Schreier FoM are written in different ways in their equations, they have only two differences. (1) In the Walden FoM SNDR is used to measure the amplitude resolution of the ADC, while in the Schreier FoM DR is used. The Schreier FoM focuses on the noise of the ADCs, and for the Walden FoM both noise and distortion are included. (2) In the Walden FoM the ADC power is doubled when the ADC resolution increases by 6 dB, while in the Schreier FoM the ADC power is quadrupled. The Schreier FoM shows the fundamental relationship between ADC power and ADC resolution because the ADC power is quadrupled to suppress the thermal noise power by 6 dB, while the Walden FoM is based on experimental evidence. Nevertheless, the Walden FoM is suitable for medium-to-low resolution ADCs, while the Schreier FoM is more suitable for high resolution and broadband ADCs [4].

Considering the target applications, the practical limitation from the technology, and the state-of-the-art, the main target specifications for the ADC are chosen and listed in Table 2.1. In Section 2.4, different ADC architectures are compared in the context of these target specifications.

2.3 ADC overview

Analog-to-digital converters can be grouped into different categories. Based on the distribution of the quantization noise in the Nyquist bandwidth, ADCs can be subdivided into Nyquist ADCs and noise-shaping ADCs. Here in this thesis, the Nyquist ADCs are defined as the ADCs with a uniformly distributed quantization noise in the Nyquist bandwidth. On the contrary, the noise-shaping ADCs are defined as the ADCs with a shaped quantization noise in the Nyquist bandwidth.

Because an ideal brick-wall anti-aliasing filter cannot be designed, both Nyquist ADCs and noise-shaping ADCs are oversampled in real receiver applications. The ratio

between half sampling rate and signal bandwidth is defined as oversampling ratio:

$$OSR = \frac{F_s}{2 \cdot BW} \tag{2.5}$$

2.3.1 Nyquist ADCs

Common types of Nyquist ADCs are flash ADCs, pipelined ADCs, successive approximation register ADCs, algorithmic ADCs, dual slope ADCs, etc. Time-interleaving is a commonly used technique to increase the bandwidth of the ADC. Figure 2.4 gives an overview of the current state-of-the-art over different ADC architectures (based on [4]). Figure 2.4a shows the signal BW versus peak SNDR plot, and Figure 2.4b presents the Schreier FoM versus signal BW plot. The target specification is indicated as well.

Flash ADCs are suitable for broadband applications, as the throughput time can be (almost) as short as the comparator delay [20]. The main limitation of flash ADCs is that the number of comparators increases exponentially with the number of bits. An N-bit flash ADC requires $2^N - 1$ comparators. Thus, it is power hungry and area consuming. The resolution for Flash ADCs is typically limited to maximally 6-7 bit, as shown in Figure 2.4a. Figure 2.4b shows that the signal BW of the published state-of-the-art flash ADCs is typically between 100 MHz and 10 GHz, with a Schreier FoM between 120 dB and 150 dB.

Pipelined ADCs are commonly used in broadband (BW \geq 10 MHz) medium-to-high resolution (40 dB \leq SNDR \leq 80 dB) applications, as shown in Figure 2.4a [12]. The principle of pipelined ADCs is to achieve the overall resolution with several pipelined stages. Ideally, the overall resolution of the pipelined ADC equals the sum of the resolutions of every stage. The quantization error of every front-end stage is usually amplified by the residue amplifier to the full-scale of the following stage, and then quantized by the following stage. In reality, the resolution and sampling rate of pipelined ADCs are commonly limited by the performance of the residue amplifier (for example gain error and offset). Recently, a continuous-time pipelined ADC was demonstrated which benefits from inherent anti-alias filtering as well [21]. Figure 2.4b shows that pipelined ADCs (including pipelined SAR ADCs) have demonstrated excellent Schreier FoM recently for broadband medium resolution applications [14, 22].

SAR ADCs are widely used mainly thanks to their power efficiency, which is shown in Figure 2.4b [23, 24]. Moreover, a 1-bit-per-cycle SAR ADC is not sensitive to the comparator offset. For low to medium sampling rate (up to tens of MHz), the core of a SAR ADC can be designed without static current consumption. The resolution of a SAR ADC is commonly limited by the accuracy of the matching inside the DAC. For medium to high resolution above 10 ENOB, usually redundancy is added and sophisticated calibration techniques are applied. Time-interleaving can be applied to further increase the signal bandwidth of the SAR ADC [25].



(a)



(b)

Figure 2.4: (a) Signal BW vs. SNDR plot and (b) Schreier FoM vs. signal BW plot for different ADCs (based on [4]). Here flash ADCs include TI flash ADCs; pipelined ADCs includes pipelined SAR ADCs and TI pipelined ADCs; SAR ADCs include TI SAR ADCs.

Figure 2.4a shows that single-channel SAR ADCs are suitable for low-to-medium speed (BW ≤ 10 MHz) low-to-medium resolution (SNDR ≤ 60 dB) power efficient applications, while adding redundancy and calibration are helpful to increase their resolution, and time-interleaving is useful to increase the ADC BW.

2.3.2 Noise-shaping ADCs

Noise-shaping ADCs trade time resolution for amplitude resolution in a more efficient way considering oversampling than Nyquist ADCs, by filtering of the quantization noise. The signal bandwidth of the noise-shaping ADCs is usually much lower than the Nyquist bandwidth. The common types of noise-shaping ADCs are $\Sigma\Delta$ ADCs and noise-shaping SAR ADCs [26]. This thesis considers voltage-controlled oscillator (VCO)-based ADCs as a type of implementation of $\Sigma\Delta$ ADCs [27].

 $\Sigma\Delta$ ADCs are composed of loop filter, quantizer and feedback DACs. They can be further divided into continuous-time $\Sigma\Delta$ ADCs and discrete-time (DT) $\Sigma\Delta$ ADCs, commonly implemented as switched-capacitor (SC) $\Sigma\Delta$ ADCs.

Switched-capacitor $\Sigma\Delta$ ADCs are suitable for narrow-band high resolution applications [28]. The switched-capacitor circuit can achieve high accuracy and high linearity at the same time [29]. Moreover, SC ADCs benefit from the easy scaling of the bandwidth by just changing the sampling clock rate. Their sensitivity to clock jitter is similar as the Nyquist ADCs.

However, compared to continuous-time $\Sigma\Delta$ ADCs, SC $\Sigma\Delta$ ADCs have two main drawbacks: (1) CT $\Sigma\Delta$ ADCs have the inherent anti-aliasing filtering, while SC $\Sigma\Delta$ ADCs (and most of the Nyquist ADCs) need an external dedicated anti-aliasing filter with an input buffer. (2) With the same amplifier unity-gain bandwidth limitation, CT $\Sigma\Delta$ ADCs can achieve much higher sampling rate than SC $\Sigma\Delta$ ADCs. Figure 2.4a shows that the signal bandwidth of the state-of-the-art SC $\Sigma\Delta$ ADCs is limited to about 20 MHz, which is more than 10 times lower than the maximal reported signal bandwidth of CT $\Sigma\Delta$ ADCs. Figure 2.4b shows that most of the SC ADCs have a Schreier FoM between 140 dB and 170 dB.

Continuous-time $\Sigma\Delta$ ADCs are widely used for medium-to-broad bandwidth (BW ≥ 1 MHz) and medium-to-high resolution (SNDR ≥ 50 dB) applications, as shown in Figure 2.4a [30, 16]. Their loop filters are implemented as continuous-time loop filters. The sampling of the input signal happens at the (input of the) quantizer, which is after the CT loop filter. Thus, CT $\Sigma\Delta$ ADCs benefit from resistive input and inherent anti-aliasing filtering. Compared to ADC architectures with explicit input sample-and-hold circuit, they can relax the anti-aliasing requirement of the input filter, and relax the driving ability requirement of the input buffer [31]. Figure 2.4b shows that CT $\Sigma\Delta$ ADCs have the potential to achieve excellent power efficiency for broadband medium resolution applications [32].

According to the location of the signal band, CT $\Sigma\Delta$ ADCs are further divided into low-pass CT $\Sigma\Delta$ ADCs and band-pass CT $\Sigma\Delta$ ADCs. For some applications, a receiver system with a bandpass $\Sigma\Delta$ ADC shows system-level advantages over the classical IQ receiver with two low-pass ADCs [33, 34]. However, it is too difficult to achieve broadband (BW = 500 MHz), medium resolution (SNDR ≈ 60 dB) and high carrier frequency ($f_c \geq 1$ GHz) at the same time. For example, a band-pass $\Sigma\Delta$ ADC with $f_c = 1$ GHz, BW = 500 MHz and SNDR ≈ 60 dB requires the loop filter amplifier to deliver ≥ 40 dB gain for the frequency 500 MHz - 1.5 GHz, which means a gain-bandwidth product (GBW) of 150 GHz [34]. In this thesis, an inverter-based amplifier is designed to achieve an open-loop GBW of about 12 GHz, with power consumption of 19 mW (details shown in Chapter 11.1). To achieve GBW of 150 GHz, a more complicated circuit for the amplifier has to be investigated, and more power would be spent, which is out of the scope of this thesis. This thesis decides not to choose the BP CT $\Sigma\Delta$ ADC.

2.4 CT $\Sigma \Delta$ ADCs compared to Nyquist ADCs

From Figure 2.4, the target specifications shown in Section 2.2 are promising to be achieved by either CT $\Sigma\Delta$ ADCs or Nyquist ADCs (mainly pipelined ADCs or TI SAR ADCs). To be able to perform the system-level comparison between CT $\Sigma\Delta$ ADCs and Nyquist ADCs, they need to be compared in the context of a typical receiver system, as shown in Figure 2.5.

Figure 2.5a shows the functional block diagram of a part of a typical receiver with a Nyquist ADC with an explicit input S&H circuit. An anti-aliasing filter is needed in this case to suppress the aliased band before the sampling; this filter is used to reduce the out-of-band interference as well. Considering the practical limitation that an ideal brick-wall anti-aliasing filter cannot be designed, the usable signal bandwidth of the Nyquist ADC is typically 1/2 to 1/4 of the Nyquist bandwidth [31, 35]. Figure 2.5a considers an optimistic OSR of 2 for the Nyquist ADC. Thus to achieve the target specifications (signal BW = 500 MHz and SNDR = 60 dB in the signal BW), a Nyquist ADC with a sampling rate of $F_S = 2$ GHz and SNDR = 57 dB for the Nyquist BW is needed, considering the noise is uniformly distributed in the Nyquist BW. In this case, even with a very sharp 6^{th} order LP filter as the anti-aliasing filter with all the poles at the signal band-edge (500 MHz), the suppression is only 39 dB for the worst-case frequency (1.5 GHz). Moreover, the signal is suppressed by 18 dB at the band-edge compared with low frequency, which is not desired for most of the applications as well. The sampling capacitance C_S has to be large enough to suppress the sampled thermal noise. The available tracking time is usually only part of the sampling clock period. Together with the sampling accuracy, they put a very stringent requirement on the bandwidth of the S&H circuit and the driving ability of the input buffer, which is typically the same amplifier used in the anti-aliasing filter in the implementation. Both of the amplifier in the anti-aliasing filter and the S&H



Figure 2.5: Part of a typical receiver with (a) a Nyquist ADC and (b) a CT $\Sigma\Delta$ ADC.

circuit must be linear enough, as their non-linearity directly intervenes with the ADC input. The clock distribution and the decimation filter add up in the total power consumption of this part of the receiver.

On the other hand, Figure 2.5b shows the corresponding part of a typical receiver with a CT $\Sigma\Delta$ ADC with OSR = N. Here a practical low OSR of about 7 is considered, which is based on a system-level design trade-off and will be explained in detail in Chapter 5. Please notice that the chosen 40 nm CMOS technology can offer the required speed of 7 GHz sampling rate for the CT $\Sigma\Delta$ ADC. Compared to the receiver with a Nyquist ADC, the extra speed is utilized by an SDM to achieve the resolution target via the noise shaping followed by the exchange of speed to resolution in the decimation filter. The inherent anti-aliasing filtering of a broadband medium resolution CT $\Sigma\Delta$ ADC is typically more than 60 dB for the worst case frequency, which is often more than the requirement of the applications. For some applications, a LP interference filter is needed to suppress the strong out-of-band interference in front of the CT $\Sigma\Delta$ ADC, such that the possible strong out-of-band interference does not limit the dynamic range of the ADC. The order of this interference filter can be much less than the anti-aliasing filter needed in Figure 2.5a, as the residue of the possible out-of-band interference that is not in the aliased band will be removed by the decimation filter. Moreover, CT $\Sigma\Delta$ ADCs typically have a resistive input. The input resistance for the target specification is about 500 Ω . The driving ability requirement for the input buffer is strongly relaxed compared to the receiver system in Figure 2.5a. Nevertheless, the CT $\Sigma\Delta$ ADC requires a sampling clock with a higher frequency and a more stringent clock jitter requirement. The decimation filter is running at a higher clock frequency, which consumes more power than in the case

Table 2.2: Comparison between receivers with a Nyquist ADC and with a CT $\Sigma\Delta$ ADC

Comparison item	Receiver with a Nyquist ADC	Receiver with a CT $\Sigma\Delta$ ADC
Anti-aliasing and interference filter	③ stringent	\bigcirc relaxed
Input buffer	© capacitive input	© resistive input
S&H circuit	© required	© not required
Clock	© lower frequency	③ higher frequency
Decimation filter	© less power	© more power

of Figure 2.5a.

Table 2.2 summarizes the comparison. This chapter shows a qualitative comparison, and a quantitative comparison will be given in Chapter 11.4. A CT $\Sigma\Delta$ ADC relaxes the requirement of the anti-aliasing and interference filter, and relaxes the driving ability requirement of the input buffer. The receiver with a CT $\Sigma\Delta$ ADC does not need an S&H circuit preceding the ADC. However, the power consumption of the clock distribution and the decimation filter is higher. Overall, it is promising to achieve a lower total receiver power with a CT $\Sigma\Delta$ ADC compared to a Nyquist ADC.

2.5 Increasing the bandwidth of CT $\Sigma\Delta$ ADCs

The previous section presents that CT $\Sigma\Delta$ ADCs have many advantages compared to Nyquist ADCs, including inherent anti-aliasing filtering, resistive input, and omitting the input S&H circuit. These properties can simplify the receiver architecture, and benefit the system integration. This is the main reason why many modern broadband receiver applications, with the whole receiver (or transceiver, or a larger system including the digital circuitry) integrated on one chip, use CT $\Sigma\Delta$ ADCs [1, 2, 9].

Figure 2.4a shows that the signal BW of most of the state-of-the-art CT $\Sigma\Delta$ ADCs is limited to around 150 MHz. The motivation of this thesis is to bring the advantages of CT $\Sigma\Delta$ ADCs to a broader bandwidth. The target ADC signal bandwidth in this thesis is 500 MHz.

In the following chapters, broadband CT $\Sigma\Delta$ ADCs refer to the ADCs with signal BW and peak SNDR performance beyond the 'Jitter = 1 ps' line in Figure 2.4a. For the target specifications and applications of this thesis, broadband CT $\Sigma\Delta$ ADCs can be interpreted as ADCs with signal BW higher or equal to 100 MHz as well (with DR ≥ 60 dB).

2.6 Conclusions

The signal bandwidth in many communication and automotive applications has been increasing exponentially in the last few decades. ADCs with broader bandwidth have been developed in the same time, which make use of the speed advantage of more advanced technology. ADCs can be subdivided into Nyquist ADCs and noise-shaping ADCs based on the distribution of the quantization noise in the Nyquist bandwidth. $\Sigma\Delta$ ADCs trade time resolution for amplitude resolution in an efficient way by means of aggressive filtering of the quantization noise. Continuous-time low-pass $\Sigma\Delta$ ADCs are promising for the target specifications thanks to their resistive input, inherent anti-aliasing filtering and potential for broad bandwidth. Compared to a receiver with a Nyquist ADC with explicit S&H circuit, a CT $\Sigma\Delta$ ADC can relax the requirements on the anti-aliasing and interference filter, and input buffer. Moreover, it does not require an S&H circuit in front of the ADC. Although it leads to higher power consumption in clocking and decimation filter, the total receiver power with a CT $\Sigma\Delta$ ADC is promising to be lower. The motivation of this thesis is to bring the advantages of CT $\Sigma\Delta$ ADCs to a broader bandwidth.
Chapter 3

Architecture choices for broadband CT $\Sigma\Delta$ ADCs

This chapter elaborates on the architecture design choices for broadband $CT \Sigma \Delta$ ADCs. It is part of the top-down approach of the broadband $CT \Sigma \Delta$ ADC design. Firstly, this chapter gives a general description and definition for CT SDMs. Afterwards, this chapter discusses SDM parameter design choices and topology design choices separately. The SDM parameter design choices include noise transfer function order, quantizer resolution, OSR, zero optimization and out-of-band gain. The stability of the SDM is discussed as well. The SDM topology design choices contain single-loop, MASH and Sturdy MASH SDMs. As there are multiple trade-offs among the SDM parameter and topology choices, the chapter further reviews and benchmark the state-of-the-art broadband medium-to-high resolution $CT \Sigma \Delta$ ADCs to learn from them. In the end, the SDM parameter design space is reduced and conclusions are drawn.

3.1 CT SDM general description and definition

This section starts with a general description and definition for CT SDMs. A simplified block diagram of a CT SDM is shown in Figure 3.1a. Here a single-loop SDM is taken as an example. A CT SDM contains a continuous-time loop filter, a quantizer and one or more feedback DACs. They compose a negative feedback loop. Although a multi-bit quantizer is drawn in Figure 3.1a, the quantizer can be a 1-bit quantizer as well. The CT loop filter has a transfer function $H_s(s)$ for SDM input U, and a transfer function $H_n(s)$ for the feedback DAC output after taking the inverse sign.

Both the linear and non-linear characteristics of the CT SDM need to be considered. This section starts with the linear characteristics. Usually, designers use a linear model to analyze the CT SDM, which is depicted in Figure 3.1b. In the linear model,



Figure 3.1: (a) Simplified block diagram of a CT SDM [19] (b) CT linear model of the SDM (c) Equivalent DT model of the SDM (based on [36]).

the CT loop filter is modelled as its continuous-time transfer functions $H_s(s)$ and $H_n(s)$. The quantizer is modelled as a sampler, a linear gain g_L and a quantization noise q. The DAC is modelled as its continuous-time transfer function $H_{DAC}(s)$, which takes the DAC pulse into account. The total delay of the feedback loop, including the quantizer delay, DAC delay and the propagation delay from quantizer to DAC, is modeled as a delay τ . In the linear model, the non-linear characteristic of the CT

SDM is ignored.

Most of the theories and analysis on the noise shaping and stability of the SDMs are done in discrete-time domain. For the purpose of analysis and design of the NTF, the equivalent discrete-time SDM model can be calculated, which is shown in Figure 3.1c. The equivalent DT loop filter transfer function for the quantization noise $H_n(z)$ can be calculated, taking into account the CT loop filter $H_n(s)$, DAC pulse $H_{DAC}(s)$, loop delay τ , and quantizer linear gain g_L . Impulse invariant transformation can be applied to guarantee that for the quantization noise q, the CT SDM model and equivalent DT SDM model have the same value for every sampled quantizer input at $n \cdot T_S$ [36]. A detail CT to DT transformation for a CT MASH SDM is given in Chapter 5.2. Notice that the DT SDM is equivalent to the CT SDM only for the quantization noise transfer function, but not for the signal transfer function (STF).

The noise transfer function is defined as the transfer function from the quantization noise q to the SDM output V, which can be calculated as:

$$NTF(z) = \frac{1}{1 + H_n(z)}$$
 (3.1)

The shape of the NTF is mainly influenced by 4 parameters – NTF order, OBG, OSR and optimized zeros. Actually, the definition on the design parameters of the NTF and the resulting properties of the NTF depends on the synthesis flow of the NTF and the SDM as well. This thesis assumes the NTF is directly synthesized using the Delta Sigma toolbox [37]. NTF order is defined as the number of zeros in the NTF(z). OBG is defined as the magnitude of the NTF at the frequency of $\frac{F_s}{2}$, which is referred as H_{inf} in the literature as well. If the NTF order is higher or equal to two, one or more pairs of NTF zeros can be designed to be at an optimized frequency instead of DC, which gives more aggressive in-band noise shaping. This technique is referred as zero optimization. In this thesis, "more aggressive" in-band noise shaping means the in-band quantization noise is suppressed more strongly (same definition as [19]).

The signal transfer function is defined as the transfer function from the SDM input U to the SDM output V. For a CT SDM, the STF is a combined transfer function of s-domain and z-domain, because the input of the SDM is not sampled. This thesis gives an approximation of the STF for the CT SDM model in Figure 3.1b as:

$$STF(z) \approx \mathcal{Z}\left\{\mathcal{L}^{-1}\left\{H_s(s)\right\}_{t=n}\right\} \cdot g_L \cdot NTF(z)$$
(3.2)

Since $H_s(s)$ and $H_n(s)$ can be designed separately, the STF can be optimized independently from the NTF.

The non-linear characteristic of the ideal CT SDM model is due to the quantizer. Especially a 1-bit quantizer shows strong non-linear behavior, and its linear gain g_L is arbitrarily defined [19]. A multi-bit quantizer shows less non-linearity compared with a 1-bit quantizer, and its linear gain g_L is well defined. The non-linear characteristic of the quantizer is modeled as its phase and gain uncertainty as well [38].



Figure 3.2: (a) NTF with different order (b) NTF with and without optimized zeros (CRFF or CRFB structure used here). The Delta Sigma toolbox is used [37].

3.2 SDM parameter design choices and related loop properties

This section discusses the SDM parameter design choices, including NTF order, zero optimization, quantizer resolution, OBG and OSR. Afterwards, the stability of the SDMs is addressed.

3.2.1 NTF order

The NTF order directly influences the slope of the NTF. Figure 3.2a shows the NTF magnitude from 1st order to 4th order, assuming the OBG = 1.5 and all the NTF zeros are at DC. For an Nth order loop filter, the NTF has a slope of $20 \cdot N$ dB/decade in the low frequency band. Table 3.1 shows the synthesized NTFs and summarizes the simulated SQNR for sine input with frequency $f_{in} = 0.0051$ F_S and amplitude $A_{in} = 0.5$ V_{ref} such that the quantizer is not overloaded. OSR = 32 in this simulation. The SQNR increases from 38.5 dB to 61.6 dB when NTF order increases from 1st order to 4th order.

3.2.2 Zero optimization

Figure 3.2b presents the 4^{th} order NTF magnitude of three cases: (1) with all zeros at DC (2) with 2 pairs of zeros at optimized frequency (3) with 1 pair of zeros at optimized frequency and the other pair of zeros at DC. For all the three cases, OSR

NTF order	quantizer	$\mathrm{NTF}(\mathrm{z})$	SQNR
1^{st} order	1-bit	$\frac{z-1}{z-0.3333}$	$38.5~\mathrm{dB}$
2^{nd} order	1-bit	$\frac{(z-1)^2}{z^2-1.225z+0.4415}$	$51.2~\mathrm{dB}$
3^{rd} order	1-bit	$\frac{(z-1)^3}{(z-0.6694)(z^2-1.531z+0.6639)}$	$59.2 \mathrm{dB}$
4^{th} order	1-bit	$\frac{(z-1)^4}{(z^2-1.493z+0.5647)(z^2-1.702z+0.7871)}$	$61.6~\mathrm{dB}$

Table 3.1: SQNR comparison for different NTF order

For all NTFs, OSR = 32, OBG = 1.5, no zero optimization. Simulated SQNR for sine input with frequency $f_{in} = 0.0051 \text{ F}_S$ and amplitude $A_{in} = 0.5 \text{ V}_{ref}$.

Table 3.2: SQNR comparison with and without optimized zeros

NTF order	optimized zeros?	$\mathrm{NTF}(\mathrm{z})$	SQNR
4^{th} order	no	$\frac{(z-1)^4}{(z^2-1.493z+0.5647)(z^2-1.702z+0.7871)}$	$61.6~\mathrm{dB}$
4^{th} order	1 pair of zeros optimized	$\frac{(z-1)^2(z^2-1.993z+1)}{(z^2-1.491z+0.5633)(z^2-1.7z+0.7863)}$	$73.4~\mathrm{dB}$
4^{th} order	2 pairs of zeros optimized	$\frac{(z^2-1.999z+1)(z^2-1.993z+1)}{(z^2-1.49z+0.563)(z^2-1.7z+0.7861)}$	$75.8~\mathrm{dB}$

For all NTFs, OSR = 32, OBG = 1.5. 1-bit quantizer is used. Simulated SQNR for sine input with frequency $f_{in} = 0.0051 F_S$ and amplitude $A_{in} = 0.5 V_{ref}$.

= 32 and OBG = 1.5 are used. Optimized zeros can greatly suppress the in-band quantization noise close to the band-edge. Table 3.2 shows the corresponding NTFs and simulated SQNR for these three cases for sine input with frequency $f_{in} = 0.0051$ F_S and amplitude $A_{in} = 0.5 V_{ref}$. One pair of optimized zeros increases the SQNR from 61.6 dB to 73.4 dB. 2 pairs of optimized zeros further increase the SQNR to 75.8 dB.

3.2.3 Quantizer resolution

Quantizer resolution is an important parameter in the $\Sigma\Delta$ ADC architecture design. On one hand, $\Sigma\Delta$ ADCs with 1-bit quantizer is not sensitive to comparator offset and its 1-bit feedback DAC is inherently linear. On the other hand, multi-bit quantization offers a lower original unshaped quantization noise. Multi-bit quantizers have lower original unshaped quantization noise and a well-defined linear gain compared to a 1bit quantizer. Thus, with the same targeted maximum stable input amplitude (MSA), the applicable out-of-band gain (OBG) is higher for SDMs with multi-bit quantizers than SDMs of the same order with a 1-bit quantizer, achieving a more aggressive noise shaping (see Section 3.2.4). This thesis refers to this property in the sense that SDMs with multi-bit quantizers are "more stable" than SDMs with a 1-bit quantizer



Figure 3.3: 4^{th} order NTF with OBG = 1.5 and 4. Both with 2 pairs of optimized zeros. OSR = 32. CRFF or CRFB structure.

(same definition of "stable" as in [19]).

3.2.4 Out-of-band gain

Loop stability and aggressiveness of filter are related to the loop filter order and OBG. OBG indicates how much the high frequency quantization noise is amplified at $\frac{F_s}{2}$. For Nth order loop filter, the theoretical maximal OBG is 2^N when the NTF is designed with a flat transfer function close to $\frac{F_s}{2}$. Since high-order (loop filter order \geq 3) 1-bit SDMs cannot tolerate too much amplification of high frequency quantization noise, their maximal applicable OBG is typically less than 2 [39]. For SDMs with multi-bit quantization, the applicable OBG is much larger. Figure 3.3 shows two 4^{th} order NTFs, one with OBG = 1.5 and the other with OBG = 4. The NTF with OBG = 4 has more aggressive noise shaping, and its in-band quantization noise level is about 33 dB lower than OBG = 1.5.

Theoretically, when the quantizer resolution increases by 1 bit, the original unshaped quantization noise reduces by 6 dB. Moreover, multi-bit quantization allows to use higher OBG than 1-bit quantization, which further improve the SQNR. Figure 3.4a shows the peak SQNR versus OBG, for a 3^{rd} order SDM in cascade-of-integrators, feedforward form (CIFF) with different quantizer resolution. Figure 3.4b shows the corresponding input amplitude to achieve peak SQNR for every case. With the same quantizer resolution, when OBG increases, the high frequency quantization noise is amplified more. Thus, when OBG increases, the maximum stable input amplitude (MSA) of the SDM is reduced. In practice, the peak SQNR should be achieved for an input amplitude higher or equal to 0.5 V_{ref} , otherwise it implies that too much quantization noise is in the system. For the simulated 3^{rd} order CIFF structure, with 1-bit quantizer, the maximum SQNR is 71 dB with OBG = 1.7. On the other hand, for a 17-level quantizer, the maximum SQNR is more than 114 dB, which can be



Figure 3.4: (a) SQNR vs. OBG with 2-level to 17-level quantizers (b) The corresponding input amplitude. 3^{rd} order CIFF structure. OSR = 32. $f_{in} = 0.001 \text{ F}_S$.

achieved with OBG from 6 to 7.9.

3.2.5 OSR

OSR is another important design parameter. Equation (2.5) shows that for a given signal BW, OSR is proportional to F_S . For narrow-band high-resolution (SNDR \geq 80 dB) $\Sigma\Delta$ ADCs, typically a high OSR (\geq 30) is used, to suppress the in-band quantization noise [40]. For broadband medium-resolution (50 dB \leq SNDR \leq 70 dB) $\Sigma\Delta$ ADCs, since the sampling frequency F_S is limited by the maximum speed offered by the technology, usually a relatively low OSR (\leq 30) is used [16].

Figure 3.5 shows two 4th order NTFs with OSR = 8 and 32, respectively. When OSR increases, the in-band quantization noise is suppressed more. Table 3.3 shows that for 4th order SDMs with 3-bit quantizer, when OSR increases from 8 to 32, SQNR increases from 67.6 dB to 121.4 dB (for input amplitude $A_{in} = 0.5 V_{ref}$).

3.2.6 Stability

The stability analysis of the SDM can be found in the literature [19]. For this analysis both the linear model of the SDM and the non-linearity of the quantizer need to be considered. Firstly, the calculation of NTF is based on the linear model of SDM. The linear model of SDM ignores the non-linearity of the quantizer. Especially for 1-bit quantizer, its non-linearity is much higher than multi-bit quantizer. This thesis only gives some guidelines which are useful in the practical design of broadband CT $\Sigma\Delta$ ADCs.



Figure 3.5: 4^{th} order NTF with OSR = 8 and 32. Both with 2 pairs of optimized zeros. OBG = 4. CRFF or CRFB structure.

Table 3.3: SQNR comparison for different OSR

NTF order	quantizer	OSR	$\mathrm{NTF}(\mathrm{z})$	SQNR
4^{th} order	3-bit	8	$\frac{(z^2 - 1.982z + 1)(z^2 - 1.887z + 1)}{(z^2 - 0.7131z + 0.1458)(z^2 - 0.6831z + 0.3985)}$	$67.6~\mathrm{dB}$
4^{th} order	3-bit	32	$\frac{(z^2 - 1.999z + 1)(z^2 - 1.993z + 1)}{(z^2 - 0.7288z + 0.1516)(z^2 - 0.7145z + 0.4083)}$	$121.4~\mathrm{dB}$

For all NTFs, OBG = 4 and 2 pairs of zeros are optimized. Simulated SQNR for sine input with frequency $f_{in} = 0.325$ ·BW and amplitude $A_{in} = 0.5$ V_{ref}. CRFF or CRFB structure.

- In general, higher-order SDMs have a worse stability than lower-order SDMs. 1^{st} order SDMs are always stable with DC input with an amplitude less than the DAC full-scale. 1^{st} order and 2^{nd} order SDMs have much better stability than high-order SDMs with order ≥ 3 [19].
- In general, multi-bit quantization has better stability than 1-bit quantization. The reason it that multi-bit quantization has less original unshaped quantization noise, and it has a well-defined linear gain as well.
- In general, higher OBG leads to worse stability. However, for SDMs with loop filter order ≥ 2 with multi-bit quantization, too low OBG (< 2) makes the SDM tonal. In practice, high-order (≥ 3) 1-bit SDMs typically have OBG around 1.5, which is a trade-off between maximizing peak SQNR and limiting the amplification of high-frequency quantization noise. SDMs with loop filter order ≥ 2 and multi-bit (≥ 2 -bit) quantization usually have OBG ≥ 2 (see Figure 3.4).

As the feedback DACs have a fixed output full scale, the total output power of the DAC feedback is limited. This total power contains signal power and a shaped quantization noise power. If the SDM input signal power is above the maximal allowed signal power of the SDM, the SDM is overloaded. Applications typically require that the SDM should recover from the overloading after the input overloading condition is removed, which is called overloading recovery. Automatic overloading recovery is generally achieved by 1^{st} order and 2^{nd} order SDMs without specific design, which is not the case for high-order (≥ 3) SDMs [41]. For high order SDMs, the overloading recovery can be achieved usually by two ways. The first way is to reset the loop filter internal nodes when the SDM overloading is detected. The second way is to put clippers at the loop filter internal nodes, such that when the overloading happens the SDM works as a 1^{st} order SDM.

3.3 SDM topology design choices

This section further discusses the topology choice between single-loop, MASH, and Sturdy MASH SDMs.

3.3.1 Single-loop SDMs

SDMs with only one Sigma-Delta loop are called single-loop SDMs, such as the SDM shown in Figure 3.1a [30]. Single-loop SDMs typically have only one quantizer and one loop filter. The order of the NTF and the quantizer resolution are still design freedoms. Single-loop SDMs are widely used in many applications because of their simplicity compared to the MASH structure.

3.3.2 MASH

MASH SDMs are referred as cascaded SDMs as well in the literature [42]. They have more than one Sigma-Delta loop. Figure 3.6 shows the simplified block diagram of an exemplary CT MASH SDM. In this figure a MASH SDM containing two Sigma-Delta loops is shown as an example, in practice a MASH SDM can contain 3 or more Sigma-Delta loops. The principle of MASH SDM is that the quantization noise of the 1^{st} loop is the input of the 2^{nd} loop. The following equations are valid for the CT MASH SDM shown in Figure 3.6:

$$V_1 = STF_1(z) \cdot U + NTF_1(z) \cdot q_1 \tag{3.3}$$

$$V_2 = STF_2(z) \cdot U_2 + NTF_2(z) \cdot q_2$$
(3.4)

$$U_2 = -a_{12} \cdot q_1 \tag{3.5}$$

Here, $STF_1(z)$ and $NTF_1(z)$ are the STF and NTF of the 1^{st} loop, and $STF_2(z)$ and $NTF_2(z)$ are the STF and NTF of the 2^{nd} loop. q_1 and q_2 are the quantization noise of the 1^{st} and 2^{nd} loop. a_{12} is a gain between 2 Sigma-Delta loops, which is called



Figure 3.6: Simplified block diagram of a CT MASH SDM.

inter-stage gain (ISG). Here for simplicity, the total delay of 1^{st} quantizer and DAC₁₂ is considered to be much lower than the sampling clock period T_s and is neglected.

In this case, the following noise cancellation filter (NCF) can be applied:

$$NCF_1(z) = STF_2(z) \tag{3.6}$$

$$NCF_2(z) = \frac{NTF_1(z)}{a_{12}}$$
 (3.7)

And the final SDM output can be calculated as:

$$V = NCF_{1}(z) \cdot V_{1} + NCF_{2}(z) \cdot V_{2}$$

= $STF_{1}(z) \cdot STF_{2}(z) \cdot U + \frac{1}{a_{12}}NTF_{1}(z) \cdot NTF_{2}(z) \cdot q_{2}$ (3.8)

Thus, the quantization noise of the 1^{st} Sigma-Delta loop is completely canceled, and the quantization noise of the 2^{nd} Sigma-Delta loop is shaped by the total order of both Sigma-Delta loops and the inter-stage gain. Typically multi-bit quantizers are used in the MASH SDMs [16]. Since the swing of the original unshaped quantization noise q_1 is lower than the maximal input swing of the 2^{nd} loop, an ISG a_{12} of larger than one can be applied to amplify the swing of q_1 . Applying ISG of larger than one further increases the SQNR.

MASH SDMs have two main advantages compared to single-loop SDMs. First, MASH SDMs can be built from 1^{st} and 2^{nd} order Sigma-Delta loops, which have much better stability than high-order (≥ 3) Sigma-Delta loops. Thus, MASH SDMs can achieve high-order noise shaping and maintain good stability as 1^{st} and 2^{nd} order SDMs. Second, when multi-bit quantizers are used in the MASH SDMs, inter-stage gains of larger than one can be applied which can further increase the SQNR.



Figure 3.7: Simplified block diagram of a CT Sturdy MASH SDM.

MASH SDMs have some drawbacks. First, they need noise cancellation filters to remove the quantization noise of all Sigma-Delta loops except for the last loop, which are not required by the single-loop SDMs. Second, the design of MASH SDMs is generally more complicated than single-loop SDMs, especially regarding circuit non-idealities, such as excess-loop-delay compensation, analog and digital filter mismatch, parasitic poles and extra phase shift in the loop, etc. An overview of the challenges and non-idealities in broadband CT $\Sigma\Delta$ ADCs is given in Chapter 4. MASH SDMs are generally used to achieve broader BW and/or higher SNDR than single-loop SDMs at the cost of increased complexity.

3.3.3 Sturdy MASH

An architecture called 'Sturdy MASH' was proposed, in which the quantizer output of the 2^{nd} loop is added back in the 1^{st} loop, to simplify the noise cancellation filter and to relax the analog and digital loop filter matching requirements [43, 44]. It has been demonstrated for broadband CT $\Sigma\Delta$ ADCs [45, 46]. Figure 3.7 presents a simplified block diagram of a CT Sturdy MASH SDM [45]. Compared to the standard MASH structure, the noise cancellation filter of a Sturdy MASH is simpler. The matching requirement between the analog filter and digital NCF is relaxed.

The Sturdy MASH proposed in [44] has two main drawbacks. This section takes a Sturdy MASH with two Sigma-Delta loops as an example. First, the quantization noise of the 1^{st} loop is not canceled for the Sturdy MASH, but shaped by the total order of 1^{st} and 2^{nd} loops. In this case, the final output of Sturdy MASH contains the quantization noise of both 1^{st} and 2^{nd} loop. Though inter-stage gain can be applied

in the Sturdy MASH, it only suppresses the quantization noise of the 2^{nd} loop. Thus, the improvement on SQNR from the ISG is much reduced, compared to standard MASH. Second, the quantization noise of the 2^{nd} loop is added in the 1^{st} loop in Sturdy MASH. Thus, in the 1^{st} loop the total power of unshaped quantization noise is higher, and only less aggressive NTF can be used for the 1^{st} loop. Because of these two drawbacks, the achievable SQNR of a Sturdy MASH is lower than the standard MASH for the same loop filter order and quantizer resolution.

The CT Sturdy MASH demonstrated in [45, 46] has one additional drawback, which limits its application for very-broad-band ADCs. The total excess loop delay in the single-loop or MASH SDMs can be calculated as (see Chapter 4.5):

$$\tau_{ELD,S.L. or MASH} = \tau_Q + \tau_{DAC} \tag{3.9}$$

Here, τ_Q is the quantizer delay, and τ_{DAC} is the DAC delay. The propagation delay from the quantizer to the DAC is considered as part of τ_{DAC} .

In the CT Sturdy MASH, the quantizer output of the 2^{nd} loop is added back in the 1^{st} loop. In Figure 3.7, one feedback loop contains Quantizer 1, DAC₁₂, Quantizer 2 and DAC₂₁. Thus, the total ELD in the loop is:

$$\tau_{ELD,SMASH} = \tau_{Q1} + \tau_{DAC12} + \tau_{Q2} + \tau_{DAC21} \tag{3.10}$$

Here, τ_{Q1} is the delay of quantizer 1, τ_{DAC12} is the delay of DAC₁₂, τ_{Q2} is the delay of quantizer 2, and τ_{DAC21} is the delay of DAC₂₁ [45, 46].

Assuming the same quantizer delay and the same DAC delay for CT single-loop, MASH and Sturdy MASH SDMs (considering the same technology, same metastability requirements, etc.), the highest achievable sampling rate of the CT single-loop or MASH SDMs is two times of that of the CT Sturdy MASH SDMs. In the literature, the highest sampling rate for multi-bit CT MASH (and single-loop) SDMs is 8 GHz [16], and the highest sampling rate for multi-bit CT Sturdy MASH SDMs is 1.8 GHz [45] ([16] and [45] are both in 28 nm CMOS). This drawback limits the application of CT Sturdy MASH for very-broad-band $\Sigma\Delta$ ADCs, for which the sampling rate has to be maximized to achieve the target very broad signal bandwidth.

3.4 Benchmarking different $\Sigma\Delta$ **ADCs**

The previous sections show that the design of CT SDM contains a lot of trade-offs between aggressiveness of noise shaping, stability and system complexity. Chapter 4 shows more trade-offs on noise, linearity, power and circuit complexity. Thus, looking at the state-of-the-art broadband medium-to-high resolution CT $\Sigma\Delta$ ADCs for their SDM parameter and topology design choices is helpful and beneficial for our design.

Figure 3.8 and Figure 3.9 present the signal BW vs. SNDR plot for the state-of-theart broadband (BW \geq 10 MHz) medium-to-high resolution (SNDR \geq 50 dB) CT



Figure 3.8: Signal BW vs. SNDR plot for CT $\Sigma\Delta$ ADCs with BW ≥ 10 MHz and SNDR ≥ 50 dB: (a) grouped by NTF order (b) grouped by quantizer resolution (based on [4]). Here NTF order is the total order including loop filter order and noise-shaping of quantizers. For a MASH/SMASH structure, NTF order is the total effective noise-shaping order of the MASH/SMASH. If more than one quantizer is used, the highest resolution is shown in (b).

 $\Sigma\Delta$ ADCs (based on [4]). Here, all of the 53 reported CT $\Sigma\Delta$ ADCs are grouped according to 4 categories: (1) NTF order (2) quantizer resolution (3) OSR and (4) single-loop/MASH/SMASH.

Figure 3.8a shows the NTF order. About 85% of these $\Sigma\Delta$ ADCs have a total NTF



Figure 3.9: Signal BW vs. SNDR plot for CT $\Sigma\Delta$ ADCs with BW \geq 10 MHz and SNDR \geq 50 dB: (a) grouped by OSR (b) grouped by single-loop/MASH/SMASH (based on [4]).

order of either 3 or 4. Please notice that all of the 3 $\Sigma\Delta$ ADCs with NTF order of 6 are bandpass $\Sigma\Delta$ ADCs (or have a bandpass mode) [34, 47, 48]. The effective noise-shaping of these bandpass $\Sigma\Delta$ ADCs is actually 3^{rd} order. So the majority of broadband medium-to-high resolution CT $\Sigma\Delta$ ADCs have 3^{rd} or 4^{th} order noise shaping, which is a trade-off between in-band quantization noise suppression and SDM stability and complexity.

Figure 3.8b presents the quantizer resolution. About 15% of these ADCs only use

1-bit quantization, since the 1-bit quantizer is not sensitive to its offset, and 1-bit DAC is inherently linear [9]. 55.6% of the ADCs in Figure 3.8b are using quantizers between 8-level and 17-level. The majority of these $\Sigma\Delta$ ADCs use flash sub-ADCs as the quantizers. Among all of the 12 $\Sigma\Delta$ ADCs with quantizer resolution ≥ 4.5 bit, only 2 $\Sigma\Delta$ ADCs use flash quantizers, and the other 10 $\Sigma\Delta$ ADCs use SAR, VCO-based, gated-ring-oscillator (GRO)-based, time-to-digital converter (TDC) or subranging quantizers [49, 27, 50, 51, 52].

Figure 3.9a shows the OSR. All of the 7 $\Sigma\Delta$ ADCs with signal BW ≥ 100 MHz have OSR between 8.6 and 16. For these 7 very-broad-band (BW ≥ 100 MHz) medium resolution (64 dB \leq SNDR ≤ 74 dB) $\Sigma\Delta$ ADCs, reducing the OSR would increase the in-band quantization noise, while increasing the OSR would require an even higher sampling clock rate, which can reduce the power efficiency or it is even limited by the technology. Among all of the 8 $\Sigma\Delta$ ADCs with SNDR ≥ 77 dB, 7 of them have OSR ≥ 20 . For them, high OSR (≥ 20) facilitates the sufficient noise-shaping of the in-band quantization noise to achieve the targeted high SNDR.

In Figure 3.9b the $\Sigma\Delta$ ADCs are subdivided into single-loop, MASH and Sturdy MASH $\Sigma\Delta$ ADCs. Among the 53 $\Sigma\Delta$ ADCs, 45 ADCs are using single-loop SDM structures. 6 MASH $\Sigma\Delta$ ADCs and 2 Sturdy MASH $\Sigma\Delta$ ADCs are reported here. Among the 7 $\Sigma\Delta$ ADCs with signal BW \geq 100 MHz, one of them uses the MASH structure and the other 6 ADCs are single-loop $\Sigma\Delta$ ADCs.

3.5 Design-space reduction

This section reviews the state-of-the-art broadband medium-to-high resolution CT $\Sigma\Delta$ ADCs in the context of the target specifications listed in Section 2.2. To achieve the target specifications (BW = 500 MHz, SNDR \geq 60 dB), an NTF of either 3rd or 4^{th} order is most promising. Since in our case the ADC non-linearity is tolerable as long as it is not limiting the SNDR target, multi-bit quantization is more beneficial because of its lower original unshaped quantization noise, better stability and enabling of more aggressive noise shaping. Flash quantizers between 8-level to 17-level are most promising from Figure 3.8b. Since this thesis targets at a very broad BW of 500 MHz, a relatively low OSR between 5 and 10 is a good choice. A higher OSR requires a sampling rate of more than 10 GHz which is not feasible to be implemented in the chosen 40nm CMOS technology considering multi-bit quantization. Both single-loop and MASH SDM structures are promising for our target specifications. This thesis decides not to choose the Sturdy MASH since the achievable SQNR of a Sturdy MASH is lower than the standard MASH for the same loop filter order and quantizer resolution, and the achievable sampling rate of a Sturdy MASH is lower than the standard MASH and single-loop SDMs. The exact system-level simulations regarding different NTF order, quantizer resolution, OSR for different single-loop and MASH structures are presented in Chapter 5.

3.6 Conclusions

Various design choices should be made for $\Sigma\Delta$ ADC architectures. The preliminary SDM design parameters are loop filter order, quantizer resolution and OSR. OBG and zero optimization should be decided according to the applications. Stability and STF should be considered as well. SDMs are further subdivided into single-loop and MASH SDMs. From the review of the state-of-the-art broadband medium-to-high resolution CT $\Sigma\Delta$ ADCs, the SDM design space is reduced for the target specifications. SDM architectures with 3^{rd} or 4^{th} order NTF, multi-bit quantization between 8-level and 17-level, OSR between 5 and 10, either with a single-loop or a MASH structure are promising. The exact SDM architecture design must consider the challenges, nonidealities and limitations from the circuit-level implementation, which is discussed in the next Chapter.

Chapter 4

Challenges of broadband CT $\Sigma\Delta$ ADCs

This chapter discusses the challenges, non-idealities and limitations of the circuit implementation of broadband $CT \Sigma \Delta$ ADCs. It is part of the bottom-up approach of the broadband $CT \Sigma \Delta$ ADC design. In this chapter, broadband $\Sigma \Delta$ ADCs refer to ADCs with signal BW of 100 MHz and higher. Firstly, the loop filter non-idealities are studied, including amplifier limited GBW, noise, non-linearity, and common-mode stability. This chapter further discusses the challenges in the quantizer implementation, including offset, quantization gain and metastability, and hysteresis. Afterwards, the DAC non-idealities are studied, including non-linearity and noise. Various clock non-idealities are discussed, like clock jitter, clock skew, and dutu-cucle error. After describing the non-idealities in the building blocks, system-level non-idealities are discussed, including excess loop delay, coefficients variation, and digital and analog filter mismatch for MASH $\Sigma\Delta$ ADCs. Supply and reference non-idealities are discussed as well. This chapter presents the speed-resolution limitations from CMOS technologies. The non-idealities, challenges and limitations from circuit implementation justify limiting our design space for the SDM architecture design. In the end, conclusions are drawn.

4.1 Loop filter non-idealities

In low-pass CT $\Sigma\Delta$ ADCs, the loop filter is a low-pass filter, which is implemented from continuous-time integrators. Figure 4.1a shows the mathematical model of an ideal CT integrator. The integrator constant c is the UGBW of the integrator in [rad/s]. Generally a CT integrator can be implemented as an RC integrator or a Gm-C integrator.

Figure 4.1b presents an RC integrator. In the RC integrator, the amplifier is used in a



Figure 4.1: (a) Mathematical model of an integrator (b) an RC integrator (c) a Gm-C integrator.

feedback loop. g_m is the transconductance of the amplifier. A resistor R_z is added in series with the integration capacitor C to create a zero in the RC integrator transfer function, to compensate the phase shift due to the limited gain of the amplifier. The transfer function of the RC integrator is:

$$\frac{V_{out}}{V_{in}} = \frac{(g_m R_z - 1)sC + g_m}{sC(1 + g_m R)}$$
(4.1)

If $R_z = \frac{1}{g_m}$, the transfer function becomes:

$$\frac{V_{out}}{V_{in}} = \frac{g_m}{sC(1+g_mR)} \tag{4.2}$$

Commonly $g_m R \gg 1$, thus the UGBW of the RC integrator is:

$$UGBW \approx \frac{1}{2\pi RC} \quad [\text{Hz}]$$
 (4.3)

Figure 4.1c shows a Gm-C integrator. It uses the amplifier in an open loop. Its transfer function is:

$$\frac{V_{out}}{V_{in}} = \frac{g_m}{sC} \tag{4.4}$$

The UGBW of the Gm-C integrator can be calculated as:

$$UGBW = \frac{g_m}{2\pi C} \quad [\text{Hz}] \tag{4.5}$$

When the order of the loop filter is larger than one, a set of two integrators can be combined into a resonator, as shown in Figure 4.2a (see Chapter 3.2.2). A resonator



Figure 4.2: (a) Mathematical model of a resonator (b) a single-opamp resonator implementation.

can be implemented as two separate integrators, or a single-opamp resonator [53, 54]. Figure 4.2b shows one circuit implementation of a single-opamp resonator [54]. It uses the amplifier in a feedback loop. The resonance condition is [55, 9]:

$$\frac{C_1}{C_2} = \frac{R_1 - R_2}{R_1} \tag{4.6}$$

And the transfer function of the resonator is [55, 9]:

$$\frac{V_{out}}{V_{in}} = \frac{R_1}{R_{in}} \cdot \frac{1 + \frac{s}{R_2 C_2}}{1 + \frac{s^2}{R_1 R_2 C_1 C_2}}$$
(4.7)

Limited GBWs of the amplifiers have influence on the transfer function of the loop filter. It can introduce both an amplitude error and a phase error [56]. The parasitic capacitance at the output node of an RC integrator and a single-opamp resonator generates a parasitic pole. The phase shift of this output parasitic pole can be compensated by adding a resistor R_z in series with the integration capacitor (C in Figure 4.1b, or C₁ in Figure 4.2b). The parasitic capacitance on the output node of a Gm-C integrator can be considered as part of the total integration capacitance. Thus, the output parasitic capacitance does not add a parasitic pole to the transfer function of a Gm-C integrator. However, the parasitic capacitance is not well defined, and it can have voltage dependency. Thus, the accuracy and linearity of the UGBW of the Gm-C integrator can be affected if the parasitic capacitance is significant in the total integration capacitance. For all the above three implementations, the amplifiers' DC

gain and the gain in the signal BW are limited. The R and C coefficients need to be chosen considering the limited GBW of the amplifier and the parasitic poles [57]. Although the amplitude and phase errors caused by the limited GBW can be partially compensated, for the robustness of the SDMs, commonly a GBW of at least $1.5 \times$ of the ADC sampling frequency F_S is required for the amplifiers in the RC integrators, when this amplifier is not in a speed-critical loop (e.g. the ELD compensation loop) [58]. Notice that a much higher GBW of the amplifier can be requested by the other functionality of the amplifier and by the applications. For example, the amplifier in the RC integrator in [16] is used as the summation amplifier in the ELD compensation loop as well (see Section 4.5), and thus an effective GBW about $6.9 \times F_S$ is designed. In the bandpass $\Sigma\Delta$ ADC reported in [34], an effective GBW of $62.5 \times F_S$ is implemented for one amplifier used in RC integrator to meet the linearity target and to ensure the coefficient accuracy. For state-of-the-art broadband medium-to-high resolution CT $\Sigma\Delta$ ADCs, the sampling rate is up to 8 GHz [16]. Thus, the GBW requirement of the amplifiers is commonly much higher than the narrow-band CT $\Sigma\Delta$ ADCs. To achieve high GBW, amplifiers are generally more complex and power consuming. For example, a fifth-order multistage multipath feed-forward op-amp is reported to achieve 250 GHz effective GBW which consumes 100 mW power [34].

The noise from the loop filter contributes to the total input-referred noise of the $\Sigma\Delta$ ADC. Generally the input-referred loop filter noise is dominated by the input stage of the loop filter, since the noise from the following stages is suppressed by the gain of the front-end stages when referred to the SDM input. Loop filter noise contains mainly two parts - the noise from resistors and the noise from the amplifiers. The amplifier noise has two contributions – thermal noise and flicker noise. To reduce the thermal noise, the transconductance g_m of the input transistor should be increased. To reduce the flicker noise, the gate area needs to be increased [59]. Notice that PMOS transistors generally feature less flicker noise than NMOS transistors with the same gate area [60]. The noise requirement makes the loop filter design in broadband $\Sigma\Delta$ ADCs challenging for the following reasons. Firstly, the same SNDR target means lower in-band noise density for broadband ADCs than narrow-band ADCs. To achieve the overall noise density, the input resistance can be lower than 1 k Ω [9, 16]. The output current of the feedback DAC is inversely proportional to the input resistance, which is higher as well. The loop filter amplifier needs to provide the current to charge the integration capacitor. Secondly, to achieve the overall noise density, the thermal noise and flicker noise from the amplifier should be low enough. Thus, a high q_m is required, which means high current consumption. To reduce the flicker noise, large gate area is needed, which is contradictory to the high GBW requirement discussed in the last paragraph.

The non-linearity generated by the loop filter contributes to the overall non-linearity of the $\Sigma\Delta$ ADC. Similar to the noise, the input referred non-linearity of the loop filter is generally dominated by the input stage, when OSR is high (≥ 16). In an RC integrator or a single-opamp resonator, the amplifier is used in a feedback loop, and the input swing of the amplifier is equal to the integrator/resonator output swing divided by

the gain of the RC integrator or single-opamp resonator. On the contrary, in a Gm-C integrator, the amplifier input swing is equal to the input swing of the integrator. Generally the amplifier in a Gm-C integrator sees a much larger input swing than the amplifier in an RC integrator or a single-opamp resonator, and thus the achievable linearity of a Gm-C integrator is worse than that of an RC integrator or single-opamp resonator. Moreover, for all the above three implementations, the output swing of the amplifier is equal to the output swing of the integrator or resonator. In the $\Sigma\Delta$ ADC architecture design, the UGBW of the integrators need to be carefully scaled, such that the output swing of the integrators and/or resonators is optimal. If the output swing of the integrator and/or resonator is too high, the non-linearity of the amplifier can limit the SNDR and/or spurious free dynamic range (SFDR) specification of the ADC. If the output swing of the integrator and/or resonator is too low, the noise and distortion from the following stages might not be sufficiently suppressed by this integrator/resonator. The non-linearity is a challenge for broadband $\Sigma\Delta$ ADCs. Firstly, to suppress the input swing of the amplifier used in an RC integrator or a single-opamp resonator, high DC gain is required for the amplifier. It is challenging to achieve high DC gain and high GBW at the same time, and commonly multi-stage and/or multi-path amplifiers are needed [30, 9, 34]. Secondly, broadband $\Sigma\Delta$ ADCs commonly have a lower OSR than narrow-band ADCs. In some cases, the UGBW of the 1^{st} integrator is comparable or lower than the signal BW, and thus the noise and non-linearity of the 2^{nd} integrator are significant when referred to the SDM input. In this case, the decision on the 1^{st} integrator output swing has direct impact on the overall noise, linearity and power consumption of the $\Sigma\Delta$ ADC.

Although Figure 4.1b and 4.1c are drawn single-ended, generally loop filters are implemented differentially. Therefore, the common-mode stability is another important aspect when designing the loop filter. Generally common-mode feedback (CMFB) circuitry is designed in the amplifier to keep its input and output common-mode voltages stable. The CMFB circuitry should be robust to have enough phase margin, gain margin and UGBW to handle the possible common-mode disturbance. In broadband $\Sigma\Delta$ ADCs, the requirement on the UGBW of the CMFB circuit is generally wider than that in narrow-band $\Sigma\Delta$ ADCs.

4.2 Quantizer non-idealities

Figure 4.3a shows the output characteristic of an ideal 1-bit comparator. Its output characteristic is:

$$V_{out} = \begin{cases} 1 & V_{in} \ge 0\\ -1 & V_{in} < 0 \end{cases}$$
(4.8)

Figure 4.3b presents the output characteristic of a 1-bit comparator with an offset



Figure 4.3: Output characteristic of (a) an ideal comparator (b) a comparator with offset V_{os} (c) a comparator with limited quantization gain G_q and (d) a comparator with hysteresis V_h .

 V_{os} . Its output characteristic can be written as:

$$V_{out} = \begin{cases} 1 & V_{in} \ge V_{os} \\ -1 & V_{in} < V_{os} \end{cases}$$
(4.9)

Generally a $\Sigma\Delta$ ADC with 1-bit quantization is insensitive to comparator offset, which only slightly increases the power of the unshaped quantization noise. The requirement for the mismatch in offsets of the various comparators in a multibit quantizer is generally more stringent than a 1-bit quantizer because the LSB size is smaller. Offset in multibit quantizers leads to extra non-linearity. Since the quantization noise power is the integral of the square of the quantization error by its probability, the comparator offset in a multi-bit $\Sigma\Delta$ ADC increases the original quantization noise power (see Chapter 10.4). Thus, it can make the SDM overloaded. The offset requirement is determined by the noise and linearity specifications of the $\Sigma\Delta$ ADC. The comparator offset can be caused by the mismatch in the comparator circuit. Moreover, for a multibit quantizer, the mismatch of the references contributes to the total quantizer offset. The design requirements of the comparator offset can be achieved by either intrinsic matching or comparator offset calibration [16]. Intrinsic matching requires large gate area for the matched transistors, which reduces the speed. In the design of a 3-bit quantizer in this thesis (shown in Chapter 11), the simulated standard deviation of a comparator in the 3-bit quantizer is about $\sigma = 2.6$ LSB, while the offset requirement is $\sigma \leq \frac{1}{6}$ LSB. A proposed comparator offset calibration technique for a multi-bit quantizer with current-mode multi-path excess loop delay compensation and local time-interleaved comparators is presented in Chapter 8.2.

Quantizer metastability is another limitation for broadband CT $\Sigma\Delta$ ADCs [61]. It is caused by the limited quantization gain G_q of the comparator, see Figure 4.3c. Assuming a comparator built from cross-coupled inverters with positive feedback, the quantization gain can be calculated as [62]:

$$G_q = e^{\frac{T_{reg}}{\tau}} \tag{4.10}$$

In this equation, T_{reg} is the regeneration time of the comparator, and τ is the regeneration time constant of the comparator. For the current state-of-the-art very-broadband (BW \geq 100 MHz) CT $\Sigma\Delta$ ADCs, the sampling frequency can be as high as 8 GHz [16]. In this case, one sampling clock period (T_s) is only 125 ps. An SDM is a feedback system, and typically the total delay in the feedback loop is kept below 1 T_S using excess loop delay compensation. This is the upper limitation of the total delay in the feedback loop, including quantizer regeneration time, propagation delay from quantizer to DACs and the DACs' delay. Thus, the quantizer regeneration time is typically much lower than T_s . However, the minimal regeneration time constant τ of the comparator is limited by the technology and by the power budget. Thus, the total quantization gain G_q generated by the comparator within the available regeneration time is limited. As a consequence, the comparator might not be able to make a good decision within the time available. This is especially a problem for GHz-rate sampling $CT \Sigma\Delta$ ADCs. Figure 4.3c shows a simplified output characteristic for a comparator with limited quantization gain G_q . For simplicity, this thesis assumes a straight line between $\left(-\frac{1}{G_q}, -1\right)$ and $\left(\frac{1}{G_q}, 1\right)$. In reality, exponentially asymptotically approaching to $V_{out} = 1$ and $V_{out} = -1$ is usually observed. The simplified output characteristic in Figure 4.3c can be written as:

$$V_{out} = \begin{cases} 1 & V_{in} \ge \frac{1}{G_q} \\ G_q \cdot V_{in} & -\frac{1}{G_q} \le V_{in} < \frac{1}{G_q} \\ -1 & V_{in} < -\frac{1}{G_q} \end{cases}$$
(4.11)

The metastability error in CT $\Sigma\Delta$ ADCs is further elaborated in detail in Chapter 10. Two solutions are proposed to overcome the metastability error limitation in this thesis – (1) metastability error compensation (Chapter 10.4) and (2) metastability shaping technique (Chapter 10.5).

Hysteresis is another non-ideality of the quantizer. It describes the memory effect of the comparator – the comparator output does not only depend on the input, but

also on its previous output, e.g. by a non-ideal reset of the quantizer. Figure 4.3d presents the output characteristic of a 1-bit comparator with hysteresis V_h . Its output characteristic can be written as:

$$V_{out}(n) = \begin{cases} 1 & V_{in}(n) \ge V_h \text{ or } \left(V_{in}(n) \ge -V_h \text{ and } V_{out}(n-1) = 1 \right) \\ -1 & V_{in}(n) < -V_h \text{ or } \left(V_{in}(n) < V_h \text{ and } V_{out}(n-1) = -1 \right) \end{cases}$$
(4.12)

The quantizer hysteresis can make the SDM unstable. In case the quantizer hysteresis is due to a non-ideal reset, it can be reduced by improving the reset, such that a lower residue is left at the input. The hysteresis problem is especially a challenge for high sampling rate quantizers, so for broadband $\Sigma\Delta$ ADCs.

Quantizers have thermal noise and flicker noise. Generally the SDMs are not sensitive to the thermal noise and flicker noise from the quantizers because their power is much less than the quantization noise and they are shaped by the NTF.

This section has studied some quantizer non-idealities. Some other non-idealities, such as the sampling aperture of the quantizer, are out of the scope of this thesis. This section shows quantizer offset, limited quantization gain, hysteresis and noise separately with figures and equations. In real circuit, these non-idealities are combined.

4.3 DAC non-idealities

This section discusses the non-idealities of the feedback DACs. Some CT SDM architectures have more than one feedback DAC, such as the exemplary 2^{nd} order CT SDM in feedback form shown in Figure 4.4. Here the feedback DACs have their output connected to the input of an integrator or a resonator of the loop filter. In this thesis, the feedback DAC whose output is connected to the input of the CT SDM is called the 1^{st} DAC, such as DAC₁ in Figure 4.4. The noise and distortion originating from the 1^{st} DAC is directly added to the SDM input without noise shaping. The noise and distortion originating from DAC₂ is 1^{st} order high-pass filtered in the signal band when referred to SDM input.

In CT $\Sigma\Delta$ ADCs, the main FB DACs are mainly implemented as current-steering DACs or resistive DACs. Figure 4.5 shows some common circuit implementations of a 1-bit DAC, or a unit cell of a multibit DAC. Figure 4.5a shows the simplified schematic of a current-steering DAC with one-side switching [30]. Here the upper side current source I_{DAC} is switched according to the DAC input $D^{+/-}$. The bottom side current sources $0.5I_{DAC}$ are static and always connected to $Out^{+/-}$. It can be implemented as bottom side switching and upper side static as well.

Figure 4.5b presents the simplified schematic of a current-steering DAC with two-side switching [16, 63]. Here both the upper current source and the bottom current source



Figure 4.4: An exemplary 2^{nd} order CT SDM in feedback form.

are switched. Sometimes this implementation requires a negative supply voltage $V_{supply,n}$ [16].

Figure 4.5c shows the simplified schematic of a resistive DAC [9]. It requires the common-mode output voltage of the DAC to be in the middle of the DAC supplies:

$$V_{cm,out} = \frac{V_{dd} + V_{ss}}{2}$$
 [V] (4.13)

Based on the noise mechanism, DAC noise contains two contributions – thermal noise and flicker noise. The thermal noise is contributed by both resistors and transistors, while the flicker noise is contributed mainly by the transistors. For current-steering DACs in Figure 4.5a and Figure 4.5b, DAC noise is contributed mainly by the DAC current sources (I_{DAC} and $0.5I_{DAC}$) and the biasing circuit which generates the biasing voltages/currents for the DAC current sources. Notice that in the current-steering DAC with one-side switching (Figure 4.5a), the non-switching current sources $0.5I_{DAC}$ contribute SDM input-referred noise, too. In that perspective, the current-steering DAC with two-side switching. For the resistive DAC in Figure 4.5c, DAC noise is contributed mainly by the DAC resistor R_{DAC} and the noise from the DAC supply V_{dd} . A low-noise LDO regulator is generally required to regulate the supply for the resistive DAC. Since the same SNDR target means lower in-band noise density for broadband ADCs than narrow-band ADCs, it is more challenging to achieve the noise specification of the DACs for broadband ADCs.

Several mechanisms can generate non-linearity in the DAC. For multi-bit DACs, mismatch between the unit DAC cells generates non-linearity [64]. The mismatch introduces both amplitude error and timing error. For current-steering DACs shown in Figure 4.5a and Figure 4.5b, the mismatch on the switching current sources contributes to the DAC non-linearity directly. The mismatch on the non-switching current sources leads in first instance to a common-mode output current but not to a non-linearity. Considering some 2^{nd} order effect, it might introduce some non-linearity as well. Typically its non-linear effect is not dominant in the CT SDM. When DAC



Figure 4.5: Simplified schematic of (a) a current-steering DAC with one-side switching (b) a current-steering DAC with two-side switching (c) a resistive DAC.

unit cells are mismatched, the DAC transfer function is not linear anymore. It has two types of impact in the content of $\Sigma\Delta$ ADCs – (1) the DAC non-linear transfer function mixes with the signal and interference, and generates harmonics and intermodulation distortion; (2) the DAC non-linear transfer function mixes with the out-of-band quantization noise which increases the in-band noise floor. The matching requirement should be considered in the sizing of the devices and in the layout. If intrinsic matching does not meet the requirements, DAC calibration techniques can be applied to improve the DAC linearity or to reduce the area [16, 65, 66, 67].

DAC non-linearity can be generated from the limited signal-dependent output impedance of the DAC together with the signal components at the DAC output node [63]. Adding a cascode transistor can increase the DAC output impedance, and thus decrease its relative influence with respect to the load impedance [30]. The error charge caused by the limited DAC output impedance can be compensated [63]. For a resistive DAC shown in Figure 4.5c, the output impedance of the DAC is almost equal to R_{DAC} . To achieve high linearity with a resistive DAC, the DAC supply V_{dd} should be regulated by a low impedance LDO regulator, and the swing on the DAC output node $Out^{+/-}$ should be minimized [9].

Signal dependent switching timing errors can degrade the DAC linearity as well. In the current steering DACs shown in Figure 4.5a and 4.5b, DAC switching can generate a glitch on the nodes V_s , V_{s1} and V_{s2} . If this glitch is not settled before the next switching moment, it will introduce a signal dependent timing error. To limit this error, the crossing point of D⁺ and D⁻ can be designed such that the glitch on these nodes are minimized. Moreover, the parasitic capacitance on the nodes V_s , V_{s1} and V_{s2} should be minimized to speed up the settling.

It is more challenging to achieve the same target linearity in dB for DACs in broadband $\Sigma\Delta$ ADCs than those in narrow-band $\Sigma\Delta$ ADCs for the following reasons. Broadband $\Sigma\Delta$ ADCs have a wider signal bandwidth and commonly a higher sampling rate. The DACs needs to deliver the required linearity for a bandwidth from DC to $\frac{F_S}{2}$, which is larger in broadband $\Sigma\Delta$ ADCs. At low frequency, the DAC linearity is commonly limited by the amplitude error. While at higher frequency, the timing error and the impact of the limited output impedance is more significant for the overall linearity performance [64]. Moreover, in broadband $\Sigma\Delta$ ADCs the available timing budget for the DAC delay is very limited. The matching requirement and the speed requirement are commonly contradictory. In many broadband $\Sigma\Delta$ ADC designs, the intrinsic matching is not enough, and thus DAC calibration is necessary [16, 65, 66, 67].

4.4 Clock non-idealities

The clock has various non-idealities, including clock jitter, clock skew, and duty cycle error. Here, clock jitter refers to the random noise on the timing of the rising and falling edges of the clock. It is generated from the thermal and flicker noise of the switching transistors in the clock generation and distribution circuit. Generally, CT $\Sigma\Delta$ ADCs are sensitive to the clock jitter on the DAC clock, while the clock jitter introduced error from the quantizer clock is shaped by the NTF [61]. The clock jitter introduced error generated at the 1st DAC whose output is connected to the SDM input is directly added to the SDM input without any noise shaping. The clock jitter introduced error generated at the DACs is influenced by the DAC pulse and number of bits. Generally, a return-to-zero (RZ) DAC with a rectangular DAC pulse is more sensitive to clock jitter than a non-return-to-zero (NRZ) DAC with a rectangular DAC pulse. Figure 4.6 shows the clock jitter introduced error in the feedback DAC output current for a 1-bit and a 2-bit NRZ DACs. A multi-bit NRZ DAC with a rectangular DAC pulse is less sensitive to clock jitter than a 1-bit NRZ DAC with a



Figure 4.6: Clock jitter introduced error in the feedback DAC output current (a) for a 1-bit NRZ DAC (b) for a 2-bit NRZ DAC. Exemplary bit sequence for illustrative purpose.

rectangular DAC pulse, without specific jitter mitigation techniques. Applying finite impulse response (FIR) DACs can make the 1-bit CT $\Sigma\Delta$ ADCs less sensitive to the clock jitter [68].

Clock skew refers to the deterministic timing error on the rising and falling clock edges compared to the ideal clock. In Section 4.3, the timing error caused by the mismatch among DAC unit cells introduces clock skew, causing DAC non-linearity. Here the timing difference due to the mismatch on the sampling and buffering of the DAC input, the switching time difference caused by the mismatch of the DAC switches, and the different RC delay of local clock routing all contribute to the effective clock skew between the unit DAC cells.

The clock can have a duty-cycle error, which refers to the effect that the real clock has a different duty-cycle than the ideal clock. Some system and circuit choices make the SDMs more sensitive to the duty-cycle error. If all the circuitry of the SDM is triggered by only one clock edge (either a rising edge or a falling edge) of the SDM fundamental input clock, the SDM in principle is not sensitive to the duty-cycle error of the SDM fundamental input clock. Some circuits use both clock edges, or are clock level sensitive. For example, for a RZ DAC which gives non-zero output when its clock is high and zero output when its clock is low, the clock duty-cycle error is transformed into a gain error in the total charge of the DAC feedback.

Broadband CT $\Sigma\Delta$ ADCs need sharp clock edges in term of absolute value as the clock period is short. For example, in the 7 GHz sampling $\Sigma\Delta$ ADC test chip shown in Chapter 11, one clock period is only 143 ps. The 10% to 90% rising edge and 90% to 10% falling edge are designed to be less than 20 ps for the 7 GHz clock. Clock edges have mainly two impacts. First, an amplitude error on the clock is converted to a time error through the slope of the clock edges. Second, slow clock edges introduce more delay in the $\Sigma\Delta$ loop.



Figure 4.7: An CT SDM with ELD compensation.

4.5 System-level non-idealities

Excess loop delay refers to the total delay of the SDM feedback loop, including the quantizer delay (mainly quantizer regeneration time), DAC delay and the propagation delay from the quantizer to the DAC [69]. As shown in Figure 4.7, ELD can be compensated typically by adding a direct feedback path from the output to the input of the quantizer through DAC_{ELD} [69, 70]. For broadband (BW ≥ 10 MHz) medium-to-high resolution (SNDR ≥ 50 dB) CT $\Sigma\Delta$ ADCs, typically ELD up to one sampling clock period is compensated, to restore the stability and SQNR to that of the non-delayed SDM.

ELD compensation is susceptible to implementation non-idealities. For example, the ELD summation commonly has a limited bandwidth, which means an effective parasitic pole is at the summation node output. Figure 4.8 shows an exemplary 1-bit single-loop SDM with ELD compensation and a parasitic pole at the ELD summation output. In the ideal model, the ELD compensation loop has a delay of 0.5 T_S , while the main feedback loop through DAC₁ has a delay of 1 T_S . The parasitic pole adds extra delay (or phase shift) in the loop. For an exemplary 4th order 1-bit SDM, simulation results show that the parasitic pole frequency at the ELD summation node should be higher than 2 F_S , to maintain the performance and the robustness of the SDM. In this thesis, an overview for the state-of-the-art high-speed ELD compensation techniques is given in Chapter 7.1. A proposed current mode multi-path ELD compensation technique is presented in Chapter 7.2. And a proposed technique to overcome the parasitic capacitance on the ELD summation node is addressed in Chapter 7.3.

An SDM architecture is defined by many coefficients, including the UGBW of the integrators, the gain and resonance frequency of the resonator, the linear gain of the quantizer, and the output swing of the DACs. The loop delay through every feedback DAC is another coefficient in time. These coefficients can have variations caused by two reasons -(1) process, voltage and temperature (PVT) variation and (2) mismatch.



Figure 4.8: An exemplary CT SDM with ELD compensation and a parasitic pole at the ELD summation output.

PVT variation has a common impact on the same type of devices on the chip. For example, due to PVT variation, the UGBW of an RC integrator can vary typically up to about \pm 30%. Usually a tunable capacitor bank is implemented for every RC integrator to calibrate the integrators' UGBW. On the other hand, mismatch causes different relative errors for similar devices on the chip [71]. Applying common-centroid layout and adding dummies are common ways to improve the matching. For more advanced CMOS technology, the layout dependent effect on the matching becomes more important. Since the coefficient variation is unavoidable, the SDM architecture should be designed robust enough to be able to tolerate a few percent of coefficient variation.

For CT MASH $\Sigma\Delta$ ADCs, digital noise cancellation filters are required to remove the quantization noise of all the stages except for the last stage (see Section 3.3.2). Perfect noise cancellation requires the digital noise cancellation filters and the corresponding analog filters to be matched, otherwise the quantization noise from the front-end stages is not perfectly canceled which can degrade the SNR of the MASH $\Sigma\Delta$ ADCs [42, 72, 73]. To reduce the quantization noise leakage, calibration of the digital NCF is typically required [42, 72, 73]. A proposed digital calibration technique for broadband CT MASH $\Sigma\Delta$ ADCs with relaxed filter requirements is presented in Chapter 6.

4.6 Supply and reference non-idealities

A stable supply and ground are essential for the resolution of broadband CT $\Sigma\Delta$ ADCs. Typically broadband $\Sigma\Delta$ ADCs require a low-impedance supply and ground over a wide bandwidth. A series resistance on the supply causes an IR drop, which reduces the speed, gain and bandwidth of some sensitive circuits. For example, a large IR drop can reduce the quantization gain of the comparator core, the GBW of the loop filter amplifier, and can increase the propagation delay of the buffers between the quantizer and the DAC. Moreover, noise and distortion can couple through the

supply or ground into the signal bandwidth, which can reduce the SNDR, DR or SFDR performance of the ADC. Broadband $\Sigma\Delta$ ADCs are more sensitive to the non-idealities of the supply and ground since their current consumption is typically higher than the narrow-band ADCs, and their in-band noise density is also lower than the narrow-band ADCs with the same DR. On-chip LDO regulators are required for sensitive supplies in some high-linearity broadband $\Sigma\Delta$ ADCs [9]. Sufficient on-chip decoupling capacitance should be designed with low routing resistance to the sensitive blocks which consumes dynamic current. Separating supplies between the sensitive blocks (e.g. input stage of the loop filter, 1st DAC, etc.) and the aggressors (e.g. quantizer) can be considered for improved isolation. Separation of ground can also be considered for improved isolation which makes the using of deep N-well necessary, and increases the layout area and complexity. Bonding wire inductance and resistance, and the supply and ground circuitry on the validation board should also be considered in the design and simulations.

CT $\Sigma\Delta$ ADCs commonly need some reference current and/or voltage. Non-idealities of the reference can degrade the ADC performance. For example, the ADC test chip in this thesis requires a reference current for the biasing block of its DAC (see Chapter 11). The noise of this reference current is copied to the DAC output current, and directly adds to the ADC input and it can degrade the DR of the ADC. Thus, a low noise reference current is needed. A bandgap circuit is typically needed to generate the reference voltage and/or current on chip in a commercial semiconductor product [74].

4.7 Limitations from CMOS technologies

To understand the speed-resolution limitations from CMOS technologies, this section further analyzes the state-of-the-art CT $\Sigma\Delta$ ADCs with Fs ≥ 1 GHz on their sampling rates, quantizer resolutions and technology nodes. Figure 4.9 shows the benchmark plot for those ADCs fabricated in 16 - 45 nm CMOS. The x-axis shows the sampling rate of the CT $\Sigma\Delta$ ADCs, and the y-axis shows their quantizer resolution. The ADCs are divided into 3 groups according to their fabrication technology: (1) 16 - 20 nm CMOS (2) 28 nm CMOS (3) 40 - 45 nm CMOS. From the benchmark it can be observed that at higher sampling rates, lesser bits are implemented in an ADC. The blue dotted line shows the state-of-the-art envelope for published 28 nm ADCs and the green dotted line shows the state-of-the-art line for 40 - 45 nm designs. From these lines it can be extracted that for the same number of bits, 28 nm CMOS technology achieves about $1.6 - 2 \times$ higher sampling rate compared to 40 nm CMOS thanks to the technology advantage. The transit frequencies (f_T) of MOS transistors of both 40 nm and 28 nm CMOS technologies are simulated. With the same current density, the NMOS transistor in the type with the lowest threshold voltage in 28 nm CMOS shows $1.6 \times$ higher f_T than that in 40 nm CMOS. In 28 nm CMOS technology smaller transistor structure can be used to achieve the same W/L



Figure 4.9: Quantizer resolution vs. sampling rate for state-of-the-art CT $\Sigma\Delta$ ADCs with Fs ≥ 1 GHz, fabricated in 16 – 45 nm CMOS (based on [4]).

as the 40 nm CMOS technology, which makes the layout more compact and reduces the routing parasitic resistance and capacitance. It is acknowledged here that the 28 nm state-of-the-art envelope only contains 2 data points. However, considering the better matching, higher speed and the miniaturization benefit of 28 nm CMOS, the $1.6 - 2 \times$ speed advantage compared to 40 nm CMOS is a realistic expectation. From this plot, it can be estimated that in the chosen 40 nm CMOS technology, the highest practically achievable sampling rate for 4-bit quantization is about 4 - 5 GHz, for 3-bit quantization is about 6 - 7 GHz, and for 2-bit quantization is about 8 GHz.

4.8 Design-space reduction

The non-idealities, challenges and limitations from circuit implementation justify limiting our design space for the SDM architecture design, with respect to choice of sampling rate F_S , NTF order, and quantizer resolution.

The maximal achievable ADC sampling rate F_S is limited by multiple factors. Firstly, commonly an RC integrator or a single-opamp resonator is used as the 1st stage of the loop filter to achieve high linearity. For the robustness of the $\Sigma\Delta$ ADC, the GBW of the amplifier used in this RC integrator or single-opamp resonator is typically at least $1.5 \times F_S$. Actually, all of the circuits must have a wider bandwidth when a higher F_S is used, otherwise the parasitic poles can degrade the performance or even make the SDM instable. If the F_S is very high, the GBW requirement of the amplifier is very high, thus the power consumption of the amplifier is very high which reduces the overall power efficiency of the ADC. Secondly, commonly ELD compensation techniques can compensate no more than 1 T_S delay in the SDM loop. When F_S is very high, e.g. 10 GHz, 1 T_S is only 100 ps. This is the total time budget for the quantizer regeneration time, the quantizer sampling delay, the propagation delay from the quantizer to DACs, and the DACs' delay. In this case, the quantizer hardly has enough regeneration time to generate the required quantization gain to combat the metastability error. Last but not least, the clock jitter introduced error is proportional to the relative clock jitter referred to the sampling clock period T_S . The higher F_S , the more stringent the absolute clock jitter requirement is. In 40 nm CMOS technology, considering multi-bit quantization, the estimated maximal achievable ADC sampling rate F_S is less than 10 GHz, which is mainly limited by the second point. On the other hand, the F_S should be at least 5 GHz such that the OSR is at least 5 to have sufficient in-band quantization noise suppression.

The maximal NTF order is limited by the circuit implementation non-idealities. To guarantee the stability for high-order low-pass SDMs with order larger or equal to 6, commonly the SQNR performance is sacrificed, which makes it less attractive. Moreover, for very low OSR between 5 and 10, the SQNR improvement is almost negligible for single-loop SDM when the NTF order is further increased above 5. Figure 3.8a shows that the state-of-the-art broadband CT $\Sigma\Delta$ ADCs mostly have noise shaping order less or equal to 5. Considering that the robustness of the $\Sigma\Delta$ ADC with coefficients variation and parasitic poles must be guaranteed, the maximal NTF order (N_O) is chosen to be less or equal to 5.

The choice on the quantizer resolution is limited by the circuit implementation challenges. Firstly, SDMs with multi-bit quantization are less sensitive to clock jitter compared to SDMs with 1-bit quantization. The higher limit on the quantizer resolution is bounded by two reasons. (1) It is bounded by the offset of a single comparator in a 1-bit quantizer, and especially the mismatch in offset between the various comparators in a multibit quantizer. For a quantizer with more bits, the offset requirement is more stringent, which requires larger area for matched devices and/or more sophisticated comparator offset calibration techniques. Notice that the larger device area and the additional circuit for offset calibration typically reduce the speed of the quantizer. (2) This thesis considers implementing the multi-bit quantizer as a flash sub-ADC, since the SDM sampling rate is about 5 GHz - 10 GHz to achieve the target signal BW and SNDR specification. The capacitive load caused by the flash quantizer increases exponentially with the number of bits. Figure 3.8 shows that the state-of-the-art broadband CT $\Sigma\Delta$ ADCs with flash quantizers mostly have no more than 17 quantization levels. Considering these limitations, the quantizer number of bits (N_Q) is limited to maximally about 4.09-bit (17-level).

At this step, the boundaries on the choice of the main design parameters are summarized in Table 4.1.

Sampling rate F_S	$5 \text{ GHz} \le F_S \le 10 \text{ GHz}$	
NTF order N_O	$N_O \le 5$	
Quantizer resolution N_Q	2-bit $\leq N_Q \leq 4.09$ -bit	

Table 4.1: Boundaries on the main SDM architecture design parameters

4.9 Conclusions

Various challenges, non-idealities and limitations exist in the circuit implementation of broadband CT $\Sigma\Delta$ ADCs. The loop filter is affected by limited GBW of the amplifiers, noise, and non-linearity. Moreover, the common-mode stability should be guaranteed by circuit design. The quantizer is affected by offset, limited quantization gain and hysteresis. The DACs contribute noise and non-linearity to the SDM as well. The clock has clock jitter, clock skew and duty-cycle error. Coefficients variation should be accommodated by the SDM architecture design, and the excess loop delay should be compensated properly in the SDM. For CT MASH $\Sigma\Delta$ ADCs, the mismatch between the digital NCF and the analog filter should be calibrated. Considering the limitations from circuit implementation, SDM sampling frequency between 5 GHz and 10 GHz, NTF order less or equal to 5 and multi-bit quantizer with number of bits no more than 4.09 seems to be feasible to be implemented in the chosen 40 nm CMOS technology.

Chapter 5

Synthesis procedure for broadband CT $\Sigma\Delta$ ADC architectures

This chapter presents a synthesis flow for broadband CT MASH $\Sigma\Delta$ modulators, on the basis of an exemplary ADC architecture for the target application defined earlier. The information from the top-down design approach shown in Chapter 3 and the bottom-up design approach shown in Chapter 4 is considered in this synthesis flow. Firstly, the choice of the main design parameters is made, including the SDM topology, NTF order, quantizer resolution and OSR, based on simulation results of ideal SDM architectures and the speed-resolution limitation of the chosen 40 nm CMOS technology. Afterwards, this chapter presents a generic synthesis flow for broadband CT MASH $\Sigma\Delta$ modulators. This chapter further shows the additional system-level optimization of an exemplary CT MASH $\Sigma\Delta$ modulator for the target application. Various system-level verification results are presented to set specifications for the building blocks and to show the feasibility and robustness of the ADC architecture. In the end, conclusions are drawn.

5.1 Choice of main design parameters

The fundamental limitations for the bandwidth and resolution of CT $\Sigma\Delta$ ADCs are the resolution (or amplitude) limitation and the speed (or bandwidth) limitation, which are given by the technology. Although the device mismatch of the technology can be overcome with calibration, the resolution is still limited by the calibration accuracy, and the calibration circuitry adds to the load which reduces the speed and increases power consumption. The SQNR and BW of a single-loop SDM is limited mainly by three parameters: (1) the gain of the loop filter, (2) the quantizer resolution, and (3) the sampling rate. The gain of the loop filter in the signal bandwidth is determined by the loop filter order, OSR, OBG and zero optimization. Its accuracy is limited by the GBW of the amplifier and the accuracy of the coefficients in the
implementation. The quantizer resolution is limited by both the amplitude accuracy and the speed limitation. The mismatch of the comparator offset is limited by the amplitude accuracy, and the quantization gain is limited by the speed limitation of the technology. The sampling rate of the CT SDM is limited mainly by the speed limitation of the technology. Because all the three parameters (the loop filter gain, the quantizer resolution and the sampling rate) are limited by the resolution and speed limitation of the technology, the signal BW is limited for a certain SQNR target for single-loop SDMs. MASH SDMs benefit from the cancellation of the quantization noise of all front-end stages, which can alleviate the signal BW limitation of the single-loop SDMs. However, the cancellation of the quantization noise relies on the matching between the transfer functions, which is limited by both the amplitude accuracy and speed/bandwidth limitation of the technology.

The analysis in Chapter 3 and 4 concludes that for the target specification, the design space is reduced to OSR between 5 and 10, NTF order less or equal to 5, quantizer number of bits between 2 and 4.09 (17-level), and either a single-loop or MASH structure. This thesis chooses several representative single-loop and MASH SDM architectures with different order and quantizer resolution. Here single-loop SDMs up to 4^{th} order are simulated. For very low OSR between 5 and 10, 5^{th} order single-loop SDMs do not improve much the peak SQNR compared to 4^{th} order singleloop SDMs, while increase the design complexity [19]. Several MASH structures up to 3 stages are considered here, since in practice an additional 4^{th} stage can hardly increase the resolution of the MASH ADC due to implementation non-idealities. For example, due to the limited GBW of the loop filter amplifier, variation of MASH SDM parameters caused by device mismatch, and parasitic poles in the loop, the actual transfer functions in a MASH architecture deviate from the ideal transfer functions. This error commonly overloads the 4^{th} stage, which does not improve the resolution of the MASH anymore. The ideal DT SDM models are optimized for different OSR between 5 and 10 and simulated in Matlab. The peak SQNR vs. OSR is plotted in Figure 5.1. Here, zero optimization is used to further improve the peak SQNR. In MASH SDMs, higher resolution quantizers in the front-end stages allow to use higher inter-stage gains. And higher resolution quantizer in the last stage can reduce the original quantization noise power which is suppressed by the overall noise shaping of the MASH. In practice, the quantizer in the 1^{st} stage has more stringent quantization gain requirement. It implies that if the quantizer in the 1^{st} stage is feasible to be implemented, then quantizers with the same resolution can be implemented in the back-end stages as well. Considering that the quantizer in the 1^{st} stage can be reused or scaled in the back-end stages to save design effort, all the quantizers of the MASH have the same resolution in this simulation.

Figure 5.1 shows that for the same SDM architecture, the peak SQNR increases when the OSR increases. Some SDM architectures show similar SQNR versus OSR, for example 17-level 3^{rd} order single-loop SDM and 9-level 4^{th} order single-loop SDM. However, the peak SQNR of 9-level 4^{th} order single-loop SDM increases faster when the OSR increases, compared to the 17-level 3^{rd} order single-loop SDM, because higher



Figure 5.1: Simulated peak SQNR vs. OSR for different SDM architectures.

order NTF benefits more from increasing the OSR (see equation (5.1)). Another observation is that for OSR = 7, the peak SQNR of 9-level/17-level $3^{rd}/4^{th}$ order single loop SDMs is limited to about 79 dB. For the same OSR, an ideal 9-level 2-1-1 MASH SDM can achieve a peak SQNR of about 111 dB. Comparing the 9-level 2-1-1 MASH SDM with the 9-level 4^{th} order single loop SDM, the original unshaped quantization noise power and the noise shaping order are the same. However, the 9-level 2-1-1 MASH SDM achieves a much higher peak SQNR of 111 dB than the peak SQNR of 68 dB for the 9-level 4^{th} order single loop SDM. The huge benefit of peak SQNR is caused by two advantages of MASH SDM: (1) 2-1-1 MASH SDM is built from 1^{st} order and 2^{nd} order SDM stages, which allow to use higher OBG and more aggressive noise shaping than 4^{th} order SDM. For example, assuming the SDM architectures are optimized with MSA of at least 0.7 $V_{FS,DAC}$ ($V_{FS,DAC}$ is the fullscale equivalent voltage from the DAC feedback), the maximal OBG for a 9-level 4^{th} order single loop SDM is 5, while the maximal OBG for a 9-level 2-1-1 MASH SDM is 16. (2) In 9-level 2-1-1 MASH SDM, inter-stage gains of $a_{12} = a_{23} = 8$ are used which suppresses the quantization noise of the 3^{rd} stage by 36.1 dB. Notice that in practice the inter-stage gains have to be reduced to create margin, otherwise the 2^{nd} and 3^{rd} stages would become overloaded by the unavoidable circuit implementation non-idealities, such as the phase and gain inaccuracy of the loop transfer functions. Thus, in practice the peak SQNR difference between single-loop and MASH SDMs is smaller than that in Figure 5.1. From this analysis, a general conclusion can be drawn regardless of the technology nodes of the implementation: MASH SDMs can achieve a certain BW and DR target with lower OSR than single-loop SDMs for the same quantizer resolution. In a certain technology, for a certain DR target, MASH

N _Q N _O	2-level	4-level	5-level	8-level	9-level	17-level
1	-7.1	-7.1	-6.4	-5.1	-6	-5.3
2	-21.2	-14.2	-11	-11.2	-9.5	-8.6
3	-37.3	-23.3	-21.7	-17.3	-16.9	-14
4	-57.4	-36.3	-36.5	-27.3	-26.3	-21.4

Table 5.1: Empirical value of C_{N_Q,N_Q} in [dB] assuming 1-tone sine input.

The values for $N_Q = 2$, 4, 8 level are extracted from Figure 4.14 - 4.16 in [19]. The values for $N_Q = 5$, 9, 17 level come from simulations with ideal SDM models in Matlab. The Delta Sigma toolbox is used [37].

SDMs can achieve broader BW than single-loop SDMs.

In the literature, empirical SQNR limits versus OSR plots can be found for single-loop SDMs with different NTF order and quantizer resolution [19]. In [75], two formulas are given for the $SQNR_{max}$ for 1st order and the 2nd order SDMs over the quantizer resolution and OSR. In the following, this thesis generalizes the simulation results in Figure 5.1 and proposes generic formulas to calculate the SQNR limits for both single-loop and MASH SDMs. A fitting parameter in these formulas is derived empirically. For single-loop SDMs, the SQNR limit can be calculated as:

$$SQNR_{max} = C_{N_O,N_Q} + 6.02 \cdot \left(N_Q + (N_O + 0.5) \cdot log_2(OSR)\right)$$
 [dB] (5.1)

In this formula, the term $6.02 \cdot \left(N_Q + (N_O + 0.5) \cdot \log_2(OSR) \right)$ is taken from literature [76], and this thesis proposes that the fitting constant C_{N_O,N_Q} is determined by both N_Q and N_O . Here, N_Q is the quantizer resolution in bit, and N_O is the NTF order. In this formula, the term $6.02 \cdot N_Q$ indicates the impact of the original quantization noise power without the noise shaping. The term $6.02 \cdot (N_O + 0.5) \cdot log_2(OSR)$ shows the impact of the N_O^{th} order noise shaping with OSR on the in-band quantization noise power. C_{N_O,N_O} is a fitting constant in dB which is influenced by four parameters: (1) OBG (2) the input power which is linked to the $SQNR_{max}$ (3) zero optimization for $N_O \geq 2$ (4) the input signal waveform. Assuming 1-tone sine input, C_{N_O,N_Q} is a constant that can be optimized for a certain NTF order and quantizer resolution. Figure 3.4 (Chapter 3.2) shows that for the same NTF order (3 in this figure), the OBG for the peak SQNR increases when the quantizer resolution increases. For the same NTF order and OBG, the corresponding input amplitude A_{in} increases when the quantizer resolution increases. The empirical value of C_{N_O,N_Q} is presented in Table 5.1. In this table, the values of C_{N_O,N_Q} are corresponding to the maximal achievable SQNR of ideal SDMs for a certain NTF order and a certain quantizer resolution. OBG and A_{in} are different for the values of C_{N_O,N_O} in this table.

Figure 5.2a shows the comparison between the simulated $SQNR_{max}$ and the calcu-



Figure 5.2: Comparison between simulated and calculated SQNR_{max} versus OSR: (a) for 3^{rd} order and 4^{th} order single-loop SDM structures; (b) for 2-2 MASH structure; (c) for 1-1-1 MASH structure (d) for 2-1-1 MASH structure.

lated $SQNR_{max}$ with equation (5.1), for 3^{rd} order and 4^{th} order single-loop SDM architectures. The calculated $SQNR_{max}$ matches very well with the simulated $SQNR_{max}$, with a maximal error of 0.6 dB. It proves that the $6.02 \cdot (N_O + 0.5) \cdot log_2(OSR)$ term predicts the trend of $SQNR_{max}$ versus OSR, which is in some textbooks on SDMs as well [76]. By cross-checking the formula (5.1) with our simulation results and the literature [19], the accuracy of the formula (5.1) with the constant values in Table 5.1 is about ± 1 dB for the concerned single-loop SDMs (N_O between 1 and 4, N_Q between 1-bit and 17-level) with OSR between 5 and 16. Notice that in [75], the fitting constant only depends on the NTF order, but not on the quantizer resolution. In our formula (5.1), the fitting constant C_{N_O,N_Q} depends on both of the NTF order and the quantizer resolution, which is more accurate than [75] especially for NTF order $N_O \geq 2$.

As a next step, this thesis generalizes the results to MASH SDMs. Chapter 3.3.2

has shown the overall noise transfer function of a 2-stage MASH. The overall noise transfer function of a M-stage MASH can be generalized as:

$$NTF_{tot}(z) = \frac{1}{\prod_{i=1}^{M-1} a_{i,i+1}} \cdot \prod_{i=1}^{M} NTF_i(z)$$
(5.2)

Here, $a_{i,i+1}$ is the inter-stage gain from the ith stage to the $(i+1)^{th}$ stage. $NTF_i(z)$ is the noise transfer function of the ith stage.

For an M-stage MASH SDM, this thesis proposes the following empirical formula to calculate the SQNR limit:

$$SQNR_{max} = \sum_{i=1}^{M} C_{N_{O,i},N_{Q,i}} + 6.02 \cdot \left(N_{Q,M} + \sum_{i=1}^{M-1} log_2 a_{i,i+1} + \left(\sum_{i=1}^{M} N_{O,i} + 0.5\right) \cdot log_2(OSR)\right) \quad [dB]$$
(5.3)

Here, $N_{O,i}$ and $N_{Q,i}$ are the NTF order and the quantizer resolution in bit of the ith stage, i = 1, ..., M. $C_{N_{O,i},N_{Q,i}}$ is the corresponding constant of the ith stage. For the MASH structures in Figure 5.1, with a 9-level quantizer in the front-end stage, the ISG from this stage to the following stage is 8. And with a 5-level quantizer in the front-end stage, the ISG from this stage to the following stage is 4. In this formula, the term $(C_{N_{O,i},N_{Q,i}} + 6.02 \cdot (N_{O,i} + 0.5) \cdot log_2(OSR))$ shows the impact of $NTF_i(z)$ on the overall $SQNR_{max}$ of the MASH. The term $6.02 \cdot log_2a_{i,i+1}$ shows the impact of the ISG from the i^{th} stage to the $(i + 1)^{th}$ stage on the overall $SQNR_{max}$ of the MASH. Notice that both $NTF_i(z)$ and $a_{i,i+1}$ are in the overall NTF of the MASH in equation (5.2). The term $6.02 \cdot N_{Q,M}$ is because only the quantization noise of the last stage is in the final MASH output shaped by $NTF_{tot}(z)$, and the quantization noise of all front-end stages are canceled completely in the ideal MASH.

Figure 5.2b - 5.2d show the comparison between the simulated $SQNR_{max}$ and the calculated $SQNR_{max}$ with equation (5.3), for 3 MASH architectures. The constants C_{N_O,N_Q} extracted from the simulations of single-loop SDMs in Table 5.1 are used here. In Figure 5.2b - 5.2d, the maximal difference between calculated $SQNR_{max}$ and simulated $SQNR_{max}$ is 1.9 dB. By cross-checking the formula (5.3) with our simulation results, the accuracy of the formula (5.3) with the constant values in Table 5.1 is about \pm 2 dB for the concerned MASH SDMs with OSR between 5 and 10.

The formulas (5.1) and (5.3) greatly simplify the assessment of the maximum achievable SQNR for various SDM architectures with different number of stages (M); different NTF order (N_O) and quantization resolution (N_Q) for these stages; the sequencing of the stages; and OSR. Although in Figure 5.1 all the quantizers of the MASH have the same resolution, the formula (5.3) can be used for the cases that the quantizers of the MASH have different resolutions. Moreover, this empirically derived generalization of the problem provides insights which are useful for $\Sigma\Delta$ ADC architecture design. Firstly, the formula (5.3) implies that the $SQNR_{max}$ does not depend on the exact sequence of the MASH structures. For example, according to this formula, a 3-bit 2-1 MASH SDM and a 3-bit 1-2 MASH SDM have the same $SQNR_{max}$. Moreover, these two formulas give insights when comparing different MASH and/or single-loop structures. For example, if the 9-level 2-2 MASH structure is compared to the 9-level 2-1-1 MASH structure, their $\sum_{i=1}^{M} C_{N_{Q,i},N_{Q,i}}$ values are very close (-19 dB for 2-2 MASH and -21.5 dB for 2-1-1 MASH), and it can be concluded that their $SQNR_{max}$ difference is mainly due to the fact that the 9-level 2-1-1 MASH has an extra ISG a_{23} which has a value of 8. Another example is to compare a 9-level 4^{th} order single-loop SDM to a 9-level 2-1-1 MASH structure. Their total NTF order is the same. The quantizer resolution of the last stage in 2-1-1 MASH is the same as the 4^{th} order single-loop SDM. Their $SQNR_{max}$ difference comes from two parts: (1) For the 9-level 4^{th} order single-loop SDM, $C_{N_O,N_Q} = -26.3$ dB; while for the 9-level 2-1-1 MASH structure, $\sum_{i=1}^{3} C_{N_{O,i},N_{Q,i}} = -21.5$ dB. (2) The effect of the inter-stage gains, represented by the term $6.02 \cdot \sum_{i=1}^{2} log_2 a_{i,i+1}$ boosts the $SQNR_{max}$ of the 9level 2-1-1 MASH structure by 36.1 dB, which is not in the formula for the 9-level 4^{th} order single-loop SDM.

With equations (5.1) and (5.3), the $SQNR_{max}$ of various single-loop and MASH structures can be calculated with different NTF order, quantizer resolution, OSR and MASH stages. For example, a 1-2 MASH SDM with OSR = 8.6 and two 17-level quantizers is reported in [16]. For this architecture, it can be calculated that the $SQNR_{max}$ is about 100 dB, assuming its maximal $a_{12} = 16$ for the 17-level quantizer.

For a certain bandwidth and resolution (SNDR or DR) target in a certain technology, an optimal OSR can be chosen to minimize the power consumption of the CT $\Sigma\Delta$ ADC. If the OSR is too high, an unnecessary high sampling rate leads to stringent requirements on the quantizer core regeneration time constant, the loop filter amplifier GBW, and the propagation delay from quantizer to DACs. So, it leads to unnecessarily high power consumption and design efforts. On the other hand, if the OSR is too low, to achieve the target SQNR, higher resolution quantizer and/or higherorder loop filter are needed [19]. For a flash-ADC-based quantizer, the total number of comparators is almost doubled when the quantizer resolution increases by 1-bit, and the comparator offset requirement is tighter. A higher-order loop filter generally requires more amplifiers, and it is more sensitive to parasitic poles and extra delay in the loop. Thus, both higher quantizer resolution and higher order loop filter lead to higher power consumption. The power consumption of the broadband CT $\Sigma\Delta$ ADCs is closely related to the circuit implementations as well.

Figure 5.1 shows the maximal achievable SQNR of ideal SDMs. However, in the real implementation, thermal noise, clock jitter, DAC non-linearity, loop filter amplifier limited GBW etc. all degrade the SNDR of the SDM. Thus a margin between the maximal SQNR of the ideal SDM and the SNDR target is necessary. To be able to

choose the optimal OSR, quantizer resolution, and order for the ADC test chip, this thesis considers the limitations from the implementation technology as well. Figure 4.9 has shown the speed-resolution limitation of the chosen 40 nm CMOS technology. Considering Figure 5.1, Figure 4.9 and the required margin between the maximal SQNR and the SNDR target, it is not very realistic to design a single-loop $\Sigma\Delta$ ADC achieving the target SNDR of 60 dB and BW of 500 MHz. Thus, this thesis decides to choose a MASH structure.

For the 1^{st} stage of the MASH, this thesis decides to design a 2^{nd} order SDM loop for the following reasons:

- 1st and 2nd order SDMs are more robust against parasitic poles and extra delay in the loop than higher order SDMs.
- A 2^{nd} order SDM stage allows to implement a pair of optimized zeros, which can largely improve the peak SQNR (see Chapter 3.2).
- With the sample quantizer resolution, 2^{nd} order SDM is less tonal than 1^{st} order SDM. Thus, dithering is not compulsory for a 2^{nd} order SDM for this application, which can simplify the overall system.
- For a 3-bit 1st order SDM, the MSA is about 1 $V_{FS,DAC}$. For a 3-bit 2nd order SDM, the MSA is about 0.9 $V_{FS,DAC}$. The difference in the signal power (1.8 dB) is not the dominant argument for this choice.

For the 3^{rd} reason, this thesis would like to give further comments. For a MASH structure with a 1^{st} order SDM as its 1^{st} stage, ideally the tones in the quantization noise of the 1^{st} stage is canceled by the MASH. However, a tonal 1^{st} order 1^{st} stage still introduces tones in the final MASH output in the following two cases: (1) When the back-end stages are overloaded, or when the analog transfer function is not ideally matched with the digital NCF, the tones in the quantization noise of the 1^{st} stage cannot be fully canceled. (2) If the tones from the quantization noise of the 1^{st} order 1^{st} stage cause some extra tones (with different phase, amplitude, or frequency) at the DAC or the loop filter because of the non-linearity of the DAC or the loop filter, those extra tones cannot be canceled by the MASH and they will be present in the final MASH output. In some published MASH ADCs with a 1^{st} order SDM as the 1^{st} stage, dithering is applied at the cost of the dynamic range reduction [16].

From the aforementioned analysis, this thesis has chosen a MASH structure with a 2^{nd} order SDM as its 1^{st} stage. From Figure 4.9 it is estimated that the maximal sampling rate with 3-bit quantization in 40 nm CMOS is about 6 - 7 GHz. For the back-end stages, this thesis chooses 1^{st} order SDMs because of its robustness and simplicity. Considering the peak SQNR potential, this thesis decides to design a 3-bit 2-1-1 MASH SDM test chip with OSR of 7 (thus $F_S = 7$ GHz). This thesis uses 3-bit (8-level) quantizers instead of 9-level quantizers to make the saving of the SDM output in an on-chip memory more efficient. With the formula (5.3), it can

be calculated that the $SQNR_{max}$ of a 3-bit 2-1-1 MASH SDM with OSR of 7 is about 106 dB, assuming its maximal $a_{12} = a_{23} = 7$ for the 3-bit quantizers. Its $SQNR_{max}$ is 6 dB higher than the 1-2 MASH SDM architecture with OSR = 8.6 and two 17-level quantizers reported in [16]. Moreover, to achieve the target 500 MHz BW, the 17-level 1-2 MASH SDM architecture with OSR = 8.6 requires to implement an SDM with 17-level quantization at 8.6 GHz, which seems not realistic (taking into account a reasonable power consumption) in 40 nm CMOS technology based on the information in Chapter 4.7 (Figure 4.9), while with the proposed 3-bit 2-1-1 MASH SDM architecture with OSR of 7, it requires to implement an SDM with 3-bit (8-level) quantization at 7 GHz, which is more realistic for the 40 nm CMOS technology.

For 3-bit 2-1-1 MASH architecture, the $SQNR_{max}$ of the ideal MASH SDM (106 dB) is much higher than the target specification of SNDR = 60 dB, which can make sure that the SNR is dominated by thermal noise instead of quantization noise, and at the same time leaves margin for system level optimization. Because of implementation difficulties (shown in Chapter 11.1), optimized zeroes are not used on the test chip. Much reduced inter-stage gains $a_{12} = 1.8$ and $a_{23} = 2.5$ are chosen to leave margin to accommodate implementation non-idealities. With these inter-stage gains, only the middle 4 (out of 8) quantization levels are triggered in the 2^{nd} and 3^{rd} loops in the ideal MASH SDM with full-scale input. With circuit implementation, due to the limited GBW of loop filter amplifiers, MASH SDM parameter variation and parasitic poles in the loop, the full-swing of the quantizers in the 2^{nd} and 3^{rd} loops are used with full-scale SDM input, justifying the choice of the ISG. Due to the reduction of the ISG from $a_{12} = a_{23} = 7$ to $a_{12} = 1.8$ and $a_{23} = 2.5$, the SQNR of the ideal MASH SDM is reduced from 106 dB to 89 dB. Since the optimized zeroes are not used on the test chip, the SQNR of the ideal MASH SDM is further reduced from 89 dB to 77.6 dB. The discrete-time model of the 3-bit 2-1-1 MASH SDM architecture is shown in Figure 5.3.

5.2 A generic synthesis flow for broadband CT MASH SDMs

In this section, the synthesis flow of a broadband CT MASH SDM architecture is presented. It is a generic synthesis flow which can be used for other broadband CT MASH SDM architectures with 1 T_S ELD compensation as well. This flow is suitable when the ELD is the dominant problem in the architecture design of broadband CT MASH $\Sigma\Delta$ ADCs. Its flowchart is shown in Figure 5.4.

- 1. Synthesize a DT MASH SDM model with zero loop delay based on the choices on the design parameters considering the target specifications. Then its internal loop filters (ILFs) and connecting loop filters (CLFs) in DT are derived [72].
- 2. Choose a CT MASH SDM model with 1 T_S loop delay, ELD compensation and



Figure 5.3: DT 2-1-1 MASH SDM model.

symbolic coefficients based on the required design freedom and implementation feasibility. Its ILFs and CLFs in CT are derived.

- 3. Perform CT-to-DT transformation with the impulse-invariant transformation method of the CT ILFs and CLFs with symbolic coefficients [36]. The equivalent DT ILFs and CLFs of the CT MASH model are obtained.
- 4. Equate the equivalent DT ILFs and CLFs (obtained in Step 3) with the desired DT ILFs and CLFs (obtained in Step 1). The coefficients of the CT MASH model are calculated by solving the equations.
- 5. Calculate the noise cancellation filters.

This synthesis flow starts with a DT MASH SDM with zero loop delay, and derive an equivalent CT MASH SDM with 1 T_S loop delay. This thesis considers it a better approach than directly synthesizing a CT MASH SDM in s-domain [73] for the following reasons: (1) It is straightforward to calculate the coefficients for a DT MASH SDM with zero loop delay when the main design parameters (OSR, order, OBG, etc.) are chosen. The CT-to-DT transformation and equations can be calculated and solved by Matlab. (2) In this synthesis flow, both the CT MASH SDM model and its equivalent DT MASH SDM model are acquired. Thus, it offers the flexibility to use the equivalent DT MASH SDM model for some simulations where it is more suitable, e.g. quantization gain simulation.

This section focuses on the mapping of the NTF. Other system-level optimization options such as STF optimization and dynamic range scaling are introduced in Section 5.3.



Figure 5.4: Flowchart of a generic synthesis procedure of broadband CT MASH SDMs.

5.2.1 Desired DT MASH SDM model

The desired DT MASH SDM model with zero loop delay is designed based on the requirements from the application and feasibility of implementation. It has been shown in the last section in Figure 5.3.

For the chosen 2-1-1 MASH architecture, 3 ILFs and 3 CLFs need to be mapped between the desired DT MASH model and the chosen CT MASH model. The internal loop filters are defined as the transfer function from the quantizer's output to the quantizer's input in the same SDM stage, and the connecting loop filters are defined as the transfer function from the quantizer's output of the upper stage to the quantizers' inputs of the lower stages. Here this thesis takes one ILF and one CLF as examples. In the desired DT MASH model (Figure 5.3), the ILF of the SDM stage 1 is the transfer function from V₁ to Q₁, which can be written as:

$$LF_{1,DT} = -\frac{2z-1}{(z-1)^2} \tag{5.4}$$

The CLF from stage 1 to stage 2 is the transfer function from V_1 to Q_2 , which can be written as:

$$LF_{12,DT} = -\frac{a_{12}z^2}{(z-1)^3} \tag{5.5}$$

5.2.2 CT MASH SDM model

This thesis proposes a broadband CT MASH SDM model with enough design freedom which is feasible for circuit implementation. One clock period ELD is compensated in every SDM stage, which is essential for broadband applications. Connection paths



Figure 5.5: Initial CT 2-1-1 MASH SDM model.

 $k_1 \sim k_{10}$ between the SDM stages offer the required design freedom to map both ILFs and CLFs [72, 77]. The initial CT model of the 3-bit 2-1-1 MASH SDM architecture is presented in Figure 5.5, which includes the paths needed for NTF mapping, but excludes the paths required for system-level optimization introduced in Section 5.3.

At this step, this thesis uses normalized sampling rate of $F_S = 1$, and dynamic range scaling of the CT model will be applied after the coefficients $a_1 \sim a_4$, $d_1 \sim d_3$ and $k_1 \sim k_{10}$ are calculated. Thus, at this step $c_1 = c_2 = c_3 = c_4 = 1$. Here, the total delay in the ELD feedback loop (e.g. from node V₁ through d₁ to node Q₁) is 1 (T_S). It indicates that at the next quantizer sampling moment, the ELD feedback signal is settled at the quantizer input. In circuit implementation, the quantizer delay is less than 1 (T_S), to allow the ELD DAC to settle.

In the CT MASH model (Figure 5.5), the corresponding ILF and CLF can be written as:

$$LF_{1,CT} = -\frac{a_1 + a_2 s}{s^2} - d_1 \tag{5.6}$$

$$LF_{12,CT} = -\frac{(a_1k_1 + a_2k_3)s + a_1k_3}{s^3} + \frac{k_5}{s} + k_9$$
(5.7)

Here, the ELD z^{-1} is not included in the equations (5.6) and (5.7), which will be

80

included in the following mapping equations (5.10) and (5.11).

5.2.3 CT-to-DT transformation

In this step, impulse-invariant transformation is applied to the CT ILFs and CLFs to get their DT equivalent [36]. Non-return-to-zero DAC pulse is used in this design, as multi-bit NRZ DAC is less sensitive to clock jitter than the return-to-zero DAC pulse. 1 T_S ELD is considered in the SDM loop. Thus, the equivalent DT transfer functions of the aforementioned CT ILF and CLF can be calculated as:

$$LF_{1,DT,eq.} = (1 - z^{-1})\mathcal{Z}\left\{\mathcal{L}^{-1}\left\{\frac{LF_{1,CT}}{s}\right\}\right\}$$
(5.8)

$$LF_{12,DT,eq.} = (1 - z^{-1})\mathcal{Z}\left\{\mathcal{L}^{-1}\left\{\frac{LF_{12,CT}}{s}\right\}\right\}$$
(5.9)

5.2.4 Equating equivalent and desired DT transfer functions

To ensure that the CT MASH model is equivalent to the desired DT MASH model regarding the noise transfer function, both ILFs and CLFs should be equivalent. Thus, the following equations can be written. Notice that the ELD z^{-1} in the CT MASH model is taken into account in the mapping here.

$$LF_{1,DT,eq.} = z \cdot LF_{1,DT} \tag{5.10}$$

$$LF_{12,DT,eq.} = z \cdot LF_{12,DT}$$
 (5.11)

By solving the equations from the mapping of 3 ILFs and 3 CLFs, the value of the coefficients $a_1 \sim a_4$, $d_1 \sim d_3$ and $k_1 \sim k_{10}$ in the CT MASH model is acquired.

5.2.5 Noise cancellation filters calculation

The principle of NCF calculation has been shown with an exemplary 2-stage MASH in Chapter 3.3.2. Here, from the CT MASH model with determined coefficients, the NTF and STF of every stage can be calculated as follows.

$$NTF_1 = \frac{(z-1)^2}{z^2} \tag{5.12}$$

$$NTF_2 = NTF_3 = \frac{z-1}{z}$$
 (5.13)

$$STF_1 = STF_2 = STF_2 = \frac{1}{z}$$
 (5.14)

Thus, the NCFs for this 3-stage MASH can be calculated as:

$$NCF_1(z) = STF_2(z) \cdot STF_3(z) \tag{5.15}$$

$$NCF_2(z) = \frac{NTF_1(z) \cdot STF_3(z)}{a_{12}}$$
(5.16)

$$NCF_3(z) = \frac{NTF_1(z) \cdot NTF_2(z)}{a_{12}a_{23}}$$
(5.17)

5.3 System-level optimization

This section discusses further steps on the system-level optimization for the broadband CT MASH SDM architecture, including signal transfer function optimization and dynamic range scaling.

Signal transfer function of the CT MASH SDM can be optimized separately to the noise transfer function. Signal feedforward paths $b_2 \sim b_5$ are added in the CT MASH SDM architecture, shown in Figure 5.6. Their value is optimized such that the STF is flat in-band, and low-pass filtering out-of-band. At the same time, the UGBW of the 1st integrator c_1 is maximized to suppress the noise and non-linearity of the 2nd integrator ($\frac{22}{s}$) and 2nd DAC (a_2).

Afterwards, dynamic range scaling is applied to the CT MASH model. For the chosen 40 nm CMOS technology, the supply voltage of thin-oxide transistors is 1.1 V. The maximal swing at 4 integrators' output are scaled to be ± 0.5 V, considering different input frequency and full-scale input amplitude. If the integrator output swing is too large, the non-linearity of the amplifier in the implementation of this integrator can deteriorate the linearity of the MASH ADC. If the integrator output swing is too small, it implies that the UGBW of this integrator can be further increased to suppress the noise and non-linearity contributed by the back-end blocks of this integrator. In the end of this step, a CT 2-1-1 MASH SDM model with all coefficients is synthesized.

5.4 System-level verification and block-level specifications

To check the performance and robustness of the CT 2-1-1 MASH SDM architecture, and to define specifications for the building blocks, various system-level verifications of the 2-1-1 MASH model have been done in Matlab. Firstly, the SQNR of the ideal CT MASH SDM model with different input power is simulated, which is plotted in Figure 5.7a. The input frequency is 159 MHz, which is slightly lower than $\frac{1}{3}$ of the signal bandwidth, such that up to 3^{rd} order harmonic (HD3) are in-band. Simulated dynamic range set by the in-band quantization noise is 77.6 dB. Simulated peak SQNR



Figure 5.6: CT 2-1-1 MASH SDM model after system-level optimization.

is 77.6 dB, which is reached at an input amplitude of $A_{in} = 0.8$ V. The simulated output spectrum for the peak SQNR is shown in Figure 5.7b.

For the chosen ADC sampling rate of 7 GS/s, clock jitter is a very important specification for the design of the clock circuitry. Chapter 4.4 shows that CT $\Sigma\Delta$ ADCs are sensitive to the jitter in the DAC clock, while less sensitive to jitter in the quantizer clock. This thesis models clock jitter in the proposed CT MASH model and simulates its effect. Figure 5.8 presents the simulated SNDR versus the jitter of the DAC clock with input amplitude $A_{in} = 0.7$ V and input frequency $f_{in} = 159$ MHz. In literature it is written that for a low-pass $\Sigma\Delta$ ADC with NRZ DAC pulse, when the SNR is completely limited by the white jitter noise, the SNR can be calculated as [61, 78]:

$$SNR_{jitter} = 10 \cdot log_{10} \frac{OSR \cdot V_{in}^2}{2\sigma_{\delta y}^2 (\frac{\sigma_j}{T_S})^2} \quad [dB]$$
(5.18)

Here, V_{in} is the input amplitude. $\delta y = y(n) - y(n-1)$ and y(n) is the SDM output. $\sigma_{\delta y}^2$ is the variance of δy , and σ_j is the standard deviation of the white jitter noise. From both Figure 5.8 and equation (5.18), for low-pass CT SDM with NRZ DAC



Figure 5.7: (a) Simulated SQNR vs. input power for CT 2-1-1 MASH architecture (b) Simulated output spectrum for the peak SQNR.



Figure 5.8: Simulated SNDR vs. clock jitter of the DAC clock.

pulse, when the SNR is completely dominated by the white jitter noise, the SNR reduces by 20 dB when the white jitter standard deviation σ_j increases by 10 times. Since the clock jitter error is designed to be not dominant in the SNDR of the ADC test chip, the DAC clock jitter specification is set to be less or equal to 100 fs.

To achieve the required quantization gain is essential for the SNDR target for broadband CT $\Sigma\Delta$ ADCs (see Chapter 4.2). Limited quantization gain is modeled using the linear gain model (equation (4.11)). The simulation with limited quantization gain is done with the DT MASH model for the following 3 reasons: (1) the CT MASH model and its equivalent DT MASH model show the same metastability behavior.



Figure 5.9: Simulated SNDR vs. quantization gain.

(2) Metastability error is a stochastic phenomenon, and for the target simulation accuracy, a very large number of samples (2^{20} sample) is simulated multiple times independently to achieve accurate result for quantization gain of 100 dB \sim 160 dB. (3) With the CT SDM model, many points (close to 100) need to be calculated by the simulator in every clock period, while with the DT SDM model, only one point needs to be calculated in every clock period. Thus, for the same number of samples, the simulation speed with DT SDM model is about 100 times faster than the CT SDM model. Figure 5.9 shows the simulated SNDR versus the quantization gain, with input amplitude $A_{in} = 0.7$ V and input frequency $f_{in} = 159$ MHz. For every quantization gain, 50 independent simulations are run. Some statistics of the simulated SNDR are plotted, such as minimal, maximal, mean values (red line) and 10%. 90% and middle values (blue line). When the quantization gain is 130 dB, all of the 50 independent simulations achieve SNDR \geq 71 dB, and 90% of the runs achieve $SNDR \ge 73$ dB. Considering the SNDR target for this test chip is 60 dB, the target specification for the quantization gain is chosen to be higher or equal to 130 dB. Here, this thesis considers applications for which multiple measurements are done and the metastability error power averages out by the large amount of samples, such as automotive radar. However, some other applications can have a more stringent requirement for the occurrence rate of the metastability error.

The robustness of the proposed CT 2-1-1 MASH architecture is checked with simulations. All the parameters (a, b, c, d, k, p in Figure 5.6) are modeled with a Gaussian distributed error with a standard deviation σ . For every σ , 50 independent simulations are run. The simulated SNDR versus parameter variation σ is plotted in Figure 5.10. Some statistics of the simulated SNDR are plotted similar as in Figure 5.9, such as minimal, maximal, mean values (red line) and 10%, 90% and middle values (blue line). For CT MASH SDMs, the parameter variations can deteriorate



Figure 5.10: Simulated SNDR with parameter variations.

the SNDR or even the stability of the SDM for the following three reasons: (1) The parameter variations make the actual NTF deviating from the ideal NTF, which can deteriorate the SNDR and even the stability. (2) Because of the parameter variations, the analog transfer functions of the quantization noise do not match with the digital NCFs anymore, which causes noise leakage of the front-end quantization noise in the MASH output and deteriorates the resolution [42, 73]. (3) The parameter variations change both the noise transfer functions and the signal transfer functions, which can make the back-end stages of the MASH overloading with either the front-end quantization noise or the input signal. When the back-end stages are overloaded, part of the information on the quantization noise of the front-end stages is lost, and the front-end quantization noise cannot be efficiently canceled. Notice that the first point applies to single-loop SDMs as well, while the last two points are specific for MASH structures. With $\sigma = 1\%$, all 50 independent simulations achieve SNDR ≥ 73.2 dB, and 90% of the runs achieve SNDR ≥ 75 dB. With $\sigma = 2\%$, 90% of the runs achieve SNDR ≥ 70.4 dB.

Many other non-idealities are modeled and simulated with the CT MASH model or the equivalent DT MASH model. Based on the simulation results, ADC blocklevel and system-level target specifications are drawn, which are summarized in Table 5.2. The DC gain and parasitic pole specifications on the integrators is presented in Chapter 6, where the analog and digital filter mismatch calibration is considered.

Ite	Target specifications		
DAC clock	$\leq 100 \text{ fs}$		
	DAC A1	$\sigma \le 6 \cdot 10^{-4}$	
DAC static mismatch	DAC A2 and K5	$\sigma \leq 1.2 \cdot 10^{-3}$	
	Other DACs	$\sigma \leq 7 \cdot 10^{-3}$	
Clock skew mismatch be	$\sigma \leq 400 \text{ fs}$		
Quantization gain	Quantizer 1	$\geq 130 \text{ dB}$	
Quantization gain	Quantizer 2 and 3	$\geq 110 \text{ dB}$	
Quantizer unit elem	$\sigma \leq 2\% V_{FS}$ quantizer		
Quantizer	$\leq 7\% V_{FS}$ quantizer		
Integrator non-linearity	Integrator 1	$h_3 \le 0.01$	
	Integrator 2	$h_3 \le 0.03$	
$V_{out} = (V_{in} - h_3 V_{in}^3) \frac{c_i}{s}$	Integrator 3 and 4	$h_3 \le 0.1$	
Loop dela	$\leq \pm 0.05 \ \mathrm{T}_S$		
Parameter variation (a, b	$\sigma \le 1\%$		

Table 5.2: MASH ADC block- and system-level target specifications

5.5 Conclusions

The synthesis of a broadband CT $\Sigma\Delta$ ADC architecture starts from a comparison of peak SQNR among multiple DT multi-bit single-loop and MASH $\Sigma\Delta$ ADC architectures. For broadband medium-to-high resolution CT SDMs, any resolution (SNDR or DR) target point imposes a bandwidth limitation for all relevant technology nodes in case of single loop SDMs. For CT MASH SDMs this limitation can be alleviated to allow for a broader bandwidth at the cost of complexity. This chapter proposes generic formulas to calculate the SQNR limits for both single-loop and MASH SDMs, which reduces the complexity of system level design and comparison significantly. Considering the speed-resolution limit of the 40 nm CMOS technology, a 3-bit 2-1-1 MASH $\Sigma\Delta$ ADC architecture is chosen for the test chip implementation because of its robustness and potential for broadband application. This MASH architecture shows 6 dB higher $SQNR_{max}$ potential than the 1-2 MASH architecture reported in [16], and it is more realistic to be implemented in the chosen 40 nm CMOS technology because of its lower sampling rate and quantizer resolution. A generic synthesis flow for broadband CT MASH SDMs is presented here, and a CT 2-1-1 MASH SDM architecture with $1-T_S$ ELD compensation is synthesized including system level optimization. Various system-level simulations are run in Matlab, which show the robustness of the chosen architecture. Block-level and system-level design specifications are drawn based on the simulation results. The synthesized CT 3-bit 2-1-1 MASH $\Sigma\Delta$ ADC architecture is feasible to be implemented in the chosen 40 nm CMOS technology.

Part II

Novel design concepts

Chapter 6

A digital calibration technique for broadband CT MASH $\Sigma\Delta$ ADCs

This chapter proposes a digital automatic calibration method for broadband continuoustime MASH $\Sigma\Delta$ ADCs [79]. The main contribution of this method is the calibration of the errors due to the limited DC gain and 2^{nd} pole of the loop filter integrators. Without the proposed calibration technique, the stringent DC gain and 2^{nd} pole requirements demand power-hungry loop filter implementation, and the noise leakage due to the limited DC gain and 2^{nd} pole can degrade the resolution of the MASH $\Sigma\Delta$ ADCs. The digital noise cancellation filters are calibrated by successive estimation of the DC gains and the 2^{nd} pole, and updating the coefficients of the FIR filters. Extensive system-level and transistor-level simulations demonstrate the effectiveness and robustness of the proposed method. For an exemplary MASH $\Sigma\Delta$ ADC with BW of 600 MHz and SQNR of 75 dB, it can reduce the DC gain requirement by 15 dB, and the 2^{nd} pole requirement by three times, making it an enabling technique for power-efficient GHz-range $\Sigma\Delta$ ADC applications. Although the exemplary 2-1-1 MASH ADC architecture in this chapter is slightly different compared to the 2-1-1 MASH architecture shown in Chapter 5 regarding the sampling rate, signal bandwidth, quantizer resolution and NTF, the proposed digital automatic calibration is generic and can be directly applied to the MASH architecture of the test chip implementation.

6.1 Introduction

Generally, two error sources negatively impact the accuracy of the SDM parameters – spread and device mismatch. The spread is caused by a process variation in the fabrication, supply voltage and temperature variations. The spread causes a common relative error on all of the same type of devices on chip, e.g. a common relative error on the resistance value of poly resistors, on the capacitance value of fringe capacitors, and on the threshold voltages of the PMOS and NMOS transistors. Device mismatch

is caused by the fabrication inaccuracy, which introduces different relative errors on the same type of devices [71].

Generally, the resolution of CT MASH $\Sigma\Delta$ ADCs is limited by the mismatch between analog loop filters and digital noise cancellation filters. Among all error sources, a dominant error source is the RC spread error, which causes the same relative error on the unity-gain bandwidth of all integrators. It can be typically up to $\pm 30\%$. To overcome the limitation of the RC spread error, digital calibration [42] or digitally assisted analog tuning [73] have been demonstrated. RC spread calibration alone is usually sufficient to regain the SQNR for MASH ADCs with several tens of MHz BW, but for broadband MASH ADCs discussed in this thesis, the DC gain and 2nd pole effects of the analog integrators need to be taken into account as well.

The work of [80] proposes a digital calibration which can compensate the SNDR drop caused by both the RC spread and the finite gain and bandwidth of the opamps. It employs a Block Least Mean Square (BLMS) algorithm for adaptive tuning of the FIR filters. To ensure the convergence of the BLMS algorithm, it uses a very small step size μ (2⁻¹² to 2⁻²²), a very large block size (2¹⁸) and it needs a long calibration time (3000 iterations). Moreover, the BLMS algorithm is relatively complicated and requires multipliers, high order FIR filters (order 40) and a dither signal generator. The work of [80] calibrates the RC spread error in the digital domain; it does not correct the UGBW of the analog integrators. To avoid overloading of the cascaded loops, extra margin must be reserved for the large RC spread error by decreasing the inter-stage gains. Consequently, the maximal achievable SQNR is reduced.

In this chapter, a digital calibration algorithm for the integrator DC gain (A_{DC}) and 2^{nd} pole (f_{p2}) is presented, which assists a preceding analog calibration of the RC spread errors as well. Firstly, the calibrations of the RC spread error and the error due to the limited A_{DC} and f_{p2} are split. The RC spread error is calibrated by the analog tuning. The digital part of the proposed calibration method focuses on the limited A_{DC} and f_{p2} , which makes it more efficient than [80]. Moreover, by analysing the error mechanisms, this thesis approximates them by a linear dependence between the NCF coefficients correction factors and the reciprocal of the DC gain A_{DC}^{-1} (or the reciprocal of the 2^{nd} pole f_{p2}^{-1}). Based on this, the proposed method shows fast convergence and low calculation complexity. It needs simple hardware and is suitable for low-cost integration. Section 6.2 describes the modelling and analysis of the loop filter non-idealities. Section 6.3 presents the error sources causing the mismatch between the analog transfer function of the quantization noise and the digital NCF. Section 6.4 introduces the analog RC calibration. Section 6.5 presents the proposed digital calibration algorithm. System-level and transistor-level simulation results are shown in Section 6.6. Finally, Section 6.7 draws the conclusions.



Figure 6.1: Model of the (a) RC integrator and (b) Gm-C integrator.

6.2 Modelling of the loop filter non-idealities

Chapter 4.1 shows that CT integrators can be implemented as either RC integrators (Figure 6.1a) or Gm-C integrators (Figure 6.1b). Here this chapter recaps their properties on linearity, DC gain and 2^{nd} pole, and further give their models with limited DC gain and 2^{nd} pole. The RC integrator shows better linearity and it has been commonly implemented as the 1st integrator at the input stage [42], where the integrator non-linearity is not shaped. However, the capacitance at the output of the amplifier (C_{out}) creates a 2^{nd} pole in the transfer function of the RC integrator, which restricts its use in the fast loops. The transfer function of the RC integrator is modelled as (6.1), in which *UGBW* is the designed UGBW of the RC integrator, and A_{DC} and f_{p2} are its DC gain and 2^{nd} pole.

$$\frac{V_{out}}{V_{in}} = \frac{UGBW}{\frac{UGBW}{A_{DC}} + s + \frac{1}{2\pi f_{p2}} \cdot s^2}$$
(6.1)

On the contrary, the Gm-C integrator uses the amplifier in an open loop. It cannot achieve as high linearity as the RC integrator. The amplifier output capacitance C_{out} is in parallel with C, which does not create a 2nd pole, and only adds to the wanted pole. Gm-C integrators are commonly used in the high-speed paths, where the integrator non-linearity is shaped [42]. Its transfer function is modelled as:

$$\frac{V_{out}}{V_{in}} = \frac{UGBW}{\frac{UGBW}{A_{DC}} + s}$$
(6.2)

The following section discusses the impact of the limited DC gain and 2^{nd} pole on the performance of MASH $\Sigma\Delta$ ADCs.

6.3 Mismatch between analog filter and digital NCF

This thesis considers the following three error sources causing the mismatch between the analog transfer function of the quantization noise and the digital NCF: (1) the RC spread (2) the limited DC gain and 2^{nd} pole of the integrators (3) the parameter variation due to the device mismatch. In the following they are discussed.

The RC spread is a well-known problem for the CMOS technology. For the exemplary 2-1-1 MASH SDM used in this chapter, $\pm 20\%$ RC time constant error causes a degradation of the peak SNDR from 75 dB to $48 \sim 50$ dB (shown in Figure 6.6). The calibration for the RC time constant error is introduced in Section 6.4.

The limited DC gain and 2^{nd} pole of the integrators make the actual analog transfer functions of the quantization noise deviating from the ideal transfer functions. Thus, the actual analog transfer functions of the quantization noise are not matched with the digital NCFs, which causes noise leakage of the front-end quantization noise to the MASH output. The leaking front-end quantization noise is shaped by an NTF with less order than the ideal NTF of the MASH. Thus, the limited DC gain and 2^{nd} pole can deteriorate the resolution of MASH SDMs. After the RC time constant error is calibrated, the limited DC gain and 2^{nd} pole become dominant for the mismatch between the analog transfer functions and the digital NCF for broadband MASH SDMs. For the exemplary 2-1-1 MASH SDM used in this chapter, the requirements on the 1^{st} - 3^{rd} integrators' DC gain are higher than 40 dB, and the requirement on the 1^{st} integrator's 2^{nd} pole is higher than 30 GHz, without the proposed digital calibration technique in this chapter (shown in Figure 6.4). To achieve a DC gain of more than 40 dB and a 2^{nd} pole of higher than 30 GHz for the 1^{st} integrator leads to stringent requirement on the DC gain and GBW of its amplifier. An amplifier with complex architecture, e.g. multistage and multipath architecture, and high power consumption is needed which increases the design complexity and power consumption of the ADC [34]. The proposed digital calibration technique to combat these errors is shown in Section 6.5.

The last error source is the parameter variation (shown in Chapter 5.4 with Figure 5.10). Due to the device mismatch in the fabrication, the SDM parameters can deviate from their ideal values. Contrary to the RC spread, the parameter variation causes uncorrelated random errors on all SDM parameters. The calibration for the errors due to the parameter variation is out of the scope of this chapter. The MASH SDM architecture should be designed with enough robustness to accommodate the parameter variation. And the circuit design should guarantee that the parameter variation is within the tolerance.



Figure 6.2: Block diagram of the proposed calibration technique

6.4 Analog RC calibration

The proposed calibration technique consists of two parts – analog calibration of the RC constants (UGBW of the integrators), and the digital calibration for the limited A_{DC} and f_{p2} by updating the NCF coefficients, as shown in Figure 6.2. The analog RC calibration is implemented similar as [73]. With no input signal, the variance of the decimation filter output is considered as the measurement of the in-band noise (IBN) [42]. With ideal NCFs, the capacitor banks in the RC integrator or the Gm-C integrator receive a code from the calibration engine. Assuming a common RC spread over the whole chip, the capacitor bank code can be swept to find a minimal value of the IBN. The minimum IBN indicates the correct capacitor bank code of the analog RC calibration.

6.5 Digital calibration algorithm

In this section, the proposed digital calibration algorithm for limited A_{DC} and f_{p2} is presented in detail, assuming pre-calibrated RC constants. Without loss of generality, a CT feedback 2-1-1 MASH $\Sigma\Delta$ ADC architecture is considered to substantiate the proposed calibration algorithm. The sampling frequency is chosen as $F_S = 8.4$ GS/s and OSR = 7, to achieve a signal BW = 600 MHz. Every loop uses 9-level quantization. The 1st integrator in the 1st loop is considered to be implemented as an RC integrator, while the remaining 3 integrators are implemented as Gm-C integrators.

6.5.1 Calculation of the correct digital NCFs

From the CT model of the loop filter with non-idealities, both the ILFs and the CLFs are derived. For the 2-1-1 MASH structure shown in Figure 6.2, in total 3 ILFs $H_{i,CT}$ and 3 CLFs $H_{ij,CT}$ are calculated. Here, $H_{i,CT}$ and $H_{ij,CT}$ contain both DAC pulse functions and the transfer functions of the CT loop filters.

Then, the impulse invariance transformation [36] is applied to calculate the effective DT noise transfer functions of every loop with the modelled loop filter non-idealities as (6.3) - (6.4). Without loss of generality, normalized sampling frequency $F_S = 1$ Hz is assumed.

$$H_{i,DT} = \mathcal{Z} \left\{ \mathcal{L}^{-1} \{ H_{i,CT} \}_{t=n} \right\}$$

$$(6.3)$$

$$NTF_i = \frac{1}{1 - z^{-1} \cdot H_{i,DT}} \tag{6.4}$$

Similarly, the effective transfer functions of the upper loops' quantization noise to the lower loops' output can be calculated as:

$$H_{ij,DT} = \mathcal{Z} \left\{ \mathcal{L}^{-1} \{ H_{ij,CT} \}_{t=n} \right\}$$

$$(6.5)$$

$$TF_{ij} = H_{ij,DT} \cdot z^{-1} \cdot NTF_j \tag{6.6}$$

Then, the correct digital NCFs which match the analog loop filters with the modelled non-idealities can be calculated as (6.7) - (6.9), such that the quantization noise of the 1st loop and the 2nd loop is completely cancelled in the final output. In (6.7), a_{12} and a_{23} are the designed inter-stage gains.

$$NCF_3 = \frac{NTF_1 \cdot NTF_2}{a_{12}a_{23}} \tag{6.7}$$

$$NCF_2 = -TF_{23} \cdot NCF_3 \tag{6.8}$$

$$NCF_1 = -TF_{13} \cdot NCF_3 \tag{6.9}$$

6.5.2 Calibration algorithm

The three calculated NCFs in Section 6.5.1 share the same denominator, which shows flat in-band magnitude with less than ± 1 dB of gain. Thus the correction on the denominator can be ignored, without compromising performance. After calibration, the digital NCFs have the form of (6.10), in which only $N_{12} - N_{15}$, $N_{22} - N_{25}$, and N_{30} $- N_{33}$ are non-zero. Three FIR filters of order 4 and tunable coefficients are required. Note that without calibration, three FIR filters of order 3, 4, 4, respectively, are already needed to implement the nominal NCFs. The extra hardware required by the proposed calibration is one extra order for one FIR filter and the tunability of the FIR coefficients.

$$NCF_{i,cal} = \sum_{j=0}^{5} N_{ij} z^{-j}$$
(6.10)

Figure 6.3a shows the calculated NCF_1 coefficients versus different DC gain of the 1st integrator $A_{DC,1}$ in an otherwise ideal system. Figure 6.3b presents the difference between the calculated coefficients and the ideal coefficients, ΔN_{1j} (referred as the correction factors in this thesis), versus $A_{DC,1}$. ΔN_{1j} are linear functions of $A_{DC,1}^{-1}$ to the 1st order. It is similar for NCF_2 and NCF_3 coefficients. ΔN_{ij} versus the 2nd pole of the 1st integrator, and the DC gains of the 2nd and 3rd integrators shows similar dependence, by ignoring very small coefficients. This property can be summarized as (6.11), in which the inter-dependent part is ignored. In (6.11), $\alpha_{ij,A_{DC,k}}$ and $\alpha_{ij,f_{p12}}$ are the simulated coefficients of the linear functions.

$$N_{ij} = N_{ij,nom} + \Delta N_{ij,A_{DC,1}} + \Delta N_{ij,f_{p12}} + \Delta N_{ij,A_{DC,2}} + \Delta N_{ij,A_{DC,3}} , \text{where } \Delta N_{ij,A_{DC,k}} \approx \alpha_{ij,A_{DC,k}} \cdot A_{DC,k}^{-1}$$
and $\Delta N_{ij,f_{p12}} \approx \alpha_{ij,f_{p12}} \cdot f_{p12}^{-1}$

$$(6.11)$$

Based on (6.11), a 4-dimensional calibration algorithm is proposed. For simplicity, the 1-dimensional calibration for $A_{DC,1}$ is explained in detail in the following, and the 4-dimensional calibration can be implemented similarly.

Initialization: The matrix of nominal FIR coefficients $[N_{ij,nom}]$ and the calculated correction factors for one initial guess $A_{DC,1,ini} [\Delta N_{ij,A_{DC,1,ini}}]$ is stored.

Logarithmic scale binary search: With initial $step = 2^4 = 16$, $[\Delta N_{ij}] = [\Delta N_{ij,A_{DC,1,ini}}]$. The IBN is measured and compared for 3 settings of the NCF coefficients: (a) $[N_{ij,nom}] + [\Delta N_{ij}]$; (b) $[N_{ij,nom}] + [\Delta N_{ij}] \cdot step$; (c) $[N_{ij,nom}] + [\Delta N_{ij}]/step$. If the setting (b) (or (c)) leads to a lower IBN, $[\Delta N_{ij}] = [\Delta N_{ij}] \cdot step$ (or $[\Delta N_{ij}] = [\Delta N_{ij}]/step$); otherwise $step = \sqrt{step}$. Logarithmic scale binary search is finished until step < 2.

Linear scale binary search [72]: With initial $step = \frac{1}{2}$. The IBN is measured and compared for 3 settings of the NCF coefficients: (a) $[N_{ij,nom}] + [\Delta N_{ij}]$; (b) $[N_{ij,nom}] + [\Delta N_{ij}] \cdot (1 + step)$; (c) $[N_{ij,nom}] + [\Delta N_{ij}] \cdot (1 - step)$. If the setting (b) (or (c)) leads to a lower IBN, $[\Delta N_{ij}] = [\Delta N_{ij}] \cdot (1 + step)$ (or $[\Delta N_{ij}] = [\Delta N_{ij}] \cdot (1 - step)$); otherwise step = step/2. Linear scale binary search is finished until $step < step_{min}$. $[\Delta N_{ij,est}] = [\Delta N_{ij}]$.

The whole calibration algorithm requires only summations, bit shifts and comparisons. No multiplications are needed. Simulations show that the required estimation accuracy of the DC gains and the 2nd pole is within \pm 10%, so the minimum step is $step_{min} = 5\%$. When the calibration is finished, the estimation of A_{DC,1} can be calculated as (6.12), in which ΔN_{ij} can be any non-zero element in [ΔN_{ij}].

$$A_{DC,1,est} = A_{DC,1,ini} \cdot \frac{\Delta N_{ij,A_{DC,1,ini}}}{\Delta N_{ij,est}}$$
(6.12)



Figure 6.3: (a) $NCF_{1,cal}$ coefficients N_{1j} and (b) Correction factors ΔN_{1j} versus 1st integrator DC gain A_{DC.1}.

30

A_{DC,1} (dB) (b)

20

50

55

40

6.6 Simulation results

10

10⁻⁵

10^{-6 __}0

Δ N₁₂ ΔN_{13}

Δ N₁,

10

To verify the effectiveness of the proposed calibration algorithm, the SNDR before and after calibration versus the DC gains of the integrators and the 2nd pole of the 1st integrator are simulated separately in Matlab/Simulink, as shown in Figure 6.4. Only one DC gain or 2nd pole parameter is swept and the SDM is otherwise ideal in every figure. Firstly, the proposed calibration is run with no input signal, and the correct NCF coefficients are found and fixed. The digital calibration is run with no input signal since in this case the noise leakage due to the mismatch between the digital NCF and the analog transfer function is the dominant error source in the IBN before calibration. Then, a sinusoid signal with -3 dBFS amplitude whose



Figure 6.4: SNDR before and after proposed calibration: (a) versus 1^{st} integrator DC gain; (b) versus 1^{st} integrator 2^{nd} pole; (c) versus 2^{nd} integrator DC gain; (d) versus 3^{rd} integrator DC gain.

frequency is at about 1/3 of the signal BW is used as the input signal to show the SNDR improvement, so that the $3^{\rm rd}$ order harmonic distortion is taken into account. When the analog loop filters are non-ideal, the quantizers can be slightly overloaded with close to full scale input. With the proposed calibration, NCFs and analog loop filters are matched, and the HD3 tones originated from the $1^{\rm st}$ and the $2^{\rm nd}$ quantizers are eliminated. Figure 6.4a, 6.4c, and 6.4d show that the DC gain requirements (for SNDR > 70 dB) is relaxed by about 15 dB for the $1^{\rm st} - 3^{\rm rd}$ integrators with the proposed calibration. As the error due to the limited DC gain of the $4^{\rm th}$ integrator is already $3^{\rm rd}$ order shaped, its DC gain requirement is very low. With the proposed calibration, the DC gain requirements of $1^{\rm st} - 4^{\rm th}$ integrators are 25 dB, 25 dB, 30 dB, 20 dB, respectively. Figure 6.4b shows that the $2^{\rm nd}$ pole requirement of the $1^{\rm st}$ integrator is relaxed from higher than 17.8 GHz to 5.6 GHz.

To verify the convergence and robustness of the proposed 4-dimensional calibration, 200 independent simulations are done with full scale input amplitude in Matlab/Simulink. The DC gains of the $1^{st} - 4^{th}$ integrators are independently uniformly



Figure 6.5: Histogram of the SNDR before and after proposed calibration: 200 independent simulations

distributed in the range of (25, 39) dB, (25, 39) dB, (30, 44) dB and (20, 34) dB, respectively. The 2nd pole of the 1st integrator is independently uniformly distributed in the range of (7, 35) GHz. Figure 6.4 shows that when the DC gain and the 2nd pole are in these ranges, they cause a significant degradation on SNDR without calibration. With the proposed calibration, the SNDR is restored close to that of the ideal SDM. Figure 6.5 shows the histogram of the simulated SNDR before and after the proposed calibration. Before calibration, the mean value of the SNDR is 68.1 dB with standard deviation $\sigma = 2.7$ dB, which is improved to 75 dB with $\sigma = 0.7$ dB after calibration. The mean value of the calibration time is 166 IBN measurements with $\sigma = 29$ IBN measurements. $2^{13} = 8192$ samples are used in every IBN measurement. It shows that the calibration algorithm always converges with large SNDR improvement. For all simulations in Figure 6.5, the IBN converges to the global minimum or to a local minimum, very close to the global minimum.

The proposed digital calibration can assist the analog calibration of the RC spread error. To verify that, the SNDR before and after digital calibration with full scale input amplitude versus different RC spread errors are simulated in Matlab/Simulink and presented in Figure 6.6. The integrator non-idealities are $A_{DC,1} = 35$ dB, $f_{p12} = 27$ GHz, $A_{DC,2} = 25.1$ dB, $A_{DC,3} = 38.6$ dB, and $A_{DC,4} = 28.1$ dB, which is one randomly generated setting from the simulation of Figure 6.5. The dashed red line shows that without the digital calibration, when the RC spread error is calibrated by analog tuning (Chapter 6.4) to about $\pm 5\%$, the SNDR is limited by the limited A_{DC} and f_{p2} , and it is difficult to select the best code for the capacitor bank. For every



Figure 6.6: SNDR vs. RC spread before and after proposed calibration

RC spread value in Figure 6.6, the proposed digital calibration is applied. Since the digital calibration shows optimal performance for the pre-calibrated RC constants, it can clearly indicate the correct code for the analog calibration. Thanks to the digital calibration, the accuracy of the analog RC calibration is improved to be better than $\pm 1\%$. After the RC spread error is calibrated, the digital calibration improves the SNDR by 12.4 dB.

The proposed calibration method is further verified by the simulation of a transistorlevel schematic of the 2-1-1 MASH loop filter in Cadence Virtuoso. The loop filter is designed in a 40 nm CMOS technology with inverter-based operational transconductance amplifiers (OTAs). The simulated DC gains of the 1st – 4th integrators are 27.9 dB, 31.3 dB, 33.9 dB, 32.2 dB, respectively, and the simulated 2nd pole of the 1st integrator is 12.3 GHz. The simulated FFT (8192 samples) before and after the proposed calibration with -3 dBFS input is shown in Figure 6.7. The expected quantization noise level is calculated with the synthesized ideal NTF. Before calibration, the IBN is -70.5 dBFS, and SNDR = 65.8 dB. After calibration, the IBN is improved to -75.2 dBFS, and SNDR is improved to 68.7 dB. The flat noise floor at low frequency (below 100 MHz) after calibration is limited by the DC gain of the integrators, which is not limiting the IBN. In another simulation with 0 dBFS input signal, the IBN is improved from -67.9 dBFS to -75.5 dBFS, very close to the IBN of the ideal SDM -76.5 dBFS.



Figure 6.7: Simulated FFT with transistor-level model of the loop filter before and after proposed calibration

6.7 Conclusions

A calibration consisting of an analog calibration of the integrators' UGBW (RC value) and a novel digital calibration of the errors due to the integrators' limited DC gain and 2^{nd} pole for CT MASH $\Sigma\Delta$ ADCs is proposed. It has very low calculation complexity, short calibration time, and requires simple hardware. Thus, it is suitable for low cost implementation. Extensive simulations show its robustness. For an exemplary 9-level 2-1-1 MASH architecture with BW of 600 MHz and SQNR of 75 dB, the proposed method relaxes the DC gain requirements of the 1st – 3rd integrators by about 15 dB, and relaxes the 2nd pole requirement of the 1st integrator from higher than 17.8 GHz to 5.6 GHz. Transistor level simulations demonstrate 7.6 dB improvement of the IBN. All these properties and results show its potential for power efficient GHz-range ADC applications.

Chapter 7

High-speed ELD compensation

This chapter discusses high-speed ELD compensation techniques for broadband CT $\Sigma\Delta$ ADCs. First, the state-of-the-art high-speed ELD compensation techniques are studied and classified. The parasitic delay in the ELD loop itself e.g. due to the poles of the summing amplifier and pre-amplifier limits the maximal sampling rate of the $\Sigma\Delta$ ADCs, which is the problem that this chapter would like to solve. A novel current-mode multi-path ELD compensation technique is proposed in Section 7.2. This section is based on our publication [81] and patent [82]. Section 7.3 further employs cross-coupled feedforward capacitors to compensate the ELD settling delay caused by the parasitic capacitance on the summation nodes, and propose a generic summation node at the quantizer input. The "generic" here refers to the summation of an arbitrary signal required by the system optimization, which can be the SDM input, or an internal node of the loop filter, or an input from a proceeding stage in a MASH structure. In the end, conclusions are drawn.

7.1 Overview for high-speed ELD compensation techniques

Chapter 4.5 has introduced the concept of the excess loop delay and the ELD compensation. Chapter 5 has shown the synthesis of a broadband CT 2-1-1 MASH SDM architecture with 1 T_S ELD compensated in every stage. The ELD feedback information needs to settle before the next sampling moment of the quantizer, otherwise the resolution and stability of the SDMs can be negatively impacted. This section presents an overview on the state-of-the-art system-level and circuit-level implementations of high-speed ELD compensation techniques based on a literature study. Firstly, a conventional implementation of the ELD compensation with a dedicated summation amplifier is presented. Alternatively, the ELD compensation path can be implemented together with the last stage of the loop filter – either an RC integrator or a Gm-C integrator. Last but not least, the ELD compensation can be implemented with passive



Figure 7.1: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with a dedicated summation amplifier: (a) with a current-steering DAC (b) with a voltage DAC.

summation, either in voltage domain or in current domain. Digital ELD compensation techniques, and the ELD compensation techniques for VCO-based quantizers and TDC-based quantizers are out of the scope of this thesis [27, 51, 77, 83].

7.1.1 ELD compensation with dedicated summation amplifier

One conventional implementation of the summation of the loop filter output and the ELD DAC output is using a dedicated summation amplifier. It is called active summation as well. Two slightly different implementations are shown in Figure 7.1. In Figure 7.1a, a current steering DAC (IDAC) is used as the ELD compensation DAC. The loop filter output voltage is converted into the current domain by the resistors R_1 , then summed with the IDAC output current. A slightly different implementation



Figure 7.2: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with a RC integrator and a resistor R_z to generate a zero.

is shown in Figure 7.1b. A voltage DAC is used as the ELD compensation DAC in Figure 7.1b. In both Figure 7.1a and 7.1b, the summation amplifier output node contains the summation result of the loop filter output and the ELD DAC output. As the implementations of Figure 7.1a and 7.1b are very similar, in the following discussion, only one of these two alternative implementations is explicitly shown.

The approach of using a dedicated summation amplifier has advantages and disadvantages. The advantage is that the output swing of the summation amplifier can be chosen relatively large, which relaxes the offset requirement of the quantizer when a multi-bit quantizer is used. The disadvantage is that it requires a dedicated amplifier. And it requires a very high GBW from this amplifier, to achieve a far parasitic pole at the output of the amplifier. Thus, this summation amplifier typically has high power consumption.

7.1.2 ELD compensation with RC integrator with zero resistor

An alternative approach to implement the ELD compensation path is to build up a zero-order path around the last integrator. Figure 7.2 shows the 1st case – the last stage of the integrator is implemented as an RC integrator, and the zero-order path is achieved by adding a resistor R_z in series with the integration capacitor C_1 . In this case, by choosing a proper value for R_z , the direct feedback path around the quantizer is achieved. This ELD compensation approach has been demonstrated in [63].

This approach has advantages and disadvantages. The advantage is that it reuses the amplifier of the integrator to build the ELD compensation path, and hence it saves the power consumption of an additional amplifier compared to Figure 7.1. The


Figure 7.3: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with a RC integrator and a capacitive DAC.

disadvantage is that this approach imposes a very high requirement on the GBW of the amplifier.

7.1.3 ELD compensation with RC integrator and capacitive DAC

Figure 7.3 shows an alternative approach to Figure 7.2. Instead of using R_z , it adds a capacitive DAC (CDAC) to the virtual ground node of the RC integrator, in parallel with the existing current-steering DAC. By doing that, a C-C path with the CDAC and the integrator capacitor C_1 is formed, which is a zero-order path. This zero-order path is the ELD compensation path around the quantizer. This ELD compensation approach has been demonstrated in [16].

This approach has pros and cons. Similar as in Figure 7.2, it reuses the amplifier of the RC integrator, and hence doesn't need any additional amplifier. The C-C path formed by the CDAC and C_1 is able to implement a wide range for the ELD compensation coefficient. However, it has two disadvantages. Firstly, the amplifier is in the ELD compensation loop, and any parasitic pole at the output of the amplifier is in the ELD compensation loop. Thus it gives stringent GBW requirement on the amplifier. Secondly, the capacitive ratio between the capacitance used in the CDAC and C_1 is determined by the ELD feedback coefficient. CDAC adds to the load for the quantizer, which increases its power consumption.



Figure 7.4: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with a Gm-C integrator and a capacitor $-C_d$.

7.1.4 ELD compensation with Gm-C integrator and capacitive DAC

The last stage of the integrator can be a Gm-C integrator as well. With this Gm-C integrator, there are two approaches to implement the ELD compensation loop. The 1^{st} approach is shown in Figure 7.4. The current-steering DAC is the feedback DAC to the input of the last integrator. The capacitor $-C_d$ is used for the ELD compensation [84]. $-C_d$ is interpreted as two capacitors C_d between V_{in}^+ and V_{out}^- , and between V_{in}^- and V_{out}^- in a differential implementation. The capacitor $-C_d$ with the integration capacitor C_1 forms a C-C path, which is a direct feedback path around the quantizer. This ELD compensation approach has been demonstrated in a 1-bit CT SDM in [84].

7.1.5 ELD compensation with Gm-C integrator, current-steering DAC and digital differentiator

An alternative approach to Figure 7.4 is to add a digital differentiator at the input of the current-steering DAC, as shown in Figure 7.5. In this approach, the quantizer output is firstly differentiated by the digital differentiator $(1 - z^{-0.5})$. After that, this differentiated quantizer output is converted to the current domain by the IDAC. The DAC current is summed with the amplifier (g_m) output current. And the summed current is integrated by the capacitor C_1 . So the quantizer output is firstly differentiated by the digital differentiator, and later integrated by the capacitor. An ELD compensation loop is thus built. This ELD compensation approach has been demonstrated in [30].

This approach has pros and cons. The advantage is that the input parasitic capacitance of the quantizer can be considered as (part of) the integration capacitor C_1 .



Figure 7.5: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with a Gm-C integrator, a current-steering DAC and a digital differentiator.

Thus, no parasitic pole is added at the summation node output. The disadvantage is that to implement the digital differentiator $(1 - z^{-0.5})$, half clock period delay $(z^{-0.5})$ is needed. Thus, the available time for the regeneration of the comparator is limited.

7.1.6 ELD compensation as passive summation in voltage domain with voltage DAC

Opposite to the active summation (Figure 7.1), passive summation is possible. Figure 7.6 shows the block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with passive summation in the voltage domain. The ELD DAC is implemented as a voltage DAC. By the RC network $R_1 - R_3$ and $C_1 - C_3$, the loop filter output and the ELD DAC output are summed at the quantizer input node. This ELD compensation approach has been demonstrated in [85].

This approach has pros and cons. The parasitic capacitance can be considered as (part of) C_3 . Thus, no parasitic pole is added at the summation node output and the passive voltage summation can achieve very large bandwidth. Another advantage is that the passive summation consumes no power. It has a few disadvantages: (1) The passive summation has no gain. The signal swing at the summation output node is typically small, which is related to the gain from the loop filter output to the ELD summation node and the coefficient of the ELD DAC (e.g. p_1 and d_1 in Figure 5.6), and the loop filter output swing. In the scenario with multi-bit quantizer, a small input swing is not preferable because it means more stringent quantization gain and offset requirement for the quantizer. (2) The parasitic pole caused by C_2 and the impedance of the ELD DAC (VDAC) limits the maximal speed of the ELD loop. (3)



Figure 7.6: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with passive summation in voltage domain.

 ${\cal R}_1$ and ${\cal C}_1$ add load to the loop filter, which increases its power consumption.

7.1.7 ELD compensation as passive summation in current domain with current-steering DAC

Passive summation can be implemented in the current domain as well. Figure 7.7a shows an exemplary CT $\Sigma\Delta$ ADC with ELD compensation with passive summation in current domain with a 1-bit quantizer. The loop filter output voltage is converted into current by a voltage-to-current (V/I) converter. The ELD DAC is implemented as a current steering DAC. The IDAC output current and Gm cell output current are summed in the current domain. The summation result is a current, which is taken as the input current by the 1-bit quantizer (comparator). This ELD compensation approach for SDMs with 1-bit quantization has been demonstrated in [86].

A passive ELD summation in the current domain has been demonstrated for a multibit CT SDM in [87]. Its block diagram is shown in Figure 7.7b. Similar as the 1-bit case, the loop filter output is converted to a current signal by a V/I converter. This current signal is summed with the ELD DAC output current signal in the current domain. The resulting current signal is converted into a voltage signal by a resistive ladder, and thus the reference information is included.

The passive summation in the current domain can achieve high bandwidth. However, the state-of-the-art approach shown in Figure 7.7b has two limitations. Firstly, the V/I and I/V conversions required in this approach add substantial delay, which is not suitable for the fast feedback loop in the multi-GHz sampling rate scenarios considered in this thesis. Another drawback is that the summation node drives 15 1-bit comparators in [87], which results in a large parasitic capacitance at the summation



Figure 7.7: Block diagram of exemplary state-of-the-art CT $\Sigma\Delta$ ADCs with ELD compensation with passive summation in current domain: (a) with 1-bit quantizer (b) with multi-bit quantizer.

output node. It generates a pole and adds extra delay in the fast feedback loop. To overcome these limitations, a novel current-mode multi-path ELD compensation technique is proposed in the next section.

7.2 Current-mode multi-path ELD compensation

This section proposes both system-level and circuit-level solutions of a current-mode multi-path ELD compensation technique for CT $\Sigma\Delta$ ADCs with multi-bit quantization and several GHz sampling rate. Thanks to the proposed solutions, the amplifier of the loop filter is not in the fast feedback loop; the delay of the pre-amplifier of



Figure 7.8: Conventional high speed SDM system architecture with multi-bit quantization and ELD compensation

the comparator is removed; and the effective regeneration time of the comparator latch is maximized. The proposed novelties enable CT $\Sigma\Delta$ ADCs with wide signal bandwidth and improved power efficiency. Extensive transistor-level simulations demonstrate their effectiveness and robustness. This section validates the proposed methods by transistor level design and simulations of an 8.4 GHz MASH $\Sigma\Delta$ ADC achieving an SQNR of 71 dB in a signal band of 600 MHz. This shows that the proposed solutions enable power-efficient multi-GHz $\Sigma\Delta$ ADC applications.

7.2.1 Introduction

The previous section has shown an overview for the high-speed ELD compensation techniques for CT $\Sigma\Delta$ ADCs, including ADCs with multi-bit quantization. Their limitations have been discussed, for ADC applications with multi-GHz sampling rate considered in this thesis. The conventional system architectures of high speed SDMs with multi-bit quantization and ELD compensation [16, 30, 63, 87] can be generalized as shown in Figure 7.8. The loop filter can be in either feedforward or feedback structure. Only one common fast feedback loop is used for the ELD compensation. In [16, 30, 63] the comparator input signal (Q_{in}) is compared to a series of reference voltages (V_{ref}) in the pre-amplifiers. In [87] pre-amplifiers are used as well. Besides their limitations discussed in the last section, in the scenario of a multi-GHz sampling frequency, the conventional system architecture has another problem – the pre-amplifiers have to be optimized for low delay, which limits the gain they can offer, while consuming a lot of power.

In this section, a novel current-mode multi-path ELD compensation scheme is proposed for very high speed CT $\Sigma\Delta$ ADCs with multi-bit quantization. The scheme is introduced via both system-level and circuit-level analysis, design and transistor-level simulations of an exemplary multi-GHz sampling CT $\Sigma\Delta$ ADC. The ELD DAC out-



Figure 7.9: Proposed high speed SDM system architecture with multi-bit quantization and current-mode multi-path ELD compensation

put and the loop filter output are summed in the current domain. The summation node is duplicated, which allows combining the reference information in the same nodes. The loop filter amplifier is not in the fast feedback loop. The delay of the preamplifier is removed, and the effective regeneration time of the comparator latch is maximized for an optimal gain. Section 7.2.2 describes the proposed system architecture. Section 7.2.3 presents the circuit implementation. Transistor-level simulation results are shown in section 7.2.4. Finally, conclusions are drawn in section 7.2.5.

7.2.2 System architecture

To exclude the loop filter amplifier from the fast feedback loop, and to remove the preamplifier delay, a high speed SDM system architecture with multi-bit quantization and current-mode multi-path ELD compensation is proposed, as shown in Figure 7.9. Similar to [87], with a V/I converter (Gm cell), the loop filter output is converted into a current signal, and then summed with the ELD DAC output current. The summation result is the comparator input signal (Q_{in}) in the current domain. The Q_{in} current signal should drive a multi-bit comparator, which requires duplicating the Q_{in} signal. Usually, this duplication can be implemented in two ways. The first way is to convert the Q_{in} current signal back to the voltage domain [87], which is not suitable for multi-GHz sampling scenarios. The second way is to duplicate the Q_{in} current signal with a current mirror. However, the BW requirement for the current mirror is very high, and it adds some delay.

The proposed solution (Figure 7.9) is to duplicate the summation node. The Gm

cell and the ELD DAC need to be duplicated as well. Instead of comparing the Q_{in} voltage signal with reference voltage signals V_{refi} in the pre-amplifiers (Figure 7.8), the corresponding reference current signals I_{refi} are added in the same summation nodes. Thus, the pre-amplifiers are removed from the system. The resulting current signal directly drives a comparator latch. In the SDM with (n+1)-level quantization shown in Figure 7.9, n Gm cells and n ELD DACs are required, where every ELD DAC is a (n + 1)-level DAC.

The novelty and advantages of the proposed system-level solution with respect to the state-of-the-art approaches [30, 16, 63, 87] are summarized as follows. Firstly, the state-of-the-art multi-bit SDMs with ELD compensation have only one fast feedback loop. The proposed system is the first ELD compensation based on multiple fast feedback loops. Compared with [30], the delay of the digital differentiator is removed from the fast feedback loop. Compared with [16] and [63], the loop filter amplifier is not in the fast feedback loop. Compared with [87], the delay of I/V conversion is removed. Moreover, the delay of the pre-amplifier is removed, and the effective regeneration time of the comparator latch is maximized for an improved quantization gain. Last, in Figure 7.8 the summation output Q_{in} needs to drive multiple preamplifiers. It implies that a large parasitic capacitance is on the node Q_{in} , which generates a pole and adds delay in the fast feedback loop. In Figure 7.9, the loop filter output drives multiple Gm cells. The parasitic capacitance on the loop filter output node generates a pole, but this pole is out of the fast feedback loops. Generally, the outer feedback loop is less sensitive to the extra delay than the fast feedback loops. A concern of the proposed system is that the number of the ELD DACs is increased, which need to be driven by the quantizer output V. Thus, the input capacitance of the ELD DACs should be minimized in the circuit design.

7.2.3 Circuit implementation

A transistor-level implementation of the proposed current-mode multi-path ELD compensation in a multi-bit SDM is shown in Figure 7.10. A Gm cell, an ELD DAC, the reference current, a latch block, the S&H circuit and buffers compose a slice of the ELD-compensated quantizer. The Gm cell is implemented as a source degenerated PMOS transconductance amplifier, which is composed of current sources I_s, a resistor R_s, a capacitor C_s, and thin-oxide PMOS transistors M₁, M₂. Source degeneration is employed to improve the linearity of the Gm cell for a large input swing of ± 250 mV. Capacitor C_s is added between the sources of M₁, M₂ to generate a zero in the transfer function of the V/I converter, and to compensate its delay. The bandwidth of the Gm cell affects the stability of the outer feedback loop, but not the fast feedback loop. The ELD DAC is implemented as a current steering DAC. NMOS switches M₃, M₄ and the current source I_{dac,u} compose a unit ELD DAC. NMOS transistors M₅, M₆ and the current source I_{ref,u} construct a reference current unit. For (n + 1)-level quantization, n reference levels need to be created, and thus (n - 1) reference current units are required for every slice. The complementary Sp[1]/Sn[1] - Sp[n-1]/Sn[n-1]



Figure 7.10: Proposed circuits of the SDM with multi-bit quantization and current-mode multi-path ELD compensation

can be configured to either Vdd (1.1 V) or gnd, which define the reference level of the slice. Among different slices, the proposed reference current circuits have the same common-mode current, and only their differential current is different. This property ensures that the circuits in different slices are working in the same common-mode condition.

After the latch block, a transmission gate (M_7, M_8) with complementary sampling clocks (clkp_{SH} and clkn_{SH}) is employed to sample the latch output in the end of the latch's regeneration phase, and to hold it on its output capacitance. Two buffers are inserted between the sampled latch output and the input of the ELD DACs. The choice of the number of buffers is a trade-off between minimizing the load capacitance of the latch, and reducing the delay of the buffers. The 1st buffer (M₉, M₁₀) is an NMOS buffer with PMOS load, while the 2nd buffer (M₁₁, M₁₂) is a CMOS buffer. The NMOS buffer has lower input capacitance than the CMOS buffer, but it consumes more power. The CMOS buffer offers rail-to-rail driving ability. For the (n + 1)-level quantization, one sampled latch output node should drive n unit ELD DACs after two buffers.

7.2.4 Simulation results

Without loss of generality, a CT feedback 2-1-1 MASH $\Sigma\Delta$ ADC with 3-bit quantization in every loop is considered to substantiate the proposed technique [79]. The sampling frequency is $F_S = 8.4$ GHz and the oversampling ratio is 7, to achieve a signal BW of 600 MHz. The proposed current-mode multi-path ELD-compensated 3-bit quantizers and the 2-1-1 MASH loop filter are designed at transistor-level in a 40 nm CMOS technology. The main DACs (all the other feedback DACs except for



Figure 7.11: Clock timing diagram and signal behavior of the proposed circuits

the ELD DACs) are modelled in Verilog-A, including the extra load for the quantizers caused by the main DACs.

Figure 8.7 shows the clock diagram and the typical behavior of some important signals from the schematic simulation. When the latch clock signal clkp_{ltc} is high, the comparator latch is in the regeneration phase. When it is low, the latch is reset, and the summation result of loop filter output, ELD DAC output, and reference current is transferred to the latch as the initial condition of the next regeneration. The 2nd row depicts the latch differential output signal. At the end of the regeneration phase, the sampling clock signal clkp_{SH} becomes high, and the latch output is sampled. clkp_{SH} should become low before resetting the latch. The 3rd and 4th rows show the input voltage and the output current of a unit ELD DAC, respectively. Since the total delay of the fast feedback loop should be less than 1 $T_{\rm S}$ (119 ps), the ELD DAC output current must settle before the starting of the next regeneration. Moreover, the total delay of the current summation and the connection circuit between the summation node and the latch must be very short, such that the ELD feedback information reaches the latch before the start of the next regeneration. In other words, at the starting moment of the next regeneration, which is the sampling moment of the quantizer, the ELD feedback information of the previous sample must already change the latch initial condition, and this change must be settled. In Figure 8.7, the ELD DAC output current is settled more than 40 ps before the next positive clock edge of clkp_{ltc}. It indicates that the total delay of the ELD feedback path is not limiting the maximal sampling rate. The duty cycle of the latch clock clkp_{ltc} is set to 62%, which maximizes the regeneration time of the latch. In this design, the



Figure 7.12: Simulated output spectrum of a 2-1-1 MASH $\Sigma\Delta$ modulator with the proposed system and circuits

maximal regeneration time of the latch is limited by the reset time and the time to acquire the next initial condition.

To verify the quantization gain of the transistor-level model of the proposed currentmode ELD-compensated 3-bit quantizers, the aforementioned 2-1-1 MASH $\Sigma\Delta$ ADC model was simulated for 134400 samples (16 us). In this model, the main DACs were modelled in Verilog-A such that if their differential input voltage is less than 316 mV, the unit DAC output cannot reach its full scale. It models 10 dB quantization gain of the main DAC switches, which is a pessimistic assumption. The transistor layout parasitic model was included in the simulation. Extra capacitance (20% of the total parasitic capacitance of that node) was added at the latch output nodes to model the increased parasitic capacitance because of the routing. Figure 8.8 depicts the simulated output spectrum (2¹⁷ samples) with -1.4 dBFS input signal. The simulated SQNR is 71 dB for BW = 600 MHz. Within the simulated 134400 samples of the main DAC output, no metastability error is observed.

The transistor-level simulations demonstrate the power efficiency of the proposed technique. The simulated power consumption of the ELD-compensated 3-bit quantizer in the 1st loop is 27.3 mW (3.9 mW per 1-bit quantization), including the Gm cells, ELD DACs, reference currents, latches, samplers and buffers. Figure 7.13 shows the simulated power breakdown of the ELD-compensated quantizer, in which "connection" indicates the power consumed by the connection circuit between the summation nodes and the latch blocks. 71% of the total power is used by the latches and the buffers, which provide the required quantization gain. The Gm cells, ELD DACs and reference currents only consume 22% of the power. Since the loop filter amplifier is not in the fast feedback loop, its GBW requirement is much relaxed. Inverter-based operational transconductance amplifiers (OTAs) were designed. Simulations show



Figure 7.13: Simulated power consumption breakdown of the quantizer with ELD compensation

that the GBW of the loop filter amplifier is 14.3 GHz, and its power consumption is $8.2~\mathrm{mW}.$

In a state-of-the-art ELD-compensated multi-bit $\Sigma\Delta$ ADC with similar BW (465 MHz), F_S (8 GHz) and SNDR (67 dB) in 28 nm CMOS technology, the power consumption of the quantizer per 1-bit quantization is 5.8 mW [16]. The loop filter amplifier in the fast feedback loop consumes 135 mW, and its GBW is about 63 GHz. Compared to this state-of-the-art [16], our proposed technique relaxes the GBW requirement of the loop filter amplifier by a factor of 4.4, since it is moved out from the fast feedback loop. Its power consumption is reduced by a factor of more than 16. The ELD-compensated quantizer itself is 33% more power efficient, since the delay of the pre-amplifier is removed and the effective regeneration time of the comparator latch is maximized.

7.2.5 Conclusions

A novel current-mode multi-path ELD compensation for multi-GHz sampling CT $\Sigma\Delta$ ADCs with multi-bit quantization is proposed to solve the problem due to the parasitic delay caused by the poles of the summing amplifier and the pre-amplifier. The loop filter amplifier is not in the fast feedback loop, and its GBW requirement is relaxed. The delay of the pre-amplifiers is removed, leaving more effective regeneration time for the comparator latch. For an exemplary 2-1-1 MASH SDM with 3-bit quantization, a BW of 600 MHz and SQNR of 71 dB are achieved. Transistor-level simulations show that the GBW requirement and power consumption of the loop filter amplifier are reduced by a factor of 4.4 and 16 with respect to existing state-of-the-art ELD compensation techniques. All these properties and simulated results show the potential of our proposed techniques for power-efficient broadband medium-resolution ADC applications.



Figure 7.14: Simplified block diagram of an exemplary CT SDM with ELD compensation and a generic summation node

7.3 Overcoming parasitic capacitance on summation node

Section 7.1.7 shows the state-of-the-art current-mode ELD compensation technique for CT SDMs with 1-bit quantization in Figure 7.7a. The previous section proposes a current-mode multi-path ELD compensation for CT SDMs with multi-bit quantization. These two approaches have one limitation for multi-GHz sampling rate $\Sigma\Delta$ ADC applications – they are sensitive to the parasitic capacitance on the summation node. The layout design and extraction of a 3-bit quantizer with the proposed current-mode multi-path ELD compensation show that the parasitic capacitance on the summation output node is very large (about 50 fF in 40 nm CMOS). The design choices that lead to this large parasitic capacitance value is elaborated in Chapter 11.1.7. This parasitic capacitance causes two problems. Firstly, it causes an incomplete settling of the ELD feedback information. Secondly, it causes a parasitic pole in the loop filter path at the Gm cell output. These two problems can degrade the performance of the SDM, or even make it unstable.

Another problem highlighted in this section is that a generic summation node is needed at the input of the quantizer in some SDM architectures for system-level optimization. It is needed in our 2-1-1 MASH architecture for the test chip implementation shown in Figure 5.6 as well. This generic summation node is missing or limiting the performance in the state-of-the-art approaches. This section proposes system-level and circuit-level solutions for the aforementioned problems.

7.3.1 System architecture

Figure 7.14 shows a simplified block diagram of an exemplary CT SDM with ELD compensation and a generic summation node. A generic summation node at the input of the quantizer is needed in some SDM architectures for the optimization of



Figure 7.15: Block diagram of an exemplary CT $\Sigma\Delta$ ADC with a generic summation node and the proposed ELD compensation technique with passive summation in current domain with a current-steering DAC with a feedforward capacitor, and gm stage with a zero: (a) for 1-bit quantization (b) for multi-bit quantization.

the thermal noise at the internal nodes of the loop filter, or for the optimization of the signal transfer function, or required by the feedforward loop filter architecture. In Figure 7.14, the node X can be the SDM input signal U, or an internal node in the loop filter.

Figure 7.15a and 7.15b show the block diagram an exemplary CT $\Sigma\Delta$ ADC with a



Figure 7.16: Proposed schematics of the SDM with multi-bit quantization, a generic summation node with current-mode multi-path ELD compensation with a feedforward capacitor, and Gm cell with a zero.

generic summation node and the proposed ELD compensation technique for 1-bit and multi-bit quantization, respectively. The generic summation node is implemented as the current summation. As shown in Fig 7.15a and 7.15b, the key features of the proposed approach are twofold. Firstly, in parallel with the current steering ELD DAC (IDAC), a feedforward capacitor C_{ff} is added between the input and output nodes of the DAC, to improve the settling of the ELD feedback information. Secondly, an additional zero is built in the Gm cell. This zero compensates the phase shift of the parasitic pole caused by the parasitic capacitance C_p and g_m . Thus, the problems caused by the parasitic capacitance C_p on both ELD path and loop filter path are compensated.

7.3.2 Circuit implementation

The transistor-level circuit implementation of the proposed SDM system is shown in Figure 7.16. Here an SDM with multi-bit quantization and the proposed generic summation node and ELD compensation approach is shown as an example, and the SDM with 1-bit quantization can be implemented in a similar way. It shows one slice of the duplicated ELD compensation loop and 1-bit comparator in detail. Here the SDM is assumed to use (n + 1)-level quantization. So the SDM has totally *n* slices of duplicated ELD compensation loop and 1-bit comparator. The parasitic capacitance C_p on the summation nodes is explicitly drawn.

The Gm cell with zero, reference current, S&H circuit and buffers are implemented in the same way as described in Section 7.2.3. Compared to Figure 7.10, it has two differences, besides the explicitly-drawn parasitic capacitance C_p : (1) It contains two Gm cells with zero here, one for the loop filter output Vi^+ and Vi^- , the other for the generic summation input X. (2) In every 1-bit current steering DAC, two crosscoupled feedforward capacitors C_{ff} are added (shown in green color in Figure 7.16). One capacitor C_{ff} is added between D^- and Io^+ , while another capacitor C_{ff} is added between D^+ and Io^- . Notice that the Io^- node is connected to sum^+ node (strictly speaking they are the same node), while the Io^+ node is connected to $sum^$ node. The sum^+ and sum^- nodes are the summation nodes of the Gm cell output current, ELD DAC output current and the reference current. They are at the same time the input of the latch block.

7.3.3 Simulation results

Figure 7.17 shows the transient simulation results on the ELD feedback settling, which confirms the effectiveness of the proposed system and circuit. In this simulation, only the ELD feedback signal settling on the summation output nodes are shown, and the Gm cell input voltages are connected to its common-mode voltage. Figure 7.17a shows the transient behavior of the voltages of sum^+ and sum^- nodes, with and without a parasitic capacitance C_p . The value of C_p comes from the post-layout extraction, which is 50 fF for the designed ADC test chip. The solid red and blue lines show sum^+ and sum^- voltages without the parasitic capacitance C_p . The settling of the summation node voltages is very fast. However, when there is a large parasitic capacitance C_p on the summation nodes, the settling becomes much slower, shown as the dashed green and magenta lines. In a target application of a 7 GHz sampling SDM, the available setting time between the switching of the ELD DAC input (D^+, D^-) and the sampling moment of the comparator latch is about 30 ps. The sampling moment is indicated in Figure 7.17a as the dashed black line. In this case, the differential voltage on the summation node $(sum^+ - sum^-)$ should be switching between -13.4 mV to +13.4 mV. Without C_p , at the sampling moment the ELD feedback signal on the summation node is fully settled. However, in the case with a large parasitic capacitance C_p , at the sampling moment, the ELD feedback signal on the summation node is only 78.7% settled. In this case, the delay from the crossing-point of D^+ and D^- nodes, to 95% settling of the sum^+ and sum^- nodes is 61 ps.

Figure 7.17b shows the effect of the feedforward capacitance C_{ff} . With the same parasitic capacitance C_p , when the feedforward capacitance C_{ff} is added, which is 1.35 fF in this design, at the sampling moment the ELD feedback signal on the summation node is 100.4% settled. The settling of the ELD feedback signal is a bit overshot because C_{ff} is a little bit too large in this case. Of course the SDM stability and resolution is not sensitive to the residue ELD settling error of 0.4%. In this case, the delay from the crossing-point of D^+ and D^- nodes, to 95% settling of the sum⁺ and sum⁻ nodes is reduced to 13 ps.

The functionality of the proposed Gm cell with zero is checked by AC simulation,



Figure 7.17: Transient simulation results on the ELD feedback settling: (a) sum^+ and sum^- voltages with and without C_p (b) sum^+ and sum^- voltages with C_p , with and without C_{ff}

which is shown in Figure 7.18. Figure 7.18a shows the magnitude of the transfer functions from the loop filter output node $Vi^{+/-}$ to the summation node $sum^+/-$, with and without the zero (generated by the capacitor C_{s1} with a value of 20 fF). Figure 11.5b shows the phase correspondingly. At $F_S/2$, which is 3.5 GHz in this design, the phase shift without C_{s1} is -6.5°. With C_{s1} which creates a zero in the transfer function of the Gm cell, the phase shift at $F_S/2$ is -1.3°. Please notice that if C_{s1} is further increased the phase can be further boosted up at $F_S/2$, but the phase between 500 MHz – 3 GHz will be higher than the desired value of 180°.

A 2-1-1 MASH SDM with 3-bit quantization is designed at transistor and layout level in a 40nm CMOS technology. In this SDM, the 3-bit quantizers, generic summation node, current-mode multi-path ELD compensation with a feedforward capacitor, and Gm cell with a zero are implemented at layout level. The other parts of the SDM are modeled at transistor level (schematic). Figure 7.19 shows the simulated output



Figure 7.18: AC simulation results on the transfer function of the gm cell with and without zero (Cs1): (a) magnitude from Vi to sum (b) phase from Vi to sum

spectrum. The red spectrum is before a calibration and the blue spectrum is after a digital calibration for the analog loop filter and digital NCF mismatch (this calibration is discussed in Chapter 6). After the Matlab calibration, the SDM achieves a SQNR of 66.2 dB for 500 MHz bandwidth. It shows that the proposed technique is verified with post-layout simulations.

7.3.4 Conclusions

In this section, a system-level and circuit-level approach to overcome the limitation caused by the parasitic capacitance on the summation node in the current-mode ELD compensation is proposed. Moreover, a generic summation node is added at the input of the quantizer to facilitate SDM system-level optimization. Transistorlevel and post-layout simulations show the effectiveness of the proposed approach. Simulation results shows that the settling time of the ELD feedback signal at the



Figure 7.19: Simulated output spectrum of a 2-1-1 MASH SDM with the proposed generic summation node, ELD compensation loop and quantizer in post-layout

summation node output is reduced from 61 ps to 13 ps thanks to the feedforward capacitance, which allows higher sampling rate and broader BW for the SDMs. The parasitic pole at the loop filter path is compensated by the added zero in the Gm cell. The proposed system and circuit enable CT $\Sigma\Delta$ ADCs with higher sampling rate and broader signal BW.

7.4 Conclusions

This chapter studies the high-speed ELD compensation techniques. The state-of-theart high-speed ELD compensation techniques are firstly studied and classified, which shows that the speed limitation of the ELD compensation is a bottleneck for multi-GHz sampling multi-bit $\Sigma\Delta$ ADCs. This chapter further proposes a current-mode multi-path ELD compensation for multi-GHz sampling multi-bit $\Sigma\Delta$ ADCs. The loop filter amplifier is not in the fast feedback loop, and its GBW requirement is relaxed. The delay of the pre-amplifiers is removed, leaving more effective regeneration time for the comparator latch. However, a large parasitic capacitance at the summation node is observed in the post-layout extraction, which limits the speed of the SDM. This chapter further proposes solutions to overcome the limitation caused by this parasitic capacitance. Cross-coupled feedforward capacitance C_{ff} can boost up the settling of the ELD feedback signal. And the phase shift in the loop filter path can be compensated by adding a zero in the Gm cell. Transistor-level and post-layout simulations show the effectiveness of the proposed techniques. Test chip measurement results are shown in Chapter 11. All these properties and results show the potential of our techniques for power-efficient broadband medium-resolution ADC applications.

Chapter 8

Comparators for broadband CT $\Sigma \Delta$ ADCs

This chapter presents two new techniques for high-speed comparators in broadband $CT \Sigma \Delta$ ADCs. In conventional $\Sigma \Delta$ ADCs, the comparator reset time restricts the maximal achievable ADC sampling rate and signal BW. Firstly, this chapter introduces a current-mode locally-time-interleaved multi-bit quantizer in Section 8.1, to move the comparator reset time out of the 1 T_S time budget. Thus, the ADC sampling rate and the signal BW can be further increased. The current-mode multi-path ELD compensation technique introduced in Section 7.2 has no gain in the summation, since it is passive. The offset originating from the comparator latch is amplified when referred to the loop filter output node, which reduces the ADC resolution, and has to be calibrated. A comparator offset calibration for the current-mode multi-path locally-time-interleaved ELD-compensated quantizer is proposed in Section 8.2. Conclusions are drawn in the end.

8.1 Current-mode locally-time-interleaved multibit quantizer

This section proposes a current-mode locally-time-interleaved (TI) multi-bit quantizer for CT $\Sigma\Delta$ ADCs with multi-GHz sampling rate. This technique allows for moving the comparator reset time out of the one sampling clock period time budget [24, 84, 88]. Thus, the ADC sampling rate can be further increased with the same requirement on the quantizer metastability, hence achieving a broader signal BW.

8.1.1 Motivation for quantizer time-interleaving

For multi-bit multi-GHz sampling CT $\Sigma\Delta$ ADCs, maximizing the effective regeneration time of the comparator referring to the ADC sampling period is essential to increase the ADC sampling rate and signal BW, while satisfying the requirement on the quantizer metastability. It requires high-speed ELD compensation [69, 77] and highspeed comparators. The state-of-the-art high-speed ELD compensation techniques have been reviewed in Chapter 7.1. In the conventional SDM system architecture with multi-bit quantization and ELD compensation shown in Figure 7.8, typically up to one sampling clock period (T_S) ELD is compensated.

Figure 8.1a shows an illustrative timing diagram for a broadband CT $\Sigma\Delta$ ADC without quantizer time-interleaving. Within 1 T_S, the multi-bit quantizer needs to complete three functions – regeneration (denoted as T_{reg}), reset (denoted as T_{rst}), and tracking the next initial condition (referred to as "input settling" and denoted as T_{set} in this chapter). The reset phase and the input settling phase can be either combined or separated: (1) If the reset phase and the input settling phase are combined, it means the latch input/output nodes are tracking the next initial condition when the latch is reset. In this case, the initial condition stored at the latch input/output nodes is relatively small. (2) If the reset phase and the input settling phase are separated, in the reset phase, the latch input/output nodes are reset. In the input settling phase, the reset switches are OFF, and the latch input/output nodes track the next initial condition. In this case, the initial condition stored at the latch input/output nodes is larger, which is beneficial for a higher quantization gain and a lower input referred comparator offset. However, it requires more complicated clock phases, and both of the reset time and the input settling time are shorter.

The minimum achievable value of the latch's regeneration time constant $\tau_{reg,min}$ is defined by the technology. Even if more power is spent by the latch, its time constant cannot be further reduced because self-loading is dominant. To achieve a certain DR (or peak SNDR) target for a certain BW, a minimum quantization gain requirement $G_{q,min}$ (in [dB]) can be derived, which should be generated by the latch (see Figure 5.9). Thus, the minimum effective regeneration time can be calculated as:

$$T_{reg,eff,min} = \log(10^{\frac{G_{q,min}}{20}}) \cdot \tau_{reg,min}$$
(8.1)

In the end of the regeneration, the latch's output is sampled and passed to the main DAC and the ELD DAC. To guarantee that the latch's output is properly sampled before the resetting of the latch, the starting of the latch output sampling should be at least τ earlier than the starting of reset phase (this time τ is referred to as "sampling to reset delay" and denoted as $\tau_{smp-rst}$ in this chapter). With the current-mode multi-path ELD compensation technique proposed in Chapter 7.2, the ELD feedback path has less delay than the main feedback path. The propagation delay from the quantizer to the main DAC is denoted as τ_{prop} , and the delay of the main DAC itself is denoted as τ_{DAC} . The sampling aperture of the DAC sampler is considered as part of the DAC delay.



Figure 8.1: (a) Illustrative timing diagram without time-interleaving (b) Illustrative timing diagram with time-interleaving.

The reset time and input settling time should be separately optimized. Reducing the reset time forces to increase the size of the reset switch, to meet the requirement on the comparator hysteresis. However, sizing up the reset switch adds more load to the latch input/output node, which increases the regeneration time constant of the latch, and reduces its quantization gain.

In Figure 8.1a, the minimal achievable ADC sampling period $T_{S,min}$ is determined by the maximal time (or delay) of two paths:

$$T_{path1} = T_{reg,eff,min} + \tau_{smp-rst} + T_{rst} + T_{set}$$

$$(8.2)$$

$$T_{path2} = T_{reg,eff,min} + \tau_{prop} + \tau_{DAC}$$

$$(8.3)$$

$$T_{S,min} = max\{T_{path1}, T_{path2}\}$$

$$(8.4)$$

The reset time T_{rst} and the input settling time T_{set} can be moved out of the 1 T_S time budget by the quantizer time-interleaving [24, 84, 88]. When T_{path1} is longer than T_{path2} , it indicates that if quantizer time-interleaving is applied, $T_{S,min}$ can be reduced.

Schematic design and simulations in 40 nm CMOS technology has been done to es-

timate the maximal sampling rate for a 3-bit SDM with an SNDR target of at least 60 dB, with the SDM architecture shown in Chapter 5. Without quantizer time-interleaving, the minimal sampling period T_S is estimated to be about 160 ps. The latch's time constant is optimized to be 5.5 ps from schematic design. From the starting of the latch regeneration to the sampling of the latch's output, the effective regeneration time $(T_{reg,eff})$ should be at least 70 ps to generate the required quantization gain. The sampling to reset delay $(\tau_{smp-rst})$ is designed to be 10 ps. The propagation delay from the quantizer to the main DAC and the delay of the main DAC itself are about 50 ps together. The reset time and input settling time are optimized to be 40 ps each. From simulation, $T_{path1} = 160$ ps, and $T_{path2} = 120$ ps. Since $T_{path1} > T_{path2}$, the quantizer time-interleaving can be applied in this case to minimize T_S .

Figure 8.1b presents an illustrative timing diagram for a broadband CT $\Sigma\Delta$ ADC with quantizer time-interleaving. Latch 1 and Latch 2 are working in a time-interleaved way [24, 84, 88]. When one latch is in the regeneration phase, the other latch is in the reset and input settling phase. Thus, the reset and input settling time is moved out of the 1 T'_S time budget (T'_S in Figure 8.1b is smaller than T_S in Figure 8.1a), allowing the whole 1 T'_S time for the comparator regeneration. The regeneration time, reset and input settling time are both doubled referred to the ADC sampling period. The minimal T'_S is defined by T_{path2} only. With quantizer time-interleaving, the minimal achievable sampling period T'_S is estimated to be about 120 ps (based on schematic simulation results). Quantizer time-interleaving can enable 1.33× higher ADC sampling rate, and hence 1.33× broader signal BW.

In general, time-interleaving has advantages if the required performance (which is speed in the context of this chapter) is at the edge of what the technology can deliver. It means that the power dissipation increases stronger than linear with speed (e.g. to reduce the regeneration time constant of the latch). In another words, the speed is very costly there, and the properties of the technology are not used then efficiently.

The disadvantage of time-interleaving is that the system has different paths. The matching errors in both the function that is performed (e.g. gain, delay) and the differences in the (local) error signals like offsets, noise and interferences contribute to the total error of the system. The matching errors in the function cause modulation at half-clock frequency, leaving up to half-clock undesired signals ('pattern noise' in case of modulated DC-offsets), down-converted (down-mixing) noise, and modulated transfer functions (part of STF and NTF modulated at half clock). In the context of the quantizer time-interleaving in $\Sigma\Delta$ ADCs, the difference in the comparator offsets between the TI quantizers down-converts the out-of-band quantization noise into the signal band, and increases the in-band noise. This effect is presented in Section 8.1.3.

	Globally-Tl	Locally-Tl	
Voltage-	Voltage-mode	Voltage-mode	
mode ELD	globally-Tl	locally-Tl	
current-	current-mode	current-mode	
mode ELD	globally-Tl	locally-Tl	

Figure 8.2: Quantizer time-interleaving implementation alternatives.

8.1.2 Quantizer time-interleaving implementation alternatives

Various quantizer time-interleaving techniques can be found in the literature [24, 84, 89, 88]. Figure 8.2 presents a two-dimensional classification on the implementation alternatives of quantizer time-interleaving techniques. First, the ELD compensation can be implemented either in the voltage-mode [88] or in the current-mode [81]. Second, the time-interleaving can be applied either at the multi-bit quantizer level (referred to as "globally-TI"), or at the 1-bit comparator level (referred to as "locally-TI"). The voltage-mode globally-TI quantizer has been proposed in [89, 88], and its block diagram is shown in Figure 8.3. The current-mode multi-path ELD compensation proposed in Chapter 7.2 has many advantages over the voltage-mode ELD compensation techniques. Thus, an implementation of the quantizer time-interleaving which is compatible with the current-mode ELD compensation is proposed in this thesis. A current-mode globally-TI quantizer requires to convert the current output from the ELD summation back into a voltage signal. The V/I and I/V conversions add substantial delay, which makes it not suitable for the ELD feedback loop in the multi-GHz sampling rate scenarios (discussed in Chapter 7). This thesis proposes a currentmode locally-TI quantizer shown in Figure 8.4, which will be discussed later. Since a current-mode ELD compensation has been chosen in this design (Chapter 7.2), a voltage-mode locally-TI quantizer (which is not compatible with the current-mode ELD compensation) has not been investigated in this thesis.

In an SDM with the voltage-mode globally-TI quantizer (Figure 8.3), the quantizer input voltage (V_{in}) needs to be switched to either multi-bit quantizer input at half SDM sampling frequency $(\frac{1}{2} F_S)$. The input parasitic capacitance (C_p) of a *n*-bit quantizer is $(2^n - 1) \times$ of the input parasitic capacitance of one of its composing comparators. The parasitic capacitance C_p needs to be charged in this switching period. The resistance R_{ON} of the switches and C_p define the charging time constant, which limits the speed of the quantizer input settling. The input settling time is part of the total ELD, which eventually limits the minimal achievable T_S .

In an SDM with the proposed current-mode locally-TI quantizer (Figure 8.4), each 1-bit comparator is locally time-interleaved with two comparator latches with current



Figure 8.3: Block diagram of an ELD-compensated multi-bit SDM with the voltagemode globally-TI quantizer [88, 89].



Figure 8.4: Block diagram of an ELD-compensated multi-bit SDM with a proposed current-mode locally-TI quantizer.

input, on top of the current-mode multi-path ELD compensation technique in Figure 7.9. The input current $(I_{in,n})$ of the TI comparators is switched to latch 1 when Φ_1 is high, and it is switched to latch 2 when Φ_2 is high. Since the time-interleaving happens in the current domain, R_{ON} of the TI switches is in series with the large output impedance $R_{out,gm}$ of the Gm cell, and R_{ON} is much lower than $R_{out,gm}$. The time constant is determined by the transconductance g_m of the Gm cell, and the input parasitic capacitance of the latch. Since R_{ON} of the TI switches does not determine the time constant, its requirement is much relaxed compared to the voltage-mode TI. This is beneficial for high-speed operation.

In general, it is more advantageous for high-speed operation to implement both of the ELD summation and the time-interleaving in the current domain for the following two

reasons: (1) summation is easier to be implemented in the current domain than in the voltage domain; (2) in the current domain, the requirement on the R_{ON} of the switches is relaxed. To benefit from the speed advantages offered by the combination of the current-mode multi-path ELD compensation and the quantizer time-interleaving, the proposed current-mode locally-TI quantizer technique is the preferred way for multi-GHz-sampling $\Sigma\Delta$ ADCs.

8.1.3 Sensitivity to comparator offset

The current-mode multi-path ELD compensation technique proposed in Chapter 7.2 has no gain in its summation, since passive summation is used. The offset originating from the comparator latch is amplified when referred to the loop filter output node. Thus it is important to check how sensitive is the SNDR of the SDM to the comparator offset, in the scenarios with and without the quantizer time-interleaving.

Figure 8.5 presents the simulated SNDR versus the standard deviation $\sigma_{V_{OS}}$ of the comparator offset V_{OS} , with and without the quantizer time-interleaving. The 2-1-1 MASH SDM architecture in Chapter 5 is modelled in Matlab/Simulink. $\sigma_{V_{OS}}$ in this simulation is referred to the quantizer's input full-scale (V_{FS}). For every σ_{VOS} , 50 independent simulations are run. Some statistics of the simulated SNDR are plotted, such as minimal, maximal, mean values (red line) and 10%, 90% and middle values (blue line). When $\sigma_{V_{OS}}$ is 10%, the simulated SNDR is 59.7 \pm 8 dB ($\pm \sigma$) without the quantizer time-interleaving, which is limited by the comparator offset (Figure 8.5a). With the quantizer time-interleaving, the simulated SNDR is 57.8 \pm 9.5 dB $(\pm \sigma)$ for the same $\sigma_{V_{OS}}$ (Figure 8.5b). The SNDR is a bit more sensitive to the comparator offset, since the out-of-band quantization noise is mixed down in-band due to the different offset of the TI comparators. Fortunately this increased in-band quantization noise due to mixing down is shaped by the NTF, nevertheless the SNDR is in average reduced by 1.9 dB when $\sigma_{V_{OS}}$ is 10%. In both cases, to ensure that the SDM's SNDR is not limited by the comparator offset, σ_{VOS} should be less or equal to 2% of the quantizer's input full-scale. Notice that the simulation results shown in Figure 8.5b are generally valid for SDMs with quantizer time-interleaving, regardless of the implementation alternatives shown in Section 8.1.2.

8.1.4 Circuit implementation

A transistor-level implementation of the proposed current-mode locally-TI multi-bit quantizer in a SDM is shown in Figure 8.6. A Gm cell, an ELD DAC, a reference DAC, two TI comparators, the S&H circuit and buffers compose a slice of the ELDcompensated quantizer. The Gm cell, ELD DAC, reference DAC, and buffers are discussed in Chapter 7.2.

The transistors M_7 - M_{22} compose one locally-TI comparator with current input. The



Figure 8.5: Simulated SNDR versus the standard deviation of the comparator offset σ (a) without the quantizer time-interleaving (b) with the quantizer time-interleaving.

transistors $M_9 - M_{12}$ and $M_{17} - M_{20}$ compose two NMOS latches. In the following description, the latch composed of $M_9 - M_{12}$ is referred as latch 1, and the latch composed of $M_{17} - M_{20}$ is referred as latch 2. The transistors M_7 , M_8 , $M_{13} - M_{16}$, M_{21} and M_{22} work as the switches between the summation nodes (sum⁺ and sum⁻) and the latch input/output nodes (Vo1^{+/-} and Vo2^{+/-}). They are used as the reset switches of the latches as well. The input current is transformed to voltage via the ON resistance of M_8 , M_{14} , M_{16} and M_{22} . Two complementary clock signals $clkp_{ltc}$ and $clkn_{ltc}$ are needed.

During the reset and input settling phase, the latch output nodes are set to a commonmode voltage plus a small differential voltage. For example, when latch 1 is in reset and input settling phase, the switches M_7 , M_8 , M_{13} , M_{14} are ON. The differential output current of the summation nodes (sum⁺ and sum⁻) is converted to a differential voltage on the latch input/output nodes (Vo1^{+/-}) by the ON resistance of M_8/M_{14} .



Figure 8.6: Proposed circuits of the SDM with locally-TI multi-bit quantizer.

The common-mode voltage on the latch input/output nodes (Vo1^{+/-}) is much lower than the threshold voltage of the NMOS, to turn off the positive feedback of the NMOS latch. Thus, no extra tail NMOS or PMOS transistor is required to turn off the latch. During the regeneration mode, the Drain-to-Source voltage V_{DS} is maximized and the transconductance (g_m) of the NMOS M₁₀/M₁₂/M₁₈/M₂₀ is maximized as well. Moreover, at the beginning of the regeneration, a large peak current is required to charge the internal nodes of the latch, such that the NMOS M₁₀/M₁₂/M₁₈/M₂₀ turn ON and the positive feedback starts. This peak current is not limited by the removed tail NMOS or PMOS transistor in this design. It is defined by the PMOS M₉/M₁₁/M₁₇/M₁₉.

A concern of the designed NMOS latch is that it has static current flowing on one branch, even when the latch output nodes are settled. This static current can be turned off when the decision has been made by the latch with the self-cut-off comparator technique presented in [34], at the cost of some extra load on the latch output nodes which reduces the speed of the latch. In this design, the self-cut-off comparator technique is not used. From schematic simulation results, an NMOS latch with PMOS load is employed to achieve 8.4 GHz sampling rate with more than 120 dB of quantization gain at the cost of 0.91 mW power per comparator latch.



Figure 8.7: Clock timing diagram and some internal nodes of the proposed circuits: (a)–(b) latch clock and sampler clock; (c)–(d) latch 1 and latch 2 outputs; (e) sampled quantizer outputs after buffers; (f) unit ELD DAC output current.

After the TI comparators, two transmission gates $(M_{23} - M_{26})$ with two complementary sampling clocks $(clkp1_{SH}/clkn1_{SH}, clkp2_{SH}/clkn2_{SH})$ are employed to sample the TI comparators output at the end of the corresponding regeneration phase, and to hold it on their common output capacitance. The samplers $(M_{23} - M_{26})$ work as a multiplexer.

8.1.5 Simulation results

Without loss of generality, a CT feedback (FB) 2-1-1 MASH $\Sigma\Delta$ ADC with 3-bit quantization in every loop is considered to substantiate the proposed technique [79]. The sampling frequency is $F_S = 8.4$ GHz and the oversampling ratio is 7, to achieve a signal BW of 600 MHz. The proposed locally-TI 3-bit quantizers, the 2-1-1 MASH loop filter and the FB DACs are designed at transistor-level in a 40 nm CMOS technology.

Figure 8.7 shows the clock diagram and the internal nodes of the quantizer and ELD DAC from the schematic simulation. The latch clock signals $clkp_{ltc}$ and $clkn_{ltc}$ are at the half of the SDM sampling rate (4.2 GHz), and they are with 50% duty cycle.

Figure 8.7c and 8.7d depict the differential output signal of the TI comparators. At the end of the regeneration phase, the corresponding sampling clock signal $clkp1_{SH}$ (or $clkp2_{SH}$) becomes high, and the comparator output is sampled. $clkp1_{SH}$ (or $clkp2_{SH}$) should become low before resetting the comparator in order to hold the data. In Figure 8.7c, (1) indicates the reset and the input settling time, which is 1 T_S (119 ps). (2) indicates that at the beginning of the regeneration phase of the latch 1, the common-mode voltage of Vo1^{+/-} settles to about $\frac{1}{2}V_{dd}$. (3) indicates the exponential settling of the latch outputs $Vo1^{+/-}$. Although the regeneration time when the sampler is disconnected from the latch is 0.54 T_S (discussed in detail later), the exponential settling phase is typically much less than $0.54 T_S$ since the absolute value of the initial condition is larger than $\frac{1}{G_q}$ in most of the cases (G_q is the quantization gain of the latch). Figure 8.7e and 8.7f show the input voltage and the output current of a unit ELD DAC, respectively. Since the total delay of the fast FB loop should be less than 1 T_S , the ELD DAC output current must settle before the starting of the next regeneration of either comparator. In Figure 8.7f, the ELD DAC output current is settled about 15 ps before the next positive clock edge of $clkp_{ltc}$ or $clkn_{ltc}$, which is a tradeoff between maximizing the effective regeneration time of the latch, and the robustness of the ELD FB loop. The ELD FB loop is robust in this design, since the ELD FB loop can still be closed when the total additional delay caused by e.g. PVT variation, device mismatch, and layout-dependent effect is less than 15 ps. The peaks indicated by the arrows, caused by the clock feedthrough, do not degrade the resolution of the SDM.

In this design, the time between the starting of the regeneration, and the starting of the sampling of the latch output, is optimized to be 54% of T_S. During this time period, the latch in the regeneration phase is disconnected from the summation nodes, and its samplers are OFF. Thus the load on the latch output nodes (Vo1^{+/-} or Vo2^{+/-}) is minimized, and the time constant τ of the latch is minimized. The simulated time constant τ of the latch at schematic level is 5.5 ps with load, and the latch offers the majority part of the total quantization gain. When clkp1_{SH} (clkp2_{SH}) is high, which is optimized to be 34% of T_S for this design, the corresponding pass gates turn ON, and the latch output is sampled at the parasitic capacitance on S^{+/-} nodes. In this time period, although the load on the latch output nodes is increased, the positive feedback of the latch can still produce some quantization gain, if the latch output nodes are not settled. The total effective regeneration time is 0.88 T_S.

The aforementioned 2-1-1 MASH $\Sigma\Delta$ ADC was simulated for 134400 samples (16 us). The transistor layout parasitic model was included in the simulation. Extra capacitance (20% of the total parasitic capacitance of that node) was added at the latch output nodes to model the routing parasitic capacitance. Figure 8.8 depicts the simulated output spectrum (2¹⁷ samples) with -3.1 dBFS input signal. The simulated SQNR is 66.3 dB for BW = 600 MHz. Within the simulated 134400 samples of the main DAC output, no metastability error is observed. The transistor-level simulations demonstrate the power efficiency of the proposed technique. The simulated power consumption of the ELD-compensated 3-bit quantizer in the 1st loop is 28 mW (4).



Figure 8.8: Simulated output spectrum of a 2-1-1 MASH $\Sigma\Delta$ modulator with the proposed system and circuits.

mW per 1-bit quantization). Monte-Carlo simulations with random mismatch show that the total input-referred offset has a standard deviation $\sigma = 3$ LSB (LSB is the least significant bit of the 3-bit quantizer), which requires calibration to maintain the ADC resolution. A comparator offset calibration for this current-mode multi-path TI quantizer is proposed in Section 8.2.

Table 8.1 compares this work to state-of-the-art $\Sigma\Delta$ ADCs. Thanks to the proposed current-mode locally-TI quantizer, the quantizer effective regeneration time is significantly increased from 0.54 T_S [16] to 0.88 T_S, which increases the quantization gain by several tens of dB. The test chip measurement results with the proposed technique will be presented in Chapter 11.

8.1.6 Conclusions

A novel current-mode locally-TI quantizer is proposed for multi-GHz sampling CT $\Sigma\Delta$ ADCs. It is compatible with the current-mode multi-path ELD compensation technique presented in Chapter 7.2. Applying quantizer time-interleaving can enable $1.33 \times$ higher ADC sampling rate and signal bandwidth. In proposed current-mode TI quantizer, the R_{ON} of the TI switches does not determine the time constant, and hence the R_{ON} requirement is relaxed compared to the voltage-mode TI quantizer. The reset time and the time to track the next initial condition of the comparator are moved out of the 1 T_S ELD time budget. For an exemplary 2-1-1 MASH SDM with 3-bit quantization, a BW of 600 MHz and an SQNR of 66.3 dB are demonstrated by transistor level simulations in 40 nm CMOS technology. All these properties and simulated results show the potential of the proposed technique for broadband ADC

	This work	[88]	[84]	[16]
Quantizer resolution	8-level	14-level	2-level	17-level
TI quantizer?	Current-mode TI	Voltage-mode TI	Voltage-mode TI	No
Effective reg. time	$0.54 {\rm ~T}_S + 0.34 {\rm ~T}_S$	N.A.	$0.5~\mathrm{T}_S$	$0.54~{\rm T}_S$
Sampling rate (F_S)	8.4 GHz	$5~\mathrm{GHz}$	$6~\mathrm{GHz}$	8 GHz
Signal BW	600 MHz	$156 \mathrm{~MHz}$	$60 \mathrm{~MHz}$	$465 \mathrm{~MHz}$
SNDR	66.3 dB	66.5 dB	$67.6~\mathrm{dB}$	$67~\mathrm{dB}$
1-bit quant. power	4 mW^{a}	$3.2 \mathrm{~mW^{b}}$	N.A.	5.8 mW^{c}
CMOS technology	40 nm	40 nm	65 nm	28 nm
Note	simulation	measurement	measurement	measurement

Table 8.1: Comparison to state-of-the-art Σ Δ ADCs

^a Including Gm cells, ELD DACs, reference DACs, 2x TI comparators, samplers and buffers ^b Calculated from [88], excluding ELD compensation

^c Comparator power, excluding ELD compensation

applications.

8.2 Comparator offset calibration

This section proposes an on-chip comparator offset calibration for broadband CT $\Sigma\Delta$ ADCs with current-mode multi-path ELD compensation (Section 7.2) and locally-TI multi-bit quantizer (Section 8.1).

8.2.1 Introduction

Section 7.1 presents an overview for high-speed ELD compensation techniques, including the active summation and the passive summation. Section 7.2 proposes a current-mode multi-path ELD compensation technique. The ELD compensation techniques with passive summation have one drawback – no gain can be made with passive summation, and thus the input swing of the multi-bit quantizer is reduced, which gives stringent requirements on the comparator offset to not degrade the DR and linearity of the ADC. The circuit implementation of the current-mode multi-path ELD compensation presented in Section 7.2 has an attenuation of 18 dB from the loop filter output to the sampled initial condition of the comparator latch, resulting from the ELD coefficient and an optimization for low phase shift. Thus, the offsets originating from the comparator latches are amplified by 18 dB when referred to the loop filter output.

Generally the offset requirement can be achieved either by intrinsic matching or by calibration. For the test chip designed in this thesis, since the current-mode multipath ELD compensation and the locally-TI multi-bit quantizer are applied, the preamplifier is removed since the reference currents are added on the ELD summation nodes. Transistor-level Monte-Carlo simulations show that the total input-referred (referred to the loop filter output node) comparator offset has a standard deviation $\sigma = 3$ LSB (LSB is the least significant bit of the 3-bit quantizer), which is dominated by the comparator latches (M₉ - M₁₂ and M₁₇ - M₂₀ in Figure 8.6). If the offset requirement is chosen to be achieved by the intrinsic matching, which requires $\sigma \leq \frac{1}{6}$ LSB, the gate area of the transistors in the comparator core should be $324 \times$ larger ($18^2 = 324$), to reduce the offset standard deviation σ by $18 \times [71]$. The parasitic capacitance of such huge transistors would reduce the speed and increase the power consumption of the comparator significantly. Thus, this thesis concludes that the comparator offset requirement cannot be achieved with the intrinsic matching for the target speed, and thus an offset calibration is required.

Transistor-level Monte-Carlo simulations show that the input referred offset has several contributors. Besides the comparator latches as the dominant contributor, the offset contributed by the PMOS transistors of the Gm cell (M₁, M₂ in Figure 8.6), and the switches between the summation nodes and the latches (M₇, M₈, M₁₃ - M₁₆, M₂₁ and M₂₂ in Figure 8.6) needs to be calibrated as well. The calibration range is chosen to be more than $\pm 6\sigma$ of the total input-referred comparator offset for robustness. After offset calibration, the input-referred offset residue should be less than 0.5 LSB, to maintain the ADC resolution.

8.2.2 System architecture

To achieve the required calibration range and accuracy, an on-chip two-step coarsefine calibration is proposed, as shown in Figure 8.9. The coarse calibration is done by injecting calibration current (I_{cal1} , I_{cal2}) directly at the latches' input, to cover the required calibration range. The fine calibration is achieved by adding a calibration current-steering DAC at the ELD summation node, to achieve the required calibration accuracy. Fine calibration range is more than 2 coarse calibration steps for robustness.

The SDM has two modes — a normal operation mode and a calibration mode. The mode is defined by the control signal in Figure 8.9 (this control signal is called "ena_cali"). In the calibration mode (ena_cali = 1), the quantizer input is disconnected from the loop filter output, and the quantizer input is connected to a fixed voltage, which is the common-mode voltage of the quantizer input during normal operation mode, with a differential voltage of 0. The ELD DAC input is disconnected from the quantizer output, and connected to a fixed code 0 (middle code). The differential reference current is also set to 0 in the calibration mode. Thus, the differential input of the latch is 0 when the offset is 0. The calibration algorithm is proposed per



Figure 8.9: Proposed SDM system architecture with current-mode multi-path ELD compensation, locally-TI multi-bit quantizer, and comparator offset calibration circuits.

1-bit quantizer slice, shown as the calibration logic in Figure 8.9.

The principle of calibration algorithm is as follows. In the beginning, latch 1 is under calibration, and only its output is connected to the calibration logic. The coarse calibration code for latch 1 (referred to as coarse code 1) is swept from minus full scale to plus full scale by the minimum step of the coarse code. For every coarse code, the latch 1 output is recorded by the calibration logic. By design it is guaranteed that for the coarse code minus full scale, latch 1 output is always -1 for all possible cases of device mismatch (for a coverage of more than $\pm 6\sigma$). During the sweeping of the coarse code 1, when latch 1 output becomes +1, the correct calibration code is detected, and the sweeping is stopped. The calibration logic set the coarse code 1 by one minimum step back, to compensate the effect of the hysteresis. Now one round of the coarse calibration of latch 1 is finished. Sweeping is used in the calibration algorithm instead of the binary searching, since it is less sensitive to the hysteresis effect of the comparator. A unary weighted coarse calibration circuit can guarantee its transfer function to be inherently monotonic. The flowchart of one round of coarse calibration for latch 1 is shown in Figure 8.10.

Then, the coarse calibration of latch 2 starts. It is similar as the coarse calibration of latch 1. Schematic and post-layout simulations show that the calibration of latch 1 and latch 2 are not completely independent. The dependency can be caused by the following two reasons. (1) The calibration DAC and the summation node are shared by the fine calibration of latch 1 and latch 2, and the memory effect can cause



Figure 8.10: Flowchart of one round of coarse calibration (latch 1 as an example).

dependency. (2) Some dependency can be generated from the TI comparators (M₇ - M₂₂ in Figure 8.6). During the transient between the reset and input settling phase to the regeneration phase, the comparison result of one latch can influence the initial condition of the other latch. Because the calibration of latch 1 and latch 2 are not completely independent, several iterations on the coarse calibration of latch 1 and latch 2 is finished when their detected coarse codes are not changed. From schematic and post-layout simulations, the coarse calibration codes always converge after a few iterations (e.g. latch 1 \rightarrow latch 2 \rightarrow latch 1 \rightarrow latch 2 \rightarrow latch 1).

After the coarse calibration iterations are finished, the fine calibration starts. Firstly latch 1 is under fine calibration. It also uses a similar principle by sweeping the fine code 1 from minus full scale to plus full scale with the minimum step, and detecting the changing of latch 1 output. After the changing of latch 1 output is detected, the sweeping stops, and the fine code 1 is set back by 1 minimum step. After the fine calibration of latch 1 is finished, the fine calibration of latch 2 starts. Several iterations on the fine calibration of latch 1 and latch 2 are also necessary. The fine calibration of both latch 1 and latch 2 is finished when their detected fine codes are not changed (or changed within one minimum step). From schematic and post-layout simulations, the fine calibration codes always converge within one minimum step after

a few iterations (e.g. latch $1 \rightarrow \text{latch } 2 \rightarrow \text{latch } 1 \rightarrow \text{latch } 2 \rightarrow \text{latch } 1$). Notice that one minimum fine calibration step is designed to be about $\frac{1}{6}$ LSB. Thus, even if the fine calibration codes toggles within one minimum step in the last two iterations of the fine calibration, the residue comparator offset still meets the requirements.

The coarse calibration circuit is separated for latch 1 and latch 2, while the fine calibration circuit is combined for those two time-interleaved latches. It is implemented as a current steering DAC (referred to as calibration DAC). In the calibration mode, both two latches work in the time-interleaved way, even though only one latch output is sampled and passed to the calibration logic. The reason is to keep the environment of the quantizer in the calibration mode as close as possible to the normal operation mode. So, in both calibration mode and normal operation mode, the calibration DAC input toggles between two fine codes, implemented with a multiplexer (MUX). A delayed version of the latch clock signal ($clkp_{ltc}/clkn_{ltc}$ in Figure 8.6) is used as the select signal for the multiplexer.

The requirement of the full swing and steps of the coarse and the fine calibration is as follows. In the worst case, the full swing of the coarse calibration should cover all the possible cases of the comparator offset (for a coverage of more than $\pm 6\sigma$). The full swing of the fine calibration should be larger than the largest coarse calibration step (the coarse calibration circuit can have different step size due to its own mismatch). And the largest fine calibration step should be less than the calibration accuracy required by the system (the calibration DAC may have different step size due to its own mismatch).

The flowchart on the operation principle of the current-mode locally-TI multi-bit quantizer and offset calibration, including both the calibration mode and the normal operation mode, is presented in Figure 8.11.

8.2.3 Circuit implementation

The transistor-level circuit implementation of an SDM with the proposed comparator offset calibration is shown in Figure 8.12. It shows one slice of the multi-bit quantizer in detail. In the following, only the additional circuit and functions compared to Figure 8.6 are explained in detail. They are used for the offset calibration.

The coarse calibration circuit is implemented as an array of small PMOS transistors, in parallel with the PMOS transistor of the latch. Their gates are connected to a series of control signals (ctrl1, ctrl2, ..., ctrln). E.g., when the control signal ctrl1 is the same as the clock signal of the latch $clkp_{ltc}$, the small PMOS transistor M_{33} is offering a certain amount of current at the same time of M_{17} . Thus, it implements a coarse step for the offset calibration. When ctrl1 is VDD (1.1 V), the PMOS transistor M_{33} is always OFF. The elements of coarse calibration circuit can be either unary weighted or binary weighted, or a combination of both (segmented into a unary part and a binary part). A unary weighted coarse calibration circuit is intrinsically monotonic.


Figure 8.11: Flowchart of locally time-interleaved comparators and offset calibration.

The fine calibration circuit is implemented as a current steering DAC (referred to as calibration DAC), similar as the ELD DAC. The unit DAC elements can be either unary weighted or binary weighted, or a combination of both (segmented into a unary part and a binary part).

The calibration mode and the normal operation mode are defined by the master control of the SDM. The control signals from the master control are shown as green lines. The master control has three functions. First, it controls the switches between the loop filter and the quantizer, and between the sampled quantizer output and the main DACs. Second, it controls the input digital code of the reference DACs. E.g., for a 3-bit quantizer, the input codes of the reference DACs are all 0 (middle code) during the calibration mode. While those codes are -3, -2, -1, 0, 1, 2, 3, respectively during the normal operation mode. Third, it controls the exact calibration procedure. With two control signals (named "coarse_not_fine_cali" and "latch1_not_latch2_cali"), it defines 4 calibration modes – coarse calibration latch 1, coarse calibration latch 2, fine calibration latch 1 and fine calibration latch 2.



In the calibration mode, the local calibration logic block in every 1-bit slice generates and stores some control signals. It also generates and stores coarse code 1/2 and fine code 1/2 during both calibration mode and normal operation mode. The calibration logic block is a digital circuit, which is integrated on chip. It controls three analog blocks – coarse calibration control, fine calibration control, and clock control block.

The latch output samplers for the fast feedback loop (ELD compensation loop) and the main feedback loop are separated. The fast feedback loop sampler is shown in more detail in Figure 8.12. When the SDM is in the calibration mode, the control signal ena_cali is high; otherwise, when the SDM is in the normal operation mode, ena_cali is low. In the calibration mode, the sampling clocks $clk1_{SH,FL}$ and $clk2_{SH,FL}$ are disabled by the clock control block. The switches controlled by the ena_cali are ON, and the fast feedback loop sampler output is set to be 0 (or 1) depending on the number of the 1-bit quantizer slice. For half of the total amount of the 1-bit quantizer slices, the fast feedback loop sampler outputs are set to 1, while for the other half they are set to 0. These sampler outputs are transferred to the ELD DACs. Thus, a middle code is generated as the ELD DAC input during the calibration mode. Notice that if the number of the unit ELD DACs per slice is not an even number, one extra unit ELD DAC is required to generate a zero differential output current from the ELD DAC.

The main feedback loop sampler is implemented similarly as the fast feedback loop sampler. The difference is that the switches controlled by ena_cali are not needed in the main feedback loop sampler. It requires two sampling $clocks - clk 1_{SH,ML}$ and $clk2_{SH,ML}$. In the normal operation mode, $clk1_{SH,ML}/clk2_{SH,ML}$ and $clk1_{SH,FL}/clk2_{SH,FL}$ are the same, which are shown as clk_{SH}/clk_{SH} in Figure 8.6 (only the positive clock signals are shown in Figure 8.7 as $clkp1_{SH}/clkp2_{SH}$). In the calibration mode, $clk1_{SH,ML}/clk2_{SH,ML}$ and $clk1_{SH,FL}/clk2_{SH,FL}$ are different, since the main feedback loop sampler's output is passed to the calibration logic blocks as an input. When latch 1 is under calibration (either coarse calibration or fine calibration), only latch 1 output is sampled and passed to the calibration logic, while latch 2 output is not sampled. In this case, a control signal named ena_cali_latch1 is high. And only $clk1_{SH,ML}$ is enabled (the same as in the normal operation mode), while the $clk_{SH,ML}$ is disabled. Similarly, when latch 2 is under calibration (either coarse calibration or fine calibration), only latch 2 output is sampled and passed to the calibration logic, while latch 1 output is not sampled. In this case, a control signal named ena_cali_latch2 is high. And only $clk_{2SH,ML}$ is enabled (the same as in the normal operation mode), while the $clk1_{SH,ML}$ is disabled.

The coarse calibration control circuit is shown in Figure 8.13. The coarse calibration control circuit for latch 2 is taken as an example, while the circuit for latch 1 can be designed similarly, by just switching clkp and clkn. The output ctrl is one of the control signals ctrl1, ctrl2, ..., ctrln in Figure 8.12. The function of this block is as follows. When the corresponding bit of coarse code 1/2 (code_P in Figure 8.13) is 1 (logic high, 1.1 V), its output (ctrl) should be the same as clkp_{ltc}. In this case, the transistors M₁ and M₄ are ON, such that the inverter composed of M₂ and M₃



Figure 8.13: Coarse calibration control circuit for latch 2.

is working. The coarse calibration control circuit should be designed to have the same delay as the clock buffer. When $code_P$ is 0, the output of the coarse calibration control circuit should be VDD. In this case, the transistors M_1 and M_4 are OFF, and the inverter composed of M_2 and M_3 is not working. The transistor M_5 is ON, and it drives the ctrl node to VDD.

Actually, the coarse calibration control circuit can be considered as an inverter with an enable signal. Its symbol is defined in Figure 8.13. It has two inputs (in, ena) and one output (out). In Figure 8.13, in $= \text{clkn}_{ltc,in}$, ena $= \text{code}_P$, and out = ctrl. This inverter with an enable signal is also used in the clock control block.

The fine calibration control circuit is shown in Figure 8.14. This circuit works as a multiplexer. Delayed version of the latch clocks $(\text{clkp}_{ltc,d} \text{ and } \text{clkn}_{ltc,d})$ are used as the select signal for the multiplexer. The multiplexer function is achieved by using two pass gates $(M_1 - M_4)$. The multiplexer output is further delayed and buffered with several inverters. In this design, the total delay from clkp_{ltc} / clkn_{ltc} to Fp/Fn is about 25% of the period of clkp_{ltc} / clkn_{ltc} (this delay is equivalent to half of the SDM sampling period T_S). The outputs Fp/Fn are one pair of the calibration DAC differential input signals Fp[1]/Fn[1], Fp[2]/Fn[2], ..., Fp[k]/Fn[k] in Figure 8.12.

The clock control circuit is shown in Figure 8.15. This circuit enables or disables the samplers' clock, $clk1_{SH,ML}/clk2_{SH,ML}$ and $clk1_{SH,FL}/clk2_{SH,FL}$, depending on the calibration procedure. The inverter with an enable signal shown in Figure 8.13 is reused here. Notice that $clkn1_{SH,ML}$, $clkn2_{SH,ML}$, $clkn1_{SH,FL}$ and $clkn2_{SH,FL}$ are the gate voltages of the PMOS transistors in the pass gates of the samplers, and $clkp1_{SH,ML}$, $clkp2_{SH,ML}$, $clkp1_{SH,FL}$ are the gate voltages of the NMOS transistors.



Figure 8.14: Fine calibration control circuit.

in the pass gates of the samplers, as shown in Figure 8.6. When the corresponding sampling clock is disabled, $clkn1_{SH,ML}/clkn2_{SH,ML}/clkn1_{SH,FL}/clkn2_{SH,FL}$ should be set to VDD (1.1 V), while $clkp1_{SH,ML}/clkp2_{SH,ML}/clkp1_{SH,FL}/clkp2_{SH,FL}$ should be set to GND (0 V).

The operation principle of the clock control circuit is as follows. When the SDM is in the normal operation mode, the control signals ena_cali, ena_cali_latch1 and ena_cali_latch2 are all logic low (0 V). Thus, all of the inverters with enable signals are working as normal inverters, and all of the sampling clock signals are enabled. When the SDM is in the calibration mode, ena_cali is logic high (1.1 V), and the fast feedback loop sampler's clock clk1_{SH,FL}/clk2_{SH,FL} are disabled – clkn1_{SH,FL}/clkn2_{SH,FL} are set to logic high (1.1 V), and clkp1_{SH,FL}/clkp2_{SH,FL} are set to logic low (0 V). In the calibration mode, when latch 1 is under calibration (either coarse or fine calibration), ena_cali_latch1 is logic high, and the main feedback loop sampler's clock for latch 2 clk2_{SH,ML} is disabled – clkn2_{SH,ML} is set to logic high (1.1 V), and clkp2_{SH,ML} are set to logic low (0 V). Thus, only latch 1 output is sampled by the main feedback loop sampler, and passed to the calibration logic block. Similarly, when latch 2 is under calibration (either coarse or fine calibration), ena_cali_latch2 is logic high, and the main feedback loop sampler, and passed to the calibration logic block. Similarly, when latch 2 is under calibration (either coarse or fine calibration), ena_cali_latch2 is logic high, and the main feedback loop sampler, and passed to the calibration logic block. Similarly, when latch 2 is under calibration (either coarse or fine calibration), ena_cali_latch2 is logic high, and the main feedback loop sampler's clock for latch 1 clk1_{SH,ML} is disabled – clkn1_{SH,ML} is set to logic low (0 V).

Figure 8.16 shows the clock and control signals during the calibration mode. A part of the coarse calibration for latch 1 is shown as an example. In this case, ena_cali_latch1



Figure 8.15: Clock control circuit.

is 1. Since only the DC offset needs to be calibrated, the calibration clock clk_{Cali} is designed to be at much lower frequency than the latch clock $clkp_{ltc}$. In the test chip, the calibration clock has a frequency $1024 \times$ lower than the sampling rate of the SDM. Although the calibration clock clk_{Cali} is at a much lower frequency, the latch clock $clkp_{ltc}/clkn_{ltc}$ is at the same frequency as the normal operation mode. Thus, any dynamic effects of the comparator offset which depends on the SDM's sampling rate are as far as possible preserved in the calibration mode. The calibration codes change at the negative clock edge of the calibration clock clk_{Cali} , while the sampled latch output is re-sampled by the calibration logic at the positive clock edge of clk_{Cali} . Thus, when one new calibration code is applied, the latch under calibration has many clock periods to stabilize the output, before the latch output is sampled by the calibration logic. In the calibration mode, the latch clock is the same as in the normal operation mode, while the sampler's clocks for the fast feedback loop $clkp1_{SH,FL}$ and $clkp2_{SH,FL}$ are disabled. When latch 1 is under calibration, the sampler's clock for the main feedback loop of latch 2 $clkp2_{SH,ML}$ is disabled, and vice versa.



8.2.4 Simulation results

Transistor-level Monte-Carlo simulations with device mismatch are performed to verify the effectiveness of the proposed comparator offset calibration technique. Monte-Carlo simulations show that after calibration the maximal residue offset is about 0.12 LSB of the 3-bit quantizer. After calibration, the original SNDR of the ADC without device mismatch is restored.

The proposed comparator offset calibration has been implemented on the test chip. The area of the 'calibration logic' in Figure 8.12 is 2030 μ m². The comparator offset calibration is run at the start-up, while the voltage and the temperature during the calibration mode should be as close as possible to the normal operation mode. Measurement results on the offset calibration are presented in Chapter 11, validating its effectiveness. In many applications, e.g. automotive radar, the operation of the receiver is duty-cycled – it has an ON time and an OFF time. The OFF time of the receiver can be used to run a foreground calibration, to track the temperature change.

8.2.5 Conclusions

This section proposes an on-chip comparator offset calibration technique for broadband CT $\Sigma\Delta$ ADCs with current-mode multi-path ELD compensation and locally-TI multi-bit quantizer. A two-step coarse-fine calibration is designed to achieve the required calibration range and accuracy. The calibrations for the locally-TI comparator latches are not independent, and hence iterations of coarse calibrations and fine calibrations of the TI comparators are required. Simulation results show that after the proposed calibration the input-referred comparator offset is reduced from 9 LSB (3σ) to 0.12 LSB. With the proposed calibration, the comparator offset requirement for the multi-bit quantizer is achieved, which achieves higher speed and better power efficiency than the approach of intrinsic matching. The proposed offset calibration guarantees the required linearity of the multi-bit quantizer with the current-mode multi-path ELD compensation and the locally-TI multi-bit quantizer techniques, and enables broadband CT $\Sigma\Delta$ ADC applications.

8.3 Conclusions

This chapter proposes two techniques for high-speed comparators in broadband CT $\Sigma\Delta$ ADC. A current-mode locally-time-interleaved multi-bit quantizer moves the reset time and the time to track the next initial condition of the comparator out of the 1 T_S ELD time budget, enabling $1.33 \times$ higher ADC sampling rate and signal BW. Compared to the voltage-mode globally-TI quantizer, the proposed current-mode locally-TI quantizer reduces the parasitic capacitance seen by the TI switches, reducing the settling time of comparator input and relaxing the R_{ON} requirement of the TI

switches. An on-chip two-step coarse-fine comparator offset calibration is proposed for broadband CT $\Sigma\Delta$ ADCs with the current-mode multi-path ELD compensation and the locally-TI multi-bit quantizers. With the proposed offset calibration, the input-referred comparator offset is reduced from 9 LSB (3σ) to 0.12 LSB. Both approaches together enable broadband CT $\Sigma\Delta$ ADCs with multi-bit quantization.

Chapter 9

DAC linearization techniques

Mismatch between the unit elements in the multi-bit DAC is often limiting the linearity of the $\Sigma\Delta$ ADCs with multi-bit quantization. Many linearization techniques for high-linearity DACs are proposed in the literature, such as sorting, data weighted averaging and randomization. However in the conventional way of implementing the DAC linearization techniques, it often increases the delay in the feedback loop, which limits the sampling rate and bandwidth of the ADC. This chapter proposes a novel implementation of the DAC linearization techniques by changing the reference codes of the multi-bit quantizer. It adds minimal to no delay in the feedback loop of the $\Sigma\Delta$ modulator, depending on the chosen DAC linearization technique, which is beneficial for high-speed $\Sigma\Delta$ ADC applications. The mapping engine is implemented in the digital domain, whose non-idealities do not interfere with the analog purity of the reference signal. It is a general purpose architecture which offers the flexibility to implement different static and dynamic DAC linearization techniques.

9.1 Introduction

Chapter 4.3 describes that mismatch between the unit DAC cells in multi-bit DACs generates non-linearity. The mismatch between the unit DAC cells creates randomly distributed errors. DAC mismatch can limit the DR and linearity performance of broadband CT $\Sigma\Delta$ ADCs with multi-bit quantization. Generally, the problem caused by the DAC mismatch can be mitigated in two ways – (1) by intrinsic matching (2) by calibration.

Many broadband multi-bit $\Sigma\Delta$ ADCs achieve the DAC linearity requirement by intrinsic matching [30]. By sizing up the matching-critical devices and careful layout (e.g. applying the common-centroid technique), a certain matching requirement can be achieved [71].

Various DAC calibration techniques can be found in the literature [90]. The static

error on the current sources of each unit DAC cell can be calibrated [91, 16]. In [66], the DAC mismatch error for every input bit is estimated and an auxiliary DAC is applied to compensate the DAC error in the analog domain. After the DAC mismatch error is estimated, it can also be compensated in the digital domain [65, 67]. When the mismatch error of every unit DAC cell is estimated, a certain sorting/mapping on the order of the unit DAC cells can be calculated by an algorithm, and the multi-bit DAC after sorting/mapping can achieve better (overall) linearity performance than with the default order [92, 64]. With sorting, the errors on the DAC unit elements mutually cancel each other, and the integral non-linearity (INL) of the DAC is improved. The overall linearity of the multi-bit SDM can be improved by switching/shuffling schemes as well, for example dynamic element matching (DEM) [93], data weighted averaging (DWA) [94, 95, 88, 96], element rotation [97]. With DWA and DEM, the power of the harmonic distortion is transformed to noise by randomizing the error. So THD improves but the noise deteriorates. Noise shaping and a relatively high OSR (typically ≥ 16) are required to reduce the noise penalty.

Besides various DAC linearization algorithms, different implementation of the DAC linearization algorithms can be found in the literature as well. This chapter assumes a 'mapping engine' is used to implement any of the above mentioned DAC sort-ing/mapping/switching techniques [64]. In $\Sigma\Delta$ ADCs, the mapping engine can be implemented as a switching matrix controlled by the logic [93, 96, 97, 88]. Traditionally, the mapping engine is implemented between the quantizers and the DACs in $\Sigma\Delta$ ADCs [93, 96]. In the state-of-the-art broadband CT $\Sigma\Delta$ ADCs, the mapping engine is implemented between the multi-bit quantizer's reference and the quantizer, to eliminate the delay of the mapping engine in the SDM feedback loop [97, 88]. These two alternative implementations are discussed in detail in the next section.

9.2 State-of-the-art implementations of DAC linearization techniques

Figure 9.1 shows the conventional implementation of DAC linearization techniques in multi-bit SDMs, with the mapping engine between the quantizer and the DAC [93, 96]. Here, the DAC linearization techniques can be sorting/mapping, DWA, randomized DWA, DEM, randomization, etc. In Figure 9.1, the mismatch sensor measures the static and/or dynamic mismatch between the unit elements of the main DAC [64]. A certain sorting, mapping or switching algorithm is implemented in the logic block. The required switches are implemented in the mapping engine. Here, τ_1 is the delay of the multi-bit quantizer, and τ_2 is the delay of the mapping engine.

The disadvantages of the conventional implementation shown in Figure 9.1 is that the mapping engine which implements the switches required by the DAC linearization techniques is in the feedback loop of the SDM. It adds capacitive and resistive load, and adds delay (τ_2) in the feedback loop. The extra delay caused by the mapping



Figure 9.1: Conventional implementation of DAC linearization techniques in SDMs with the mapping engine between the quantizer and the DAC.

engine can negatively impact the performance of the SDM, or even make it unstable, especially in very high-speed SDMs with several GHz sampling rate [97]. In Figure 9.1 the red part of the figure shows the feedback loops of the SDM, which are speed critical and sensitive to extra delay.

In very high-speed multi-bit SDMs with several GHz sampling rate, the multi-bit quantizer is commonly implemented as a flash sub-ADC, since the latency of the quantizer contributes to the total ELD and flash ADCs have a low latency. This flash ADC compares the input signal with a series of references (reference voltages or reference currents). The unit 1-bit comparators in the flash ADC (multi-bit quantizer) and the unit elements in the main DAC are typically one-to-one connected through some connection circuitry, e.g. samplers, buffers, D-flipflops. In this case, instead of putting the mapping engine between the quantizer and the main DAC, the mapping engine can be inserted between the reference and the multi-bit quantizer (flash sub-ADC). By sorting/mapping/switching the reference among the unit 1-bit comparators, an effective sorting/mapping/switching of the unit elements of the DAC is achieved. The simplified block diagram of this implementation is shown in Figure 9.2 [97, 88]. The resistor ladder in [88] is considered as an implementation of the reference DAC, which delivers the required reference voltages. The mapping engine might need the quantizer output V or not, depending on the particular DAC linearization technique which needs to be implemented. In this implementation, the delay of the mapping engine is moved out from the SDM's feedback loops.

In the state-of-the-art implementation in Figure 9.2, the mapping engine is implemented in the analog domain, which has two disadvantages. First, the mapping engine typically contains some switches, and the non-idealities of the mapping engine (switches) interfere with the analog purity of the reference signal. For example, the



Figure 9.2: State-of-the-art implementation of DAC linearization techniques in SDMs with the mapping engine between the quantizer's reference DAC and the quantizer.

mapping engine adds load to the output of the reference DAC, which increases the time constant during the switching. Another example is that the mismatch of the switches contributes to the offset of the reference signal. Second, since the mapping engine is implemented in the analog domain, it is not flexible to switch to different DAC linearization techniques. In the next section, a new implementation of the DAC linearization techniques is proposed, where the mapping engine is moved to the digital domain to overcome the aforementioned limitations.

9.3 Proposed implementation of DAC linearization techniques

Figure 9.3 shows the proposed implementation of DAC linearization techniques, where the mapping engine is moved to the digital domain. For a (n + 1)-level quantizer, the reference DAC is copied n times. The mapping engine is inserted between the reference DACs and their input codes (reference codes). By sorting/mapping/switching of the reference codes, a sorting/mapping/switching on the outputs of the reference DACs are achieved. When a voltage-mode multi-bit quantizer is considered, even if the reference DAC with voltage output (e.g. a resistor ladder) is copied n times, switches in the analog domain connecting different reference voltage levels to the multi-bit quantizer are still required to implement the mapping engine, which shows no extra benefits compared to Figure 9.2. However, when a current-mode multi-bit quantizer is considered, the duplication of the reference DACs with current output allows eliminating the switches at the DAC output in the analog domain and moving



Figure 9.3: Proposed implementation of DAC linearization techniques in SDMs with the mapping engine between the reference code and the reference DAC.

the mapping engine completely into the digital domain. This is elaborated in the following with Figure 9.4 and Figure 9.5.

In the current-mode multi-path ELD compensation proposed in Chapter 7.2, a currentmode quantizer with duplicated reference DACs already exists. The system architecture shown in Figure 7.9 can be used to implement DAC linearization techniques in an efficient way, suitable for high-speed operation. Figure 9.4 shows one of the proposed SDM architectures with a current-mode quantizer and DAC linearization techniques that do not need the quantizer output. Here, a (n + 1)-level quantizer is shown as an example. It contains *n* slices of a 1-bit comparator (latch), a (n+1)-level ELD DAC, a *n*-level reference DAC, a gm-cell and a current summation node. The reference DAC (Ref DAC) converts the reference code (Ref code) into a reference current. The (n + 1)-level quantizer has *n* reference codes, which are {code 1, code 2, ..., code n}. The mapping engine connects the inputs {code 1, code 2, ..., code n} to the outputs {ref code 1, ref code 2, ..., ref code n} with a chosen order. Thus, a certain sorting/mapping/switching can be implemented.

Two main classes of the DAC linearization techniques can be distinguished that do not need the information of the quantizer output. First, it can be a static sorting/mapping technique [64]. In this case, with the measured information on the static and dynamic mismatch of the DAC unit elements, a certain optimal order of the DAC unit elements is calculated by the logic. This order of the DAC unit elements is fixed during the operation of the SDM. The mapping engine just needs to implement a fixed connection with a certain optimal order between the input codes and the output codes. In this case, as the mapping engine is not switched during the operation of the SDM, its delay has no impact on the operation of the SDM.



Figure 9.4: Proposed SDM architecture with a current-mode quantizer and DAC linearization techniques that do not need the quantizer output.

Second, the DAC linearization technique can also be a dynamic switching technique, for example a randomization technique. In this example, the order of the DAC unit elements is randomly switched during the operation of the SDM. After each switching, the reference DAC output current and the comparator input should be settled before the next sampling moment of the comparator. Thus, the total delay of the mapping engine and the reference DAC should be within one sampling clock period of the SDM.

The previous examples show architectures for SDMs that use linearization techniques that do not need the information of the DAC input. Some linearization techniques, however, do need this information, like DWA and randomized DWA. For those techniques this chapter proposes the architecture shown in Figure 9.5. The quantizer output V, which is the same as the DAC input, is now also an input of the mapping engine.

The mapping engine is part of a feedback loop now, and its switching speed is critical, like in the conventional implementation shown in Figure 9.1. Therefore it is shown in red in Figure 9.5 as well. However, the proposed implementation in Figure 9.5 still shows advantages compared to the conventional implementation, which are elaborated in the following. The main feedback loop of the SDM and the DAC linearization



Figure 9.5: Proposed SDM architecture with a current-mode quantizer and DAC linearization techniques which need the quantizer output.

loop are (partially) decoupled. The DAC linearization loop contains the comparator (latch), mapping engine, reference DAC and the ELD summation node. The SDM main feedback loop contains the comparator (latch), main DAC, loop filter, Gm cell and ELD summation node. Thus, the SDM main feedback loop and the DAC linearization loop can be optimized separately for their different requirements. For example, the main DAC typically has a stringent noise and matching requirement. Thus, the unit element of the main DAC typically has a relatively large current (for lower thermal noise) and a relatively large area (for better matching and lower flicker noise). The reference DAC in the multi-bit quantizer typically has a relaxed noise and matching requirement compared to the main DAC, as the noise and non-linearity contributed by the reference DAC is shaped by the NTF. Thus, it can be beneficial to insert the mapping engine between the comparator and the reference DAC, compared to the conventional approach where the mapping engine is inserted between the comparator and the main DAC.

In Figure 9.5 the red part of the figure shows the main feedback loop, the ELD feedback loop and the DAC linearization loop of the SDM. The total delay of the DAC linearization loop should be maximally one sampling clock period. The total delay of the main feedback loop and the ELD feedback loop should be both maximally

one sampling clock period as well. The ELD DAC and the reference DAC can be implemented similarly (see Figure 9.6), which means they have a similar delay. It is essential to minimize the delay of the mapping engine in this scenario for high-speed $\Sigma\Delta$ ADC applications.

The proposed implementation in Figure 9.3, 9.4 and 9.5 have two advantages compared to the state-of-the-art implementation in Figure 9.2. First, the mapping engine is moved to the digital domain, and the analog purity of the reference signal is not interfered by the non-idealities of the mapping engine. Second, since the logic and the mapping engine are in the digital domain, it is a general purpose implementation and it offers the flexibility to switch to different DAC linearization techniques easily. A concern of the proposed implementation is that when a switching technique (e.g. DWA, DEM, randomization, rotation) is used, the reference DAC is in the switching path, and thus the delay of the reference DAC adds to the total delay of the path. The delay of the reference DAC is not a concern when a static sorting/mapping technique is used. A drawback of the proposed implementation is that the number of the reference DACs is increased.

Notice that the proposed approach is not capable to implement all the existing DAC linearization techniques. The proposed approach (Figure 9.5) cannot select the unit elements of the main DAC based on the current sample of the quantizer output. It can only select the order of the unit elements of the main DAC to be used based on the previous sample of the quantizer output.

Notice also that some block(s) shown in Figure 9.3, Figure 9.4 and Figure 9.5 maybe not necessary and hence omitted in certain applications. For example, some switching algorithms of the DAC unit elements (DWA, DEM, random switching, etc.) do not need the information on the mismatch of the DAC unit elements. Thus, the mismatch sensor can be omitted in those cases.

Finally, by sorting/mapping/switching of the reference code of the quantizer, the quantizer can be linearized as well. For a linearization technique which requires the information on the mismatch, e.g. a static sorting [64], a mismatch sensor which measures the mismatch information of the quantizer is needed. For a linearization technique which does not need the information on the mismatch, e.g. DWA or DEM, a mismatch sensor is not needed. By applying DWA or DEM, the quantizer mismatch is randomized and converted into noise.

This approach works on all the errors originating from the quantizer and the DAC in the $\Sigma\Delta$ loop. It can be achieved in two ways. (1) Mismatch sensors can be applied to measure the mismatch information on both the DAC and the quantizer. This information is used by a static or dynamic linearization technique to achieve the optimal performance of the $\Sigma\Delta$ loop. (2) The logic can determine an optimal sorting/mapping/switching with a static or dynamic linearization technique based on the overall performance of the $\Sigma\Delta$ loop, e.g. in-band noise, SNDR, THD. Thus, the errors originating from both the quantizer and the DAC are taken into account for



Figure 9.6: Exemplary schematic implementation of an SDM with a DAC linearization technique that does not need the quantizer output.

an optimal overall performance of the $\Sigma\Delta$ loop.

9.4 Circuit implementation

Figure 9.6 shows an exemplary schematic implementation of an SDM with a DAC linearization technique that does not need the quantizer output. It is a schematic implementation of the system in Figure 9.4. It shows one slice of the duplicated ELD compensation loop and 1-bit comparator in detail. This figure assumes the SDM uses (n + 1)-level quantization. So the SDM has totally *n* slices of duplicated ELD compensation loop and 1-bit comparator. The circuit implementation of the Gm cell, ELD DAC, latch block, samplers and buffers have been described in Chapter 7.2 (similar as in Figure 7.10).

The NMOS transistors $M_5 - M_6$ and current sources $I_{ref,u}$ compose a unit reference DAC (Ref DAC). The reference DAC converts a digital input code (reference code) into an output current. The (n+1)-level quantizer shown in Figure 9.6 has n reference codes – ref code 1, ref code 2, , ref code n. These n reference codes together form a permutation of the n input codes {code 1, code 2, ..., code n}. The mapping engine

applies a certain sorting/mapping/switching on the connection from the input codes $\{\text{code 1, code 2, ..., code n}\}$ to its output codes $\{\text{ref code 1, ref code 2, ..., ref code n}\}$. The mapping engine is implemented in digital domain as n registers. The reference codes are saved in those registers, which can be written according to the control signal from the logic.

Since the 1-bit latch blocks are one-to-one connected to the unit elements of the main DAC through the S&H circuit and buffers, by sorting/mapping/switching of the reference code of the reference DAC, a corresponding sorting/mapping/switching of the DAC unit elements is effectively achieved. For dynamic linearization techniques (e.g. DWA, DEM, randomization, rotation), the delay of the reference DAC contributes to the total delay of the DAC linearization loop. The ELD DAC typically has about $0.5T_S$ to settle (the other $0.5T_S$).

From the circuit implementation point of view, the proposed approach shows two additional advantages compared to the state-of-the-art approach (Figure 9.2). (1) In [88], the switches which implement the mapping engine between the resistor ladder (a voltage-mode reference DAC) and the multi-bit quantizer have an ON resistance R_{ON} . When the mapping engine is switching, R_{ON} of the switches is in series with the impedance of the resistor ladder. Together with the input parasitic capacitance of the quantizer a switching time constant is defined. To achieve high-speed switching, R_{ON} of the switches and the impedance of the resistor ladder should both be small, which increases the power consumption. In the proposed approach (Figure 9.6), since a current-mode reference DAC is used, the R_{ON} of the switches (M_5, M_6) is in series with the current source $I_{ref,u}$, which has a large output impedance. Thus, the R_{ON} of the switches (M_5, M_6) does not determine the switching time constant, and hence the requirement on the R_{ON} of the switches is relaxed compared to the state-of-theart approach [88]. (2) In [88], the switches (which implement the mapping engine) are working at different reference voltage levels. Thus they have different R_{ON} (and parasitic capacitance), which creates errors on the output reference voltages of the mapping engine. In the proposed approach, since a unary current-mode reference DAC is used, all the switches (M_5, M_6) are working at the same voltage level. Thus, their R_{ON} (and parasitic capacitance) are the same.

9.5 Simulation results

This section shows the simulation results of a static sorting technique as an example to demonstrate the feasibility of the proposed approach in broadband $\Sigma\Delta$ ADCs. A sorting algorithm similar as in [64] has been modelled in Matlab. In this model, a 2-1-1 MASH SDM with 3-bit quantization in every loop is considered to verify the effectiveness of the proposed technique. Normally distributed (also known as Gaussian distributed) amplitude errors have been modelled on the 7 unit elements of the 3-bit feedback DAC. The sorting algorithm is applied to the first DAC in the first stage, and uses the error information { $\varepsilon_1, \varepsilon_2, ..., \varepsilon_7$ } on that DAC. The sorting algorithm is as follows:

- 1. The common-mode error $\frac{\sum_{n=1}^{7} \varepsilon_n}{7}$ is firstly subtracted from the error information $\{\varepsilon_1, \varepsilon_2, ..., \varepsilon_7\}$ to get $\{\varepsilon'_1, \varepsilon'_2, ..., \varepsilon'_7\}$.
- 2. The maximal absolute value in $\{\varepsilon'_1, \varepsilon'_2, ..., \varepsilon'_7\}$ is chosen as the 1st element after sorting. If for example ε'_3 is chosen, then $\{\varepsilon'_3\}$ is called the sorted error, and $\{\varepsilon'_1, \varepsilon'_2, \varepsilon'_4, \varepsilon'_5, \varepsilon'_6, \varepsilon'_7\}$ is called the unsorted errors.
- 3. From the unsorted errors, every error is added to the summation of the current sorted errors. The one which gets the minimal absolute summation result is picked as the next sorted error. This sorting scheme continues until all the errors are sorted. For example, after sorting, the sorted errors are $\{\varepsilon'_3, \varepsilon'_7, \varepsilon'_1, \varepsilon'_6, \varepsilon'_4, \varepsilon'_2, \varepsilon'_5\}$.
- 4. The common-mode error $\frac{\sum_{n=1}^{7} \varepsilon_n}{7}$ is added back to the sorted errors. From the previous example, the finally sorted error permutation is $\{\varepsilon_3, \varepsilon_7, \varepsilon_1, \varepsilon_6, \varepsilon_4, \varepsilon_2, \varepsilon_5\}$.

With this sorting algorithm, the original random errors are sorted to form a certain permutation, such that the errors of the unit elements are mutually partially compensated. In the majority of cases, the INL due to the differential errors is reduced after the sorting, which reduces the harmonic distortion of the SDM. This sorting algorithm is not sensitive to the common-mode error, which causes a DC offset of the SDM.

Figure 9.7 shows the simulated SNDR of 100 Monte-Carlo runs with the 3-bit DAC with normally distributed amplitude error $\sigma = 0.3\%$, before and after the sorting. The input signal is a sine wave with amplitude 0.85 V at 190 MHz. The signal bandwidth of the SDM is 600 MHz. Before applying the sorting, the SDM shows SNDR = 66.7 \pm 3.3 dB ($\pm \sigma$), SNR = 73.3 \pm 1.9 dB ($\pm \sigma$), THD = -68.4 \pm 4.6 dB ($\pm \sigma$). After applying the sorting, the SDM shows SNDR = 70.2 \pm 2.5 dB ($\pm \sigma$), SNR = 72.5 \pm 1.9 dB ($\pm \sigma$). In average, the SNDR is improved by 3.5 dB, and the THD is reduced by 6.8 dB.

Figure 9.8a shows a typical output spectrum of the SDM in those 100 Monte-Carlo runs, before and after the sorting. Figure 9.8b shows the INL of the DAC before and after the sorting. Figure 9.8b shows that after applying the sorting algorithm, the maximum absolute value of INL is reduced from about $10 \cdot 10^{-3}$ to about $6 \cdot 10^{-3}$ (relative value referred to the DAC's full-scale output). Figure 9.8a shows that the HD2 is reduced from -68 dBc to -73.4 dBc, and the SNDR is increased from 67.1 dB to 70.2 dB.

Notice that the focus of this chapter is not on the optimization algorithm. The focus and contribution of this chapter is on a new implementation of the DAC linearization techniques in broadband CT $\Sigma\Delta$ ADCs, which is a general purpose architecture supporting various static and dynamic linearization techniques.



Figure 9.7: Simulated SNDR of 100 Monte-Carlo runs with 3-bit DAC with normally distributed amplitude error $\sigma = 0.3\%$, before and after 1-dimensional sorting.

9.6 Conclusions

This chapter proposes a new implementation of the DAC linearization techniques in broadband CT $\Sigma\Delta$ ADCs with the mapping engine in the digital domain. The system architecture of the current-mode multi-path ELD compensation is used to implement DAC linearization techniques by sorting/mapping/switching the input codes of the reference DACs. Since the mapping engine is in the digital domain, its non-idealities do not interfere with the analog purity of the reference signal. It is a general purpose architecture and it offers the flexibility to implement different static and dynamic DAC linearization techniques. Simulation results show that a static sorting of a 3-bit DAC improves the SNDR in average by 3.5 dB, and improves the THD in average by 6.8 dB. These properties and simulation results show the potential of the proposed approach for broadband high-linearity $\Sigma\Delta$ ADC applications.



Figure 9.8: (a) Simulated typical output spectrum of the SDM before and after sorting (b) Integral non-linearity (INL) of the DAC before and after sorting for this run.

Chapter 10

Overcoming metastability error limitations

This chapter studies the impact of metastability errors in broadband $CT \Sigma \Delta$ ADCs. Metastability errors are introduced in Chapter 4.2 and they stem from the inability of the quantizers to take decisions within the allocated time. In the beginning, Section 10.1 describes the scope of this chapter. Section 10.2 reviews the state of the art in metastability error modeling and metastability error mitigation techniques. Section 10.3 describes the error mechanism that is responsible for the impact the metastability has on the performance of CT SDMs, and proposes a more general model to describe the impact of the metastability error in CT SDMs. To overcome the metastability problem, two novel system-level solutions are proposed. Section 10.4 first proposes a metastability error and restores the target SNDR. Next, Section 10.5 proposes a metastability shaping technique to shape and suppress the metastability error by the NTF, in a similar way as the quantization noise is suppressed by it [98]. With this technique, the required quantization gain can be generated in two T_S, instead of one T_S. In the end, conclusions are drawn.

10.1 Introduction

In this chapter, both a metastability error compensation technique (Section 10.4) and a metastability shaping technique (Section 10.5) are proposed and investigated at the conceptual level. The effectiveness of the proposed techniques is validated through mathematical analysis, Matlab modeling and simulations. The transistor-level implementation, test chip design and validation of these two techniques are out of the scope of this thesis because of time limitation.

This chapter takes the following assumptions in the analysis. (1) The metastability

is assumed to be caused by the limited quantization gain in the loop. All of the quantization gain in the $\Sigma\Delta$ loop is modeled in the quantizer. The DAC has no quantization gain (or its quantization gain is 1). The digital output logic of the SDM has infinite quantization gain, which is out of the $\Sigma\Delta$ loop (see Figure 10.1). (2) A linear gain model is assumed, as shown in equation (4.11). (3) Since the metastability error can be described as a stochastic effect, its probability is assumed to be independent from the quantizer offset, hysteresis and noise. Thus, the results of this chapter are also valid in the presence of these non-idealities. (4) The quantizer input is uniformly distributed in the defined input swing (-1, 1).

10.2 State of the art in metastability error modeling and mitigation

In this section the state of the art in metastability error modeling and metastability error mitigation techniques are presented. Chapter 4.2 discusses the 1-bit comparator's output characteristic with limited quantization gain in equation (4.11). Chapter 5.4 shows the simulated SNDR versus the quantization gain in Figure 5.9. When the quantizer cannot take a decision within its allocated time budget, the problem with the metastability error arises. In this case the SNDR of the SDM can significantly reduce.

10.2.1 State-of-the-art modeling of metastability error mechanism

In the literature, the variation in the zero-crossing time of the feedback DAC, caused by the metastability in the quantizer, is seen as the error mechanism that is responsible for the impact the metastability has on the performance of CT SDMs [61, 99]. In [61], the quantizer output drives the feedback DAC without re-sampling. The quantizer has shorter delay when its input is large, and it has longer delay when its input is small. Thus, in the case that the DAC input is switching, the DAC output crosses earlier when the quantizer input is larger. The limited quantization gain of the quantizer causes a variation of the DAC zero-crossing time depending on the quantizer input, which adds an unshaped error in the output spectrum of the SDM.

This thesis finds that the quantizer characteristics shown in Fig. 18 in [61] describes the quantizer metastability accurately. However, this thesis finds that the error mechanism shown in [61] (variation in the DAC zero-crossing time) is not very accurate in some cases when the metastability errors happens in broadband CT $\Sigma\Delta$ ADCs. One example is that when metastability errors occur in CT $\Sigma\Delta$ ADCs, it can happen that the DAC switches are not operating in the designed state. The designed state is that one switch is ON, and the other switch is OFF. When metastability errors occur, it can happen that both switches are ON (as will be discussed later on, see Figure 10.2). It means that the DAC output does not settle to the desired value. This phenomenon cannot be explained by the variation in the DAC zero-crossing time. Section 10.3 presents a proposed more-general error mechanism and modeling to describe the impact of the metastability error in CT SDMs.

10.2.2 State-of-the-art metastability error mitigation techniques

In the literature, three techniques that mitigate the metastability error in broadband CT $\Sigma\Delta$ ADCs are:

- Applying the ELD compensation to allow typically up to 1 T_S delay in the $\Sigma\Delta$ loop, such that more time is available for the quantizer to generate the required quantization gain.
- Re-sampling the DAC input close to the main DAC with a low-jitter clock, to reduce the errors caused by the clock jitter, the clock skew, and the metastability error [30].
- Tapping the SDM output logic as close as possible to the main DAC input, such that the SDM output is as close as possible to the correct digital representation of the DAC output [30].

A block diagram of a state-of-the-art broadband CT SDM with the aforementioned implementation details is presented in Figure 10.1. Here, the DAC driver is composed of a sampler, and/or a buffer, and/or a D-flipflop (D-FF). The quantizer output is re-sampled by the DAC driver with a DAC clock clk_{DAC} , and clk_Q is the quantizer clock. A_{fb} is the analog feedback signal in the SDM which is the main DAC output as well. D_{out} is the digital output of the SDM.

Many more technologies are proposed in the literature to mitigate the metastability error in $\Sigma\Delta$ ADCs. For example, the time constant τ of the comparator core can be reduced by increasing its current (and power consumption). However, the minimal achievable time constant τ is limited by the technology (e.g. by self loading), and in that case τ cannot be further reduced even if more power is consumed. Besides, time-interleaving of the quantizer can effectively move the reset time of the quantizer out of the ELD time budget, which can maximize the available regeneration time of the comparator core [24]. The state-of-the-art technologies are limited by the available regeneration time and the achievable time constant. In this chapter, two novel techniques are proposed to overcome the metastability error limitation. A metastability error compensation technique is proposed in Section 10.4, where the principle of feedforward correction is used, and the compensation effectiveness relies on the accuracy and matching of the compensation circuit. A metastability shaping



Figure 10.1: Block diagram of a broadband CT SDM with some implementation details $% \left({{{\rm{SDM}}}} \right)$

technique is proposed in Section 10.5, where the principle of feedback correction is applied, and it relies on the gain made in the feedback loop.

10.3 Proposed modeling of metastability error mechanism

This section describes the error mechanism that is responsible for the impact the metastability has on the performance of CT SDMs in detail. Afterwards, this section proposes a more general model to describe the impact of the metastability error in CT SDMs.

The following description refers to the block diagram of a state-of-the-art broadband CT SDM shown in Figure 10.1. The output of the quantizer is in the ideal case a perfect staircase signal (or a combination of ideal block waveforms) which on the one hand can be translated to a digital signal that forms the output of the SDM D_{out} , and on the other hand, via an ideal DAC, to an analog waveform A_{fb} which is made equal by the $\Sigma\Delta$ loop to the input signal A_{in} in the band of interest. The output of the SDM D_{out} is then the correct digital representation of the DAC output A_{fb} and thus of the input signal of the SDM A_{in} .

Assuming now a non-ideal quantizer with a finite quantization gain. In that case, the latches in the quantizer, which form the core of the comparator, can be not fast enough to reach the maximum output levels for certain (small) input signals. The DAC has no quantization gain, and it converts an unsaturated input to an unsaturated output A_{fb} . The SDM digital output logic has infinite quantization gain, and D_{out} is always a saturated digital signal. In this case, the SDM output D_{out} is no longer the digital counterpart of the DAC output A_{fb} , and the information inside and outside the $\Sigma\Delta$ loop misaligns. Within the signal bandwidth, the DAC output A_{fb} is made equal by



Figure 10.2: Metastability errors observed during simulations of an exemplary CT SDM (a) both DAC switches are ON for 1 T_S (b) one DAC switch is OFF, and the other switch is half-OFF for 1 T_S .

the $\Sigma\Delta$ loop to the input signal A_{in} , no matter the output waveform of the DAC. In this case, D_{out} is not the correct digital representation of A_{fb} and thus of A_{in} .

The analysis in the previous paragraph is based on the assumptions of a simplified SDM model given in Section 10.1. With a realistic circuit implementation of a broadband CT $\Sigma\Delta$ ADC, the metastability error is more complicated. When the quantizer output is metastable, the behavior of the DAC circuitry may be different from the desired operation. Figure 10.2 shows two examples of metastability errors that appear in the simulations of the schematic and post-layout model of the CT SDM in this

thesis. In these simulations, the quantizer is in post-layout, and the rest of the SDM is in schematic. Figure 10.2 shows the output current of a unit DAC cell of DAC 1 in the 1^{st} SDM stage (see Figure 5.6). It is implemented as a current-steering DAC with one-side switching (Figure 4.5a). The blue lines and red lines are the output current I_{out}^+ and I_{out}^- of the unit DAC cell. The SDM sampling clock period is about 143 ps $(F_S = 7 \text{ GHz})$. When no metastability error happens, the legal output of the unit DAC cell is $\pm \frac{1}{2}I_{DAC}$ (here about $\pm 170 \ \mu$ A). In Figure 10.2a, at t = 413.2 ns, both switches of the unit DAC cell are ON for $1 T_S$ because of a metastability error. Thus, the differential output current is about 0 A, which is illegal and D_{out} does not have such a level to represent it. In Figure 10.2b, at t = 885.2 ns, for $1 T_S$, one DAC switch is OFF, but the other DAC switch is not fully ON (it gradually turns "half ON"), because of a metastability error. In this case, because the current source I_{DAC} from the top is still ON, some extra charge is stored at V_s node (Figure 4.5a). Thus for the next T_S , even if the DAC input is a fully settled digital signal (no metastability), the extra charge at V_s node causes a strange and illegal waveform of the DAC output current. When a metastability error happens like Figure 10.2b, D_{out} cannot represent the real DAC feedback current, either.

The error mechanism that is responsible for the impact the metastability has on the performance of CT SDMs is summarized in the following. When the quantizer output is metastable, with certain probability, the SDM output D_{out} is no longer the digital counterpart of the DAC output A_{fb} . The DAC output A_{fb} is made equal by the loop to the input signal A_{in} , no matter the output waveform of the DAC. The difference between D_{out} and A_{fb} (both referred to their full-scale to make them unit-less) is defined as the metastability error of the CT SDM in this thesis, which is added out of the $\Sigma\Delta$ loop by the digital output logic.

$$E_{meta} = D_{out} - A_{fb} \tag{10.1}$$

In equation (10.1), A_{fb} is normalized by the full-scale feedback charge of the DAC. Figure 10.3 shows an illustrative plot of output currents of an ideal DAC and a real DAC implementation. In the left figure, the blue line shows the output current of an ideal DAC, when its input is not metastable. Here a non-return-to-zero DAC with rectangular pulse is considered. The area under the blue line is the full-scale feedback charge of the ideal DAC. The red line shows one case of the output current of the ideal DAC when the DAC's input is metastable (not saturated), and the ideal DAC has no quantization gain here. In the case of the red line, A_{fb} is defined as the area under the red line normalized by the area under the blue line. In the right figure, the green line shows the output current of a real DAC implementation, when the DAC input is not metastable. Green line shows a certain settling behavior of the DAC output current. The difference between the green line and the blue line causes both an amplitude error and a timing error (delay), which can change the NTF of the SDM. Both the amplitude error and the timing error can be compensated to restore the ideal NTF. The difference between the green line and the blue line is not the metastability error. The area under the green line is the full-scale feedback charge



Figure 10.3: Output currents of an ideal DAC and a real DAC implementation

of the real DAC implementation. The light brown line shows one case of the output current of the real DAC when the DAC's input is metastable. In the case of the light brown line, A_{fb} is defined as the area under the light brown line normalized by the area under the green line.

This thesis makes a few remarks on the definition of the metastability error in equation (10.1). (1) The formula assumes that the difference between D_{out} and A_{fb} is due completely to a metastability error. With a real circuit implementation, the DAC's noise and non-linearity make A_{fb} deviating from D_{out} as well, which is not taken into account in this discussion, and thus not in this definition of the metastability error. (2) In this definition, the DAC output A_{fb} is normalized by its full-scale feedback charge. The shape of the DAC pulse and the transient of the DAC settling, when the quantizer output is fully settled (not metastable), change the full-scale feedback charge of the DAC output. But those two effects do not have an impact on the definition of the metastability error here. (3) This definition is not restricted to the assumptions made in Section 10.1; it is generally valid for CT $\Sigma\Delta$ ADCs. It is still valid even if a more sophisticated model for the quantizer is used, or if the quantization gain and thresholds of the DAC driver and DAC are considered (in Figure 10.1). The metastability errors shown in Figure 10.2 are covered by this definition as well.

A simplified model of a CT SDM with the metastability error E_{meta} is shown in Figure 10.4, with a 1st order equivalent of the quantizer and the DAC. In this figure, some details in Figure 10.1 are omitted, such as the DAC driver. This model is generally valid for CT SDMs with metastability error and ELD compensation, irrelevant to the implementation details. When V is metastable, it is an imperfect analog staircase signal (due to the finite quantization gain of the quantizer). The 'DAC' function here represents the analog transfer function from its (analog) staircase input to its output, which represents the imperfect transfer function of the actual DAC block. All the quantization gain of the $\Sigma\Delta$ loop is modelled at the quantizer (defined as G_q). For the SDM output the situation is different: the output logic is translating the imperfect staircase signal of the quantizer to a discrete-amplitude output signal V' (that can be represented by a digital signal D_{out}). The difference between this output signal and the signal at the output of the DAC function is modelled by E_{meta} . V is the DAC input and the input of the digital output logic, and V' is the SDM



Figure 10.4: Block diagram of a CT SDM with metastability error

output which is considered always as a fully settled digital signal (V' in Figure 10.4 is equivalent to D_{out} in Figure 10.1). Under the assumptions in Section 10.1 and the aforementioned modeling, V and A_{fb} have the same information. The equation (10.1) can be re-written as:

$$E_{meta} = V' - V \tag{10.2}$$

If the sampled quantizer input is very small $(V_{in} < \frac{1}{G_q})$, where V_{in} is referred to its full-scale to make it unit-less), the quantizer output is not a fully settled signal. In this case, a metastability error happens. Because the metastability error E_{meta} is added out of the $\Sigma\Delta$ loop, it is not shaped by the NTF. The transfer function of the metastability error E_{meta} to the SDM output is 1. The result is that in the output spectrum of the SDM, the in-band noise level increases.

This effect is shown in Figure 10.5. Figure 10.5a shows the simulated output spectrum (2^{17} samples) of a 4^{th} order SDM with ideal quantizer, and its quantization noise shows 80 dB/dec slope. Figure 10.5b shows the output spectrum (2^{17} samples) when the total loop quantization gain $G_q = 80$ dB. In this case the metastability error happens with a certain probability (in average 8 times per 10000 samples, with the 9-level quantizer used in this simulation). The metastability error can be modelled as a white noise, and it is not shaped. So the output spectrum shows a flat increased in-band noise. The problem is that the metastability error can degrade the SNDR and the dynamic range of the $\Sigma\Delta$ ADC. Simulations show that with a quantization gain of 80 dB, the SNDR is reduced from 75.3 dB to 55.6 dB, by 19.7 dB, compared to the case with an ideal quantizer (see Figure 5.9).

10.4 Metastability error compensation

This section proposes a novel metastability error compensation (MEC) circuit, to measure and compensate the metastability error of the SDM. Firstly, its operation principle is presented in 10.4.1. The circuit non-idealities in the MEC circuit are considered in 10.4.2. In the end, conclusions are drawn.



Figure 10.5: (a) Simulated output spectrum with ideal quantizer (b) Simulated output spectrum with 80 dB quantization gain.

10.4.1 Operation principle

The operation principle of the proposed metastability error compensation circuit is introduced here. Figure 10.6 shows the block diagram of a CT $\Sigma\Delta$ ADC with the metastability error compensation circuit. The MEC circuit is composed of 2 extra quantizers (2^{nd} quantizer and 3^{rd} quantizer), 2 extra DACs (DAC 3 and DAC 4), an analog delay element, a digital delay element, an analog summation function and a digital summation function.

The operation principle of the metastability error compensation circuit is as follows. If the quantizer output in the SDM (represented as V in Figure 10.6) is in metastable state, it will be resampled by the 2^{nd} quantizer and amplified by the quantization gain of the 2^{nd} quantizer. The probability of the metastability error at the 2^{nd} quantizer output (represented as V' in Figure 10.6) is reduced by its quantization gain. The metastability error of the 1^{st} quantizer is converted partly to the quantization noise of the 2^{nd} quantizer, and partly to the metastability error of the 2^{nd} quantizer, depending on its quantization gain.

Both 1^{st} quantizer and 2^{nd} quantizer outputs are converted to analog signals by DAC 3 and DAC 4, respectively. The analog delay element delays DAC 3 output for the same time as the delay of the 2^{nd} quantizer (which is the regeneration time of the 2^{nd} quantizer). Then the output of the analog summation is the subtraction of the corresponding analog signals of V and V' for the same sample, which is the majority part of the metastability error of the SDM ($-E_{meta}$). So the output of the 3^{rd} quantizer is the digital representation of the majority part of the metastability error of the SDM. The digital delay element delays V' for the same time as the delay of the 3^{rd} quantizer. For the output of the digital summation (V_c in Figure 10.6), the majority part of the metastability error in the SDM is compensated, and only a



Figure 10.6: Block diagram of a CT SDM with the proposed metastability error compensation circuit

residue metastability error E'_{meta} with a much reduced power than E_{meta} is left (the detailed power calculation for E'_{meta} is shown later).

The requirement on the building blocks of the metastability error compensation circuit is as follows. The 2^{nd} quantizer should have the same resolution (number of quantization levels) as the 1^{st} quantizer. The 1^{st} quantizer is assumed to be an mlevel multi-bit quantizer (m ≥ 2), and it is composed of m-1 pre-amplifiers and m-1 latches. The pre-amplifiers compare the input signal with a series of references (voltage or current). Then, in this case, the 2^{nd} quantizer is composed of only m-1 latches. The 2^{nd} quantizer performs a re-quantization of the 1^{st} quantizer's output at the 1-bit level: every latch of the 2^{nd} quantizer directly samples the output of one fixed latch of the 1^{st} quantizer, and then performs the regeneration. The 2^{nd} quantizer needs to process all the quantization levels of the 1^{st} quantizer, since the information on which bit is metastable is not known to it.

Another requirement is that DAC 3 and DAC 4 should be the exact replica of DAC 1, which means that they have the same quantization gain as DAC 1, and they behave the same as DAC 1 even when their input is metastable. In the following analysis, the quantization gain of DAC 1, DAC 3 and DAC 4 are assumed to be all one, for simplicity. The quantization gain of the 1st quantizer and the 2nd quantizer are assumed to be G_{q1} and G_{q2} , respectively. For the moment, all the components are assumed to be ideal and the 3rd quantizer has infinite resolution.

The mathematical analysis of the metastability error compensation is described as follows for $\Sigma\Delta$ ADCs with 1-bit quantizers, with the help of the diagrams shown in Figure 10.7. Figure 10.7a shows the diagram of the quantization noise and the metastability error of the conventional $\Sigma\Delta$ ADC shown in Figure 10.4. The red line is the output characteristic of the quantizer in the SDM. The distance in y-axis



Figure 10.7: Diagram of the quantization noise and the metastability error: (a) of the conventional SDM shown in Figure 10.4; (b) of the SDM with the metastability error compensation circuit shown in Figure 10.6.

between the red line and the $V_{out} = V_{in}$ line is the quantization error of this quantizer (referred as the quantization noise as well), and it is denoted as q_1 . The quantization noise is added in the $\Sigma\Delta$ loop at the quantizer, and it is shaped by the NTF. The light red area is the metastability error in the conventional $\Sigma\Delta$ ADC, which is denoted as E_{meta} . The metastability error is added out of the $\Sigma\Delta$ loop by the digital output logic, and hence unshaped. In the conventional $\Sigma\Delta$ ADC, the metastability error happens with a probability of $\frac{1}{G_{q_1}}$. Thus, the probability density function of the metastability error is

$$f(x)_{meta} = \begin{cases} \frac{1}{2G_{q1}} & , \text{ if } -1 \le x \le 1\\ 0 & , \text{ if else} \end{cases}$$
(10.3)

The power of the metastability error in the conventional SDM can be calculated as

$$P_{E_{meta}} = \int_{x=-1}^{1} x^2 \cdot \frac{1}{2G_{q1}} dx = \frac{1}{3G_{q1}}$$
(10.4)

Similarly, the probability density function of the quantization noise of the quantizer in the SDM q_1 is

$$f(x)_{q_1} = \begin{cases} \frac{1}{2 \cdot (1 - \frac{1}{G_{q_1}})} & , \text{ if } - (1 - \frac{1}{G_{q_1}}) \le x \le 1 - \frac{1}{G_{q_1}} \\ 0 & , \text{ if else} \end{cases}$$
(10.5)

The power of the quantization noise q_1 of the quantizer in the SDM can be calculated as

$$P_{q_1} = \int_{x=-(1-\frac{1}{G_{q_1}})}^{1-\frac{1}{G_{q_1}}} x^2 \cdot \frac{1}{2 \cdot (1-\frac{1}{G_{q_1}})} dx = \frac{1}{3} \left(1 - \frac{1}{G_{q_1}}\right)^2$$
(10.6)

Even with relatively low total loop quantization gain G_{q1} (e.g. 40 dB), the power of the metastability error is far lower than the unshaped quantization noise. However, the quantization noise is in the $\Sigma\Delta$ loop and shaped by the designed noise transfer function. So the in-band quantization noise in the final output is very much suppressed. The metastability error is added out of the $\Sigma\Delta$ loop and directly on the SDM output, which is not shaped. So the power of the metastability error seems to be very low, but it can cause dramatic degradation on the overall performance of the $\Sigma\Delta$ ADCs.

Figure 10.7b shows the diagram of the quantization noise and the metastability error of the $\Sigma\Delta$ ADC with the metastability error compensation circuit shown in Figure 10.6. The blue line is the overall output characteristic of the 1st quantizer and the 2nd quantizer. The light green part is the same as in Figure 10.7a. The light blue area is the quantization noise of the 2nd quantizer. When the input voltage (referred to its full-scale) of the 1st quantizer is within $(\frac{1}{G_{q1}G_{q2}}, \frac{1}{G_{q1}})$ or $(-\frac{1}{G_{q1}}, -\frac{1}{G_{q1}G_{q2}})$, the output of DAC 1 and DAC 3 have metastability errors, and the output of DAC 4 has no metastability error because of the additional quantization gain of the 2nd quantizer G_{q2} . So at the output of DAC 4, the metastability error happens with a probability of $\frac{1}{G_{q1}G_{q2}}$, which is G_{q2} times lower than the conventional $\Sigma\Delta$ ADC. The metastability error at the output of DAC 4 is referred as the residual metastability error E'_{meta} , for a metastability error compensation circuit that is ideal, except for the limited quantization gain of the 2nd quantizer (this assumption is referred as the ideal metastability error compensation circuit in the following). Its power can, similarly as in (10.4), be calculated as

$$P_{E'_{meta}} = \frac{1}{3G_{2,tot}}$$
(10.7)

Here, $G_{2,tot} = G_{q1} \cdot G_{q2}$.

The probability density function of the quantization noise of the 2^{nd} quantizer (denoted as q_2) is

$$f(x)_{q_2} = \begin{cases} \frac{1}{2G_{q_1} \cdot (1 - \frac{1}{G_{q_2}})} & \text{, if } - (1 - \frac{1}{G_{q_2}}) \le x \le 1 - \frac{1}{G_{q_2}}\\ 0 & \text{, if else} \end{cases}$$
(10.8)

The power of the quantization noise of the 2^{nd} quantizer is

$$P_{q_2} = \frac{1}{3G_{q1}} \left(1 - \frac{1}{G_{q2}} \right)^2 \tag{10.9}$$

Notice that the original metastability error E_{meta} in Figure 10.4 is divided into two parts in the $\Sigma\Delta$ ADC with the metastability error compensation circuit, and q_2 is the majority part.

$$E_{meta} = q_2 + E'_{meta} \tag{10.10}$$

The corresponding analog signal of the quantization noise of the 2^{nd} quantizer q_2 is actually the output of the analog summer in Figure 10.6 with a minus sign. This signal is quantized by the 3^{rd} quantizer and subtracted from the digital output of the SDM. So in the $\Sigma\Delta$ ADC with an ideal metastability error compensation circuit in Figure 10.6, its final output after compensation V_c only contains unshaped residual metastability error E'_{meta} and shaped (by the designed NTF) quantization noise of the 1^{st} quantizer q_1 , and the quantization noise of the 2^{nd} quantizer q_2 is completely canceled.

The metastability error for unary multi-bit quantizers can be analyzed similarly. For an m-level quantizer (considered as a flash sub-ADC here) composed of m-1 comparators with a quantization gain of G_{q1} , the equation (10.4) can be re-written as:

$$P_{E_{meta}} = \int_{x=-\frac{1}{m-1}}^{\frac{1}{m-1}} x^2 \cdot \frac{(m-1)^2}{2G_{q1}} \, dx = \frac{1}{3 \cdot (m-1) \cdot G_{q1}} \tag{10.11}$$

Figure 10.8 shows the simulated 3^{rd} quantizer input signal of a $\Sigma\Delta$ ADC with the metastability error compensation circuit whose 1^{st} quantizer and 2^{nd} quantizer are 9-level, and $G_{q1} = 80$ dB, $G_{q2} = 40$ dB. The time domain signal has a specific property – for a probability of $1 - \frac{8}{G_{q1}} = 99.92\%$ (8 in the numerator because the 9-level quantizer is composed of 8 1-bit comparators, and any comparator can become metastable), no metastability error happens in the $\Sigma\Delta$ loop, and the 3^{rd} quantizer input is 0. And for a probability of 0.08%, a metastability error happens in the $\Sigma\Delta$ loop, and the 3^{rd} quantizer input is within \pm 0.5 LSB₁ of the 1^{st} quantizer (in this case it is within \pm 1/8; LSB₁ denotes the least significant bit of the 1^{st} quantizer).

In practice, the 3^{rd} quantizer has limited resolution, and it has its own quantization noise. One important design consideration is, when no metastability error happens in the $\Sigma\Delta$ loop (99.92% probability in Figure 10.8), the 3^{rd} quantizer should NOT introduce a new error. This is achieved by giving the 3^{rd} quantizer a legal output level 0. In this case, when the 3^{rd} quantizer input is within ± 0.5 LSB₃ (LSB₃ denotes the least significant bit of the 3^{rd} quantizer), the 3^{rd} quantizer output is 0. Later it will be shown that the metastability error compensation circuit has a ± 0.5 LSB₃ error-insensitive interval thanks to the legal output level 0.

If the 3^{rd} quantizer has a legal output 0, when its input is 0 (99.92% probability in Figure 10.8), its output is 0 as well. In this case, the 3^{rd} quantizer does nothing, and hence does not introduce any error. In other words, only when the metastability errors happen in the $\Sigma\Delta$ loop ($\frac{m-1}{G_{q1}} = 0.08\%$ probability in Figure 10.8), the 3^{rd} quantizer is effectively working, and its quantization noise is important. The power of the


Figure 10.8: Simulated 3^{rd} quantizer input when $G_{q1} = 80$ dB, and $G_{q2} = 40$ dB

introduced error in the final output V_c by the 3^{rd} quantizer with limited resolution can be calculated as

$$P_{E_{q3}} = \frac{m-1}{G_{q1}} \cdot P_{q3} = \frac{m-1}{G_{q1}} \cdot \frac{LSB_3^2}{12}$$
(10.12)

In equation (10.12), E_{q3} is the introduced error in the final output by the 3^{rd} quantizer, q_3 is the quantization error of the 3^{rd} quantizer, and m is the number of quantization levels of the 1^{st} and 2^{nd} quantizer.

Because of this property, the 3^{rd} quantizer does not need a very high resolution, even though the target resolution of the $\Sigma\Delta$ ADC after compensation is very high. E.g. from simulation, for a 4^{th} order $\Sigma\Delta$ ADC with 9-level quantizer in the $\Sigma\Delta$ loop and $G_{q1} = 80$ dB, when the 3^{rd} quantizer has 17 quantization levels (4.09 bit) for the input swing (-1/8, 1/8), the overall output after compensation V_c achieves SNDR = 75.1 dB (12.2 ENOB). In this case the SNDR is limited by the in-band shaped quantization noise $NTF \cdot q_1$ (SNDR = 76.5 dB for the ideal $\Sigma\Delta$ ADC). Notice that without the metastability error compensation circuit, SNDR = 60.9 dB which is completely limited by the in-band metastability error. In this case, the simulated output spectrum with and without the metastability error compensation circuit for 2^{17} samples is shown in Figure 10.9. For this exemplary 4^{th} order $\Sigma\Delta$ ADC, the OSR is 10.

Figure 10.10 presents the simulated SNDR versus total loop quantization gain (G_{q1}) with and without the metastability error compensation circuit. The $\Sigma\Delta$ ADC uses a 9-level quantizer in the $\Sigma\Delta$ loop. The MEC circuit employs a 17-level 3^{rd} quantizer for its input swing (-1/8, 1/8). Without the MEC, to achieve SNDR ≥ 75 dB, the total loop quantization gain of the $\Sigma\Delta$ loop (G_{q1}) should be at least 120 dB.



Figure 10.9: Simulated output spectrum of a 4^{th} order SDM with and without metastability error compensation circuit, $G_{q1} = 80$ dB, $G_{q2} = 40$ dB



Figure 10.10: Simulated SNDR versus total loop quantization gain (G_{q1}) with and without the metastability error compensation circuit with a 17-level 3^{rd} quantizer

With the metastability error compensation circuit, the requirement for the total loop quantization gain (G_{q1}) is relaxed to 80 dB. In this case, the quantization gain of the 2^{nd} quantizer should be at least 40 dB.

This thesis crosses check the simulation results in Figure 10.9 and 10.10 with the equations (10.11) and (10.12). In Figure 10.9 and 10.10, a sine wave with an amplitude of 1 (referred to SDM input full scale) is used as the SDM input. The signal power

 $P_s = \frac{1^2}{2} = 0.5 \tag{10.13}$

For a 9-level quantizer with a quantization gain G_{q1} of 80 dB (red line in Figure 10.9), the calculated metastability error power with equation (10.11) is $P_{meta,80dB} = 4.2 \cdot 10^{-6}$. And the calculated metastability error limited SNR is:

$$SNR_{meta,80dB} = 10 \cdot log_{10} \frac{P_s \cdot OSR}{P_{meta,80dB}} = 60.8 \text{ dB}$$
 (10.14)

The in-band metastability error power is much larger than the in-band quantization noise power. The calculated $SNR_{meta,80dB}$ matches very well with the simulation results in Figure 10.9 and 10.10.

For a 9-level quantizer with a quantization gain G_{q1} of 120 dB, the calculated metastability error power with equation (10.11) is $P_{meta,120dB} = 4.2 \cdot 10^{-8}$. And the calculated metastability error limited SNR is:

$$SNR_{meta,120dB} = 10 \cdot log_{10} \frac{P_s \cdot OSR}{P_{meta,120dB}} = 80.8 \text{ dB}$$
 (10.15)

In this case, the in-band metastability error power is less than the in-band quantization noise power, so the overall SNR (75.8 dB) is not degraded much compared with the SNR of the ideal SDM (76.5 dB).

For a 9-level quantizer with a quantization gain G_{q1} of 80 dB and the MEC with a 17level 3^{rd} quantizer (blue line in Figure 10.9), the quantization noise power introduced by the 3^{rd} quantizer is:

$$P_{E_{q3}} = \frac{8}{10^4} \cdot \frac{(\frac{1}{16} \cdot \frac{2}{8})^2}{12} = 1.6 \cdot 10^{-8}$$
(10.16)

The in-band 3^{rd} quantizer's quantization noise power (unshaped) is less than the in-band 1^{st} quantizer's quantization noise power (shaped by the NTF), so a 17-level quantizer offers enough resolution as the 3^{rd} quantizer.

In the simulations, the probability and the histogram of the metastability error are compared with the analysis. The simulations prove that the assumption that the quantizer input is uniformly distributed in the defined input swing (-1, 1) is valid, and the analysis and calculation are correct.

10.4.2 Consideration of the circuit non-idealities

This section so far considers only the limited resolution of the 3^{rd} quantizer. In this sub-section, the other main non-idealities of the building blocks of the metastability error compensation circuit is considered.

is:

As mentioned in the last sub-section, because the 3^{rd} quantizer is designed to have a legal output level 0, if the 3^{rd} quantizer input is within \pm 0.5 LSB₃, the 3^{rd} quantizer output is 0. When no metastability error happens in the $\Sigma\Delta$ loop (with a very high probability of $1 - \frac{m-1}{G_{q1}}$), the ideal input of the 3^{rd} quantizer is 0, and the desired output is 0. So if the circuit non-idealities (e.g. mismatch of the unit elements of DAC 3 and DAC 4, the sampled thermal noise at the input of the 3^{rd} quantizer, etc.) introduce a small error at the input of the 3^{rd} quantizer which is less than \pm 0.5 LSB₃, it introduces no error in the final output of the SDM V_c for a very high probability. In short, at the input node of the 3^{rd} quantizer, if the total error caused by the circuit non-idealities is less than \pm 0.5 LSB₃, it is suppressed by the total loop quantization gain of the $\Sigma\Delta$ loop (G_{q1}) in the final output V_c .

$$P_{E_{DAC2,mm}} = \frac{m-1}{G_{q1}} \cdot P_{DAC2,mm}$$
(10.17)

$$P_{E_{DAC3,mm}} = \frac{m-1}{G_{q1}} \cdot P_{DAC3,mm}$$
(10.18)

$$P_{E_{Vin3,tn}} = \frac{m-1}{G_{q1}} \cdot P_{Vin3,tn}$$
(10.19)

In equation (10.17) – (10.19), $P_{E_{DAC2,mm}}$, $P_{E_{DAC3,mm}}$ and $P_{E_{Vin3,tn}}$ represent the power of the errors in the final output V_c caused by the unit element mismatch of DAC 3, the unit element mismatch of DAC 4, and the sampled thermal noise on the 3^{rd} quantizer input, respectively. And $P_{DAC2,mm}$, $P_{DAC3,mm}$ and $P_{Vin3,tn}$ represent the original power of the errors caused by the unit element mismatch of DAC 3, the unit element mismatch of DAC 4, and the sampled thermal noise on the 3^{rd} quantizer input, respectively. In this case, the interval (-0.5 LSB₃, 0.5 LSB₃) is defined as the error-insensitive interval.

However, if the total error at the 3^{rd} quantizer input is (with a non-negligible probability) higher than 0.5 LSB₃ or lower than -0.5 LSB₃, it means that even when no metastability error happens in the $\Sigma\Delta$ loop, the 3^{rd} quantizer output is not 0 but LSB₃ or -LSB₃. In this case, some extra error is introduced into the final output V_c even when no metastability error happens, and hence the total error power is not suppressed by the total loop quantization gain. It degrades the overall resolution of the SDM a lot, which can be even lower than the conventional SDM without metastability error compensation circuit.

As shown in Figure 10.11, the error-insensitive interval can be increased by removing two 1-bit comparators with reference -0.5 LSB_3 and 0.5 LSB_3 . Thus, when the 3^{rd} quantizer input is within (-1.5 LSB₃, 1.5 LSB₃), its output is 0. In this case the errorinsensitive interval is enlarged by a factor of 3. Then, the requirement of the DAC 3 and DAC 4 unit element mismatch, and the input thermal noise of the 3^{rd} quantizer, etc. is further relaxed. This quantizer is referred to the non-uniform quantizer in the following. Its drawback is a slightly reduction of the SNDR because when the



Figure 10.11: Output characteristic of a 17-level uniform quantizer (solid blue line) and a 15-level non-uniform quantizer (dash red line). The complete input and output swing are not shown here.

metastability error in the $\Sigma\Delta$ loop is in the interval of (-1.5 LSB₃, -0.5 LSB₃) or (0.5 LSB₃, 1.5 LSB₃), the output of this non-uniform quantizer is 0, and this small metastability error is not compensated. From simulation, compared to the 17-level uniform 3^{rd} quantizer, with the 15-level non-uniform quantizer the SNDR drops by 0.6 dB.

Similar as the DAC 3 and DAC 4 unit element mismatch error, and the sampled thermal noise at the 3^{rd} quantizer input node, thanks to the error-insensitive interval, the performance of the metastability error compensation circuit is not sensitive to the offset errors on the reference of the 3^{rd} quantizer. From the simulation, the standard deviation of the reference offset error should be less than 4% of its input swing $\left(-\frac{1}{8}, \frac{1}{8}\right)$ for the 15-level non-uniform 3^{rd} quantizer.

Because the 3^{rd} quantizer is not in a feedback loop, the requirement of its quantization gain is very low. From the simulation, higher than 20 dB quantization gain is enough for the 3^{rd} quantizer. As the 3^{rd} quantizer is not in a feedback loop, only its throughput is required to be the same as the sampling rate of the SDM, but its latency is not required to be one clock period. The digital delay in Figure 10.6 should equal the latency of the 3^{rd} quantizer. Thus the 3^{rd} quantizer can be implemented as e.g. a pipelined ADC (instead of a flash ADC) to reduce the number of comparators and its power consumption.

Table 10.1 summarizes the requirements on the building blocks of the metastability error compensation circuit. In all simulations, the SDM uses a 9-level quantizer in the $\Sigma\Delta$ loop, and the total loop quantization gain is 80 dB. The SNDR target after

Scenarios	17-level uniform 3^{rd} quantizer	15-level non-uniform 3^{rd} quantizer
DAC 3 and DAC 4 linearity	8-bit	7-bit
2^{nd} quantizer quantization gain	$\geq 40 \text{ dB}$	$\geq 40 \text{ dB}$
3^{rd} quantizer quantization gain	$\geq 20 \text{ dB}$	$\geq 20 \text{ dB}$
3^{rd} quantizer reference offset σ	1.3% of $\left(-\frac{1}{8}, \frac{1}{8}\right)$	4% of $\left(-\frac{1}{8}, \frac{1}{8}\right)$
Sampled thermal noise at 3^{rd} quantizer input	\leq -55 dB	\leq -46 dB

Table 10.1: Summary of the requirements on the building blocks of the metastability error compensation circuit

compensation is 74 dB (12 ENOB), including all considered circuit non-idealities in the metastability error compensation circuit. The ideal $\Sigma\Delta$ ADC has SNDR = 76.5 dB. It shows that even for an overall resolution target of 12 ENOB, the requirements of the building blocks of the MEC circuit are relatively low, mainly thanks to the error-insensitive interval. The MEC circuit is promising to be implemented in modern CMOS technologies.

The aforesaid analysis uses single-loop $\Sigma\Delta$ ADCs as examples. The metastability error compensation circuit can be implemented in MASH $\Sigma\Delta$ ADCs as well. For MASH $\Sigma\Delta$ ADCs, generally the metastability error generated in the 1^{st} $\Sigma\Delta$ loop is dominant in the overall metastability error in the final output, because the metastability error generated in the back-end $\Sigma\Delta$ loops is shaped by the noise transfer function of the preceding $\Sigma\Delta$ loop(s). So typically it is enough to apply the metastability error compensation circuit only to the 1^{st} $\Sigma\Delta$ loop output.

10.4.3 Power, area, and performance considerations

The MEC circuit is proposed to overcome the metastability limitation and enable a higher sampling rate and signal bandwidth for a certain technology. For the 3bit 2-1-1 MASH SDM test chip designed in this thesis (Chapter 11), the practically achievable minimum time constant of the comparator core is about 6.5 ps, considering the layout effect and loading. Without the MEC circuit, 120 dB of quantization gain is required for the 1st quantizer (Figure 10.10). Assuming the regeneration time of the quantizer is maximized to be 0.5 T_S (the other 0.5 T_S is for the propagation time from the quantizer to the DAC, and the DAC delay), the maximal achievable sampling rate of the SDM is

$$F_{S,max,w/o\ MEC} = \frac{1}{\ln(10^{\frac{120}{20}}) \times 6.5 \cdot 10^{-12} \times 2} \approx 5.57\ \text{GHz}$$
(10.20)

With the proposed MEC circuit, the quantization gain requirement for the 1^{st} quantizer is reduced to 80 dB. Two scenarios are considered in the estimation of the maximal achievable sampling rate of the SDM with the MEC circuit. In the first scenario, assuming the quantizer regeneration time is 0.5 T_S , the maximal achievable sampling rate of the SDM is increased to

$$F_{S,max,with MEC,case1} = \frac{1}{ln(10^{\frac{80}{20}}) \times 6.5 \cdot 10^{-12} \times 2} \approx 8.35 \text{ GHz}$$
(10.21)

In the second scenario, assuming the propagation delay from the quantizer to the DAC and the DAC delay is independent from the MEC circuit, and only the regeneration time of the quantizer can be reduced by applying the MEC circuit, the maximal achievable sampling rate of the SDM is

$$F_{S,max,with MEC,case2} = \frac{1}{(ln(10^{\frac{80}{20}}) + ln(10^{\frac{120}{20}})) \times 6.5 \cdot 10^{-12}} \approx 6.68 \text{ GHz}$$
(10.22)

The aforementioned example shows that when the maximal achievable sampling rate (and signal BW) of the state-of-the-art SDM for a certain technology is limited by the metastability error to $F_{S,max,w/o\ MEC}$, the proposed MEC circuit can mitigate the metastability limitation and enable a $1.2 \times \sim 1.5 \times$ higher maximal achievable sampling rate.

The extra power and area needed for the MEC circuit should also be considered. The quantization gain requirement of the 2^{nd} quantizer is about $2\times$ less than the 1^{st} quantizer in dB, which means the time constant of the 2^{nd} quantizer can be $2\times$ larger than the 1^{st} quantizer to save power. The quantization gain requirement of the 3^{rd} quantizer is only ≥ 20 dB, which can be achieved with power-efficient circuit. DAC 3 and DAC 4 can be implemented as a scaled version of DAC 1 to save power, at the cost of slightly degraded matching performance [100]. The analog delay and summation can be implemented with a sample-and-hold circuit, which is passive [101]. The total power consumption of the MEC circuit is estimated to be $0.5 \times \sim 1 \times$ of that of the main $\Sigma\Delta$ loop, considering the $\Sigma\Delta$ loop consumes power for loop filter and ELD compensation, which are not needed in the MEC circuit [100].

DAC 3 and DAC 4 should be the exact copy or a scaled copy of DAC 1, to guarantee the required matching. The rest blocks of the MEC circuit do not have stringent matching requirements. The area of the MEC circuit is estimated to be $0.5 \times \sim 1 \times$ of that of the main $\Sigma\Delta$ loop as well.

The minimum time constant of the comparator core is limited by the technology. Without the MEC circuit, even if the power of the MEC circuit is spent in the 1^{st} quantizer, its time constant cannot be further reduced (e.g. since self-loading is dominant), and the SDM sampling rate is still limited by the metastability error and thus cannot be increased.

10.4.4 Conclusions

A novel metastability error compensation circuit is proposed in this section. It reproduces the metastability error out of the $\Sigma\Delta$ loop, measures it and compensates it in the digital domain. For the exemplary 4th order 9-level $\Sigma\Delta$ ADC, the MEC circuit can relax the total quantization gain requirement in the $\Sigma\Delta$ loop by 40 dB for the same 74 dB SNDR target. The unique property of the error-insensitive interval is introduced in the MEC circuit. Thanks to the error-insensitive interval, the requirements on the building blocks in the MEC circuit are not stringent and the MEC circuit is promising for implementation in modern CMOS technologies. The proposed MEC circuit overcomes the metastability limitation, and enables CT $\Sigma\Delta$ ADCs with $1.2 \times \sim 1.5 \times$ higher maximal sampling rate, achieving broader bandwidth and/or higher resolution.

10.5 Metastability shaping technique

The previous section proposes a MEC circuit to reproduce the metastability error out of the $\Sigma\Delta$ loop, measure it and compensate it. This section proposes an alternative solution to combat the metastability error – noise shaping the (original) metastability error. With this technique, more quantization gain needs to be generated in the $\Sigma\Delta$ loop, which means more regeneration time is needed for the quantizer (assuming the minimum time constant τ is limited, see equation (4.10)). Thus, ELD compensation is relevant in this section.

10.5.1 Conventional ELD compensation

The ELD compensation has been shown in Chapter 4.5. Figure 10.4 depicts the block diagram of an exemplary CT SDM with the metastability error and the ELD compensation. This section takes a 4^{th} order single-loop 1-bit CT SDM as an example. However, the proposed metastability shaping technique can be applied to other CT SDM architectures as well, including multi-bit SDMs and MASH structures.

The impact of the metastability error for the exemplary 4^{th} order 1-bit SDM is shown in Figure 10.12. Compared to Figure 10.5, this figure shows the same information on the impact of the metastability error, but now for a different SDM example. The blue line shows the simulated output spectrum (2^{19} samples) of a 4^{th} order SDM with an ideal 1-bit quantizer. The red line shows the output spectrum (2^{19} samples) when the total loop quantization gain $G_1 = 60$ dB. In this simulation, the simulated SNDR of the SDM with an ideal quantizer is 81.2 dB. The simulated SNDR is dropped to 46.5 dB, for the conventional SDM with total loop quantization gain of 60 dB.



Figure 10.12: Simulated output spectrum of a 4^{th} order 1-bit SDM with and without metastability error

10.5.2 Proposed metastability shaping technique

The block diagram of a continuous-time $\Sigma\Delta$ ADC with the proposed metastability shaping technique is shown in Figure 10.13. Two quantizers are used in the SDM. One T_S delay (z^{-1} in the z-domain) is modelled for each quantizer, including its regeneration time, the DAC delay and propagation delay from the quantizer to the DACs. The direct feedback loop from quantizer 1 output (V1) to its input through DAC 3 is the inner ELD compensation loop, which is the 1^{st} order feedback loop of the SDM. The feedback loop from quantizer 2 output (V) to quantizer 1 input through DAC 2 is the outer ELD compensation loop, which is the 2^{nd} order feedback loop of the SDM. The feedback loop through DAC 1 is the main feedback loop of the SDM, which has stringent requirement on the total quantization gain. The advantage of the proposed technique is as follows. A 2^{nd} ELD compensation loop can compensate the delay of another one T_S , allowing more time to generate more quantization gain with a 2^{nd} quantizer in the $\Sigma\Delta$ loop. The inner ELD compensation loop has the quantization gain from only one quantizer generated from $1 T_S$ timing budget, but fortunately this loop does not require high quantization gain. The outer ELD compensation loop and the main feedback loop have the quantization gain from two quantizers generated from 2 T_S timing budget, which fulfills the quantization gain requirement to suppress the metastability error. The metastability shaping technique is an example of nested loop design, whose advantage is to offer the freedom of separate optimization for decoupled problems.

The proposed metastability shaping technique can also be applied to multi-bit SDMs. In that case, quantizer 2 should have the same resolution (number of quantization levels) as quantizer 1. Quantizer 2 performs a re-quantization of the quantizer 1's



Figure 10.13: Block diagram of a continuous-time $\Sigma\Delta$ ADC with the proposed metastability shaping technique

output at the 1-bit level: every latch of quantizer 2 directly samples the output of one fixed latch of quantizer 1, and then performs the regeneration. Quantizer 2 does not need a reference input.

With the metastability shaping technique, the total quantization gain of the main feedback loop through DAC 1 is actually the summation of quantization gain of quantizer 1 and quantizer 2. Because quantizer 2 has extra regeneration time (one T_S minus the propagation delay and DAC delay) to generate quantization gain, the total quantization gain in the main feedback loop can be much larger than the conventional SDM in Figure 10.4. On the other hand, although the inner ELD compensation loop through DAC 3 has only the quantization gain.

To map the SDM to the designed noise transfer function, the loop filter feedback transfer function $H'_n(s)$ and its parameters need to be re-calculated, compared with the loop filter in the conventional SDM $H_n(s)$. Here in Figure 10.4 and 10.13, the loop filter is drawn as in the feed-forward structure. But the proposed metastability shaping technique can be used for SDMs with feedback structure or any hybrid feed-forward/feedback structures as well. Moreover, the proposed technique can be further extended to using 3 and more quantizers.

The mathematical analysis of the metastability shaping technique is described as follows for $\Sigma\Delta$ ADCs with 1-bit quantizers. The case with multi-bit quantizers is similar. Actually, it has some similarity compared with the mathematical analysis of the metastability error compensation circuit shown in the previous section. The diagrams in Figure 10.7 can still be used. Figure 10.7a represents the quantization noise and the metastability error for the conventional SDM in Figure 10.4. Similar as in the last section, the red line shows the output characteristic of the quantizer in the conventional SDM in Figure 10.4. The light green area is the quantization noise q_1 , and the light red area is the metastability error of the conventional SDM E_{meta} . If the same definition of the quantization gain of the 1^{st} quantizer G_{q1} is made, equations (10.3) - (10.6) and (10.11) hold for the conventional SDM in Figure 10.4 as well.



Figure 10.14: Calculated noise transfer functions for the quantization noise of quantizer 1 and quantizer 2 $\,$

Figure 10.7b represents the quantization noise and the metastability error for the SDM with the proposed metastability shaping technique in Figure 10.13. Similar as in the previous section, the blue line shows the overall output characteristic of the quantizer 1 and quantizer 2. The light green area is the quantization noise of quantizer 1 q_1 , and the light blue area is the quantization noise of quantizer 2 q_2 . The light red area is the metastability error E'_{meta} added at the SDM output for the SDM with the proposed metastability shaping technique. If the same definition of the quantization gain of the 2^{nd} quantizer G_{q_2} is made, equations (10.7) - (10.10) hold for the SDM with the metastability shaping technique in Figure 10.13 as well.

Both of the quantization noise of quantizer 1 and quantizer 2 are shaped by a fullorder noise transfer function, as shown in Figure 10.14. Full-order means the same order as the loop filter. The blue line shows the ideal noise transfer function of the conventional SDM. The dotted red line shows the actual NTF of the quantization noise of the quantizer 1 (q_1) , which is exactly the same as the ideal NTF. With the design freedom offered by the SDM parameters, only one actual NTF (either for the quantization noise of the quantizer 1 or that of the quantizer 2) can be mapped to the ideal NTF. The actual NTF of q_1 is chosen to be mapped to be the same as the ideal NTF, and thus the actual NTF of quantization noise of the quantizer 2 is the result of this mapping. The dashed green line shows the actual NTF of the quantization noise of the quantizer 2 (q_2) . Its in-band part overlaps with the ideal NTF, while its high-frequency part is a bit different. Its out-of-band gain is slightly higher than the ideal NTF, but it does not harm the stability of the SDM, because the power of the quantization noise of quantizer 2 (q_2) is much lower than the quantization noise of quantizer 1 (q_1) .

Figure 10.15 shows the simulated output spectrum (2^{19} samples) of the 4^{th} order 1-bit



Figure 10.15: Simulated output spectrum of a 4^{th} order SDM with and without the metastability shaping technique, each quantizer has 60 dB gain



Figure 10.16: Simulated SNDR versus the quantization gain per quantizer with and without the metastability shaping technique

SDM with and without the proposed metastability shaping technique. Each quantizer has 60 dB quantization gain in this simulation ($G_{q1} = G_{q2} = 60$ dB). The simulated SNDR with the proposed metastability shaping technique is 81 dB, which is the same as the ideal SDM with an ideal quantizer.

Figure 10.16 presents the simulated SNDR versus the quantization gain per quantizer with and without the proposed metastability shaping technique. With the metastability shaping technique, the quantization gain of quantizer 1 and quantizer 2 are as-

sumed to be equal. Without the metastability error compensation circuit, to achieve SNDR ≥ 80 dB, the total quantization gain of the $\Sigma\Delta$ loop should be at least 120 dB. With the metastability shaping technique, the requirement for the quantization gain per quantizer is relaxed to 60 dB. For a quantization gain of 60 dB per quantizer, the metastability shaping technique improves the SNDR of the SDM from 47.3 dB to 80.8 dB by 33.5 dB.

10.5.3 Conclusions

This section presents a novel metastability shaping technique for broadband CT $\Sigma\Delta$ ADCs. A 2^{nd} ELD compensation loop can compensate the delay of another one T_S , allowing more time to generate more quantization gain with a 2^{nd} quantizer in the $\Sigma\Delta$ loop. The quantization noise from both quantizers experiences the same in-band noise shaping of the original full-order NTF, and the stability of the SDM is not sacrificed. For a quantization gain of 60 dB per quantizer, the metastability shaping technique improves the SNDR of the SDM by 33.5 dB. The proposed metastability shaping technique breaks the metastability limitation for broadband CT $\Sigma\Delta$ ADCs, and enables CT $\Sigma\Delta$ ADCs with broader bandwidth, higher resolution and better power efficiency.

10.6 Conclusions

This chapter proposes two system-level solutions to combat the metastability errors. For CT $\Sigma\Delta$ ADCs, the metastability error is added out of the $\Sigma\Delta$ loop at the SDM output, and hence it is not shaped by the NTF. In the first approach, a novel metastability error compensation circuit is presented, which reproduces the metastability error out of the $\Sigma\Delta$ loop, measures it and compensates it in the digital domain. The power of the metastability error is linked to the quantization gain via equations. For the exemplary 4^{th} order 9-level $\Sigma\Delta$ ADC, the MEC circuit can relax the total quantization gain requirement in the $\Sigma\Delta$ loop by 40 dB for the same 74 dB SNDR target, which can result in increasing the sampling rate and signal bandwidth. In the second approach, a metastability shaping technique is shown, which employs a 2^{nd} ELD compensation loop to compensate the delay of another one T_S . This extra one T_{S} of compensated delay is used by a 2^{nd} quantizer to generate the required additional quantization gain. Thus, the majority of the original metastability error is transferred to the quantization noise of the 2^{nd} quantizer and shaped by the same order of NTF. For a moderate quantization gain of 60 dB per quantizer, the metastability shaping technique improves the SNDR of the exemplary 4^{th} order SDM by 33.5 dB. Both approaches can relax the quantization gain requirement of the quantizer in one T_S timing budget significantly, and they are promising for implementation in modern CMOS technologies. These two novel approaches can overcome the metastability limitations for broadband CT $\Sigma\Delta$ ADCs, and enables CT $\Sigma\Delta$ ADCs with broader bandwidth, higher resolution and better power efficiency.

Part III

Test chip design and validation

Chapter 11

A 6 GHz 500 MHz BW 2-1-1 MASH CT $\Sigma \triangle$ ADC

This chapter presents the circuit-level implementation, simulation and validation results of a 6-GHz-sampling 500-MHz-BW 2-1-1 MASH $CT \Sigma \Delta$ ADC test chip. Section 11.1 shows the circuit-level implementation of the test chip, including perspectives regarding schematic and layout design. Section 11.2 presents schematic and post-layout simulation results. Section 11.3 presents the measurement results. The validation results are compared to the state-of-the-art broadband $CT \Sigma \Delta$ ADCs in Section 11.4. Section 11.5 concludes this chapter.

11.1 Circuit-level implementation

This section presents the circuit-level implementation of a 2-1-1 MASH CT $\Sigma\Delta$ ADC test chip in 40 nm CMOS technology. First, this section shows the implementation block diagram of the MASH $\Sigma\Delta$ ADC. Afterwards, the circuit-level implementation of the main building blocks – loop filter, DACs, quantizers and clock circuitry are elaborated. This section further discusses some measurement-assistance blocks, including demultiplexers (demux) and on-chip memory. In the end, the layout floorplanning of the MASH $\Sigma\Delta$ ADC is presented.

For this ADC test chip, the following design concepts presented in Part II are implemented on chip:

- The current-mode multi-path ELD compensation (Chapter 7.2).
- The techniques to overcome the parasitic capacitance on the summation node (Chapter 7.3).
- The current-mode locally-time-interleaved multi-bit quantizer (Chapter 8.1).



Figure 11.1: Implementation block diagram of the CT 2-1-1 MASH SDM. It is drawn as single-ended here, but the actual implementation is differential.

• The comparator offset calibration (Chapter 8.2).

Besides, the registers shown in Figure 9.6 are implemented on chip to enable applying a static sorting technique on the unit DAC elements. However, the mismatch sensor and logic in Figure 9.6 are not implemented. Moreover, the metastability error compensation technique and the metastability shaping technique shown in Chapter 10 are not implemented on chip.

11.1.1 2-1-1 MASH SDM architecture

Chapter 5 shows the CT 2-1-1 MASH SDM model in Figure 5.6. Figure 11.1 presents the implementation block diagram of the CT 2-1-1 MASH $\Sigma\Delta$ ADC. 4 active RC integrators compose the 2-1-1 MASH loop filter, with inverter-based amplifiers. Integration capacitors C₁ – C₄ are tunable with a {-40%, +120%} range and 1.25% step. The tunable range is up to +120% such that the designed ADC sampling rate is tunable between 3.5 GHz and 7 GHz. R_{b2}, R_{b4}, and R_{b5} implement signal feed-



Figure 11.2: (a) A differential-pair-based amplifier. (b) A pseudo-differential inverter-based amplifier.

forward paths b_2 , b_4 and b_5 . The signal feedforward path b_3 is implemented by a Gm cell to the ELD summation node of the 1st loop. $R_{k1} - R_{k5}$, DAC 5 — DAC 7 and DAC 11 – DAC 12 feed the quantization errors into the subsequent stages. The ELD-compensated 3-bit quantizer in the 1st stage has 2 Gm cells and 1 ELD DAC in every 1-bit slice, and those in the 2nd and 3rd stages have 1 Gm cell and 2 ELD DACs in every 1-bit slice. DAC 1 – DAC 7 are referred to as main DACs, and DAC 8 – DAC 12 are referred to as ELD DACs. The noise cancellation filters (NCF₁ – NCF₃) are implemented off-chip as 10-tap FIR filters. The mismatch between the analog filter and NCF is calibrated off-chip with 10-bit coefficient accuracy, similarly to the digital calibration shown in Chapter 6.

11.1.2 Loop filter and amplifiers

The 2-1-1 MASH loop filter needs 4 amplifiers, as shown in Figure 11.1. The amplifier can be implemented either as a differential-pair-based amplifier [30] or an inverterbased amplifier [9, 102, 103, 104]. At least two dimensions for classification can be applied here: (1) differential vs. pseudo-differential; (2) current-biasing vs. voltagebiasing. Although an inverter-based amplifier can be implemented as differential and with current-biasing as well [103], the voltage headroom of the PMOS and NMOS transistors which contribute to the transconductance g_m of the amplifier is reduced compared to the pseudo-differential inverter-based amplifier with voltage-biasing [9]. A differential-pair-based amplifier with current-biasing [30] and a pseudo-differential inverter-based amplifier with voltage-biasing [9] are compared, since these two implementations are the most promising candidates as the loop filter amplifier considered in this thesis. Simplified transistor-level implementations of a 1-stage differential-pairbased amplifier with current-biasing and a 1-stage pseudo-differential inverter-based amplifier with voltage-biasing are shown in Figure 11.2a and 11.2b, respectively. Their advantages and disadvantages are compared as follows:

- Voltage headroom: In a differential-pair-based amplifier (Figure 11.2a), at least 3 transistors are stacked between the supply, if the load R_L is implemented as a PMOS transistor. If R_L is implemented as a resistor, 2 transistors and a resistor are stacked between the supply. In an inverter-based amplifier (Figure 11.2b), only 2 transistors are stacked between the supply. Thus, in a pseudo-differential inverter-based amplifier, the transistors have more voltage headroom and higher current density. This benefit of the inverter-based amplifier is more obvious when a more advanced technology node with a lower supply is considered.
- Transconductance g_m : In the differential-pair-based amplifier, only the transconductance g_m of the NMOS transistor contributes to the gain of the amplifier. In the inverter-based amplifier, the transconductance g_m of both of the NMOS transistor and the PMOS transistor contributes to the gain. With an inverter-based amplifier, a higher total g_m can be made with the same current, compared to the differential-pair-based amplifier.
- Common-mode stability: The common-mode voltage of an amplifier should be stabilized around the designed value, typically through a common-mode feedback (CMFB) loop. Since the differential-pair-based amplifier has a differential structure, it has the advantage of an easy common-mode control. The inverter-based amplifier (in Figure 11.2b) has a pseudo differential structure. The stability of the CMFB loop in an inverter-based amplifier is typically more sensitive to PVT variation and device mismatch.
- Input capacitance: In the differential-pair-based amplifier, only the gate capacitance of the NMOS transistor $C_{g,n}$ contributes to the input parasitic capacitance. In the inverter-based amplifier, the gate capacitance of both of the NMOS transistor $C_{g,n}$ and the PMOS transistor $C_{g,p}$ contributes to the total input parasitic capacitance. The ratio between the width of the PMOS transistor W_p and the width of the NMOS transistor W_n (considering they have the same length) should be optimized for speed and common-mode voltage, which depends on the CMOS technology as well. In the inverter-based amplifier designed in this thesis in 40 nm CMOS technology, the ratio W_p/W_n is optimized to be about 2.
- Output capacitance: For a differential-pair-based amplifier, the drain capacitance of the NMOS transistor $C_{d,n}$ contributes to the output capacitance. If R_L is implemented as a PMOS transistor, the drain capacitance of this PMOS transistor $C_{d,p}$ contributes to the output capacitance as well. Otherwise if R_L is implemented as a resistor, the parasitic capacitance of the resistor contributes to the output capacitance of the amplifier. For an inverter-based amplifier, the drain capacitance of both of the NMOS transistor $C_{d,n}$ and the PMOS transistor $C_{d,p}$ contributes to the output capacitance. Typically, W_p/W_n in the inverter-based amplifier is larger than W_p/W_n of the differential-pair-based amplifier (when R_L is implemented as a PMOS transistor).

- Output swing: Typically a differential-pair-based amplifier has a smaller output swing than an inverter-based amplifier. In some literature, the inverter-based amplifier is claimed to have an almost rail-to-rail output swing [105]. However, when the output swing of the inverter-based amplifier is too large (e.g. larger than $V_{dd}/2$), the non-linearity of the inverter-based amplifier can become dominant in the total non-linearity of the $\Sigma\Delta$ ADC, and limits the linearity of the ADC. This thesis optimizes the differential output swing of an inverter-based amplifier to be about ± 0.5 V (with a 1.1 V supply).
- Linearity: The source degeneration technique can be used to improve the linearity for both of the differential-pair-based amplifier and the inverter-based amplifier [106]. In the context of the loop filter amplifier in this thesis, the amplifier is used in RC integrators (in a close-loop configuration), and its input swing is very small (about ±20 mV) since the input swing equals to the output swing suppressed by the gain of the amplifier. Together with an optimized output swing discussed in the previous point, the work of [9] demonstrates better than 100 dB linearity (THD) with inverter-based amplifiers for a signal BW of 25 MHz.
- Thermal noise: This thesis gives a quantitative comparison on the relationship between the total current I_{tot} , transconductance of the amplifier G_m and the square of thermal noise current i_{tn}^2 between the differential-pair-based amplifier and the inverter-based amplifier, used in an RC integrator in a broadband CT $\Sigma\Delta$ ADC. The RC integrator has a certain DC gain and 2^{nd} pole requirement (see Chapter 6). The amplifier can be modeled as its transconductance G_m , its output capacitance C_{out} , and its output resistance R_{out} . Here C_{out} is the total capacitance at the output node of the amplifier, which is typically not dominated by the parasitic output capacitance of the amplifier itself, but dominated by the load of the amplifier (e.g. parasitic capacitance of the integration capacitor, routing capacitance and input parasitic capacitance of the back-end stage). Thus, to achieve a certain 2^{nd} pole requirement, the amplifier has a fixed requirement on its transconductance G_m . It is discussed in the last point that the input swing of the amplifier is very small (about $\pm 20 \text{ mV}$). Typically in the context of a RC integrator in a broadband CT $\Sigma\Delta$ ADC, the transistors which contribute to G_m are working in the saturation region. In the literature, for a MOS transistor in saturation region, the ratio between the transconductance of a transistor g_m and its drain current I_d can be calculated as $\frac{g_m}{I_d} = \frac{2}{V_{GS} - V_t}$ [59]. It shows that $\frac{g_m}{I_d}$ depends on the biasing condition. It is reasonable to assume that with a proper optimization, $V_{GS} - V_t$ has a similar value among the NMOS M_1/M_2 in Figure 11.2a, and the NMOS M_2/M_4 and PMOS M_1/M_3 in Figure 11.2b. In this analysis, the ratio between g_m and I_d is assumed to be a constant $\frac{g_m}{I_d}$, which is valid for both the NMOS M₁/M₂ in Figure 11.2a, and the NMOS M_2/M_4 and PMOS M_1/M_3 in Figure 11.2b. In the differentialpair-based amplifier, $G_m = g_{m,N}$. Since M₁ uses half of the total current I_{tot} $(I_{tot} = I_s \text{ in Figure 11.2a})$, it can be calculated that the required total current

Table 11.1: Quantitative comparison between a differential-pair-based amplifier and an inverter-based amplifier on current and thermal noise for the same transconductance G_m

	I_{tot}	i_{tn}^2
Differential-pair-based amplifier	$\frac{2 \cdot G_m}{\frac{g_m}{I_d}}$	$4kT\gamma G_m$
Inverter-based amplifier	$\frac{G_m}{\frac{g_m}{I_d}}$	$4kT\gamma G_m$

is $I_{tot} = \frac{2 \cdot G_m}{\frac{G_m}{T_d}}$ under the aforementioned assumptions. In the inverter-based amplifier, $G_m = g_{m,N} + g_{m,P}$. It can be calculated that $I_{tot} = \frac{G_m}{\frac{g_m}{I_1}}$. Furthermore, it can be calculated that in both cases the input thermal noise current square $i_{tn}^2 = 4kT\gamma G_m$ [59]. This quantitative comparison is summarized in Table 11.1. It shows that to achieve the same amplifier G_m , the inverter-based amplifier needs $2 \times$ less current than the differential-pair based amplifier. It also shows that to achieve the same input-referred thermal noise current density, the inverter-based amplifier needs $2 \times$ less current than the differential-pair based amplifier. So the inverter-based amplifier has a clear advantage $(2\times)$ for both transconductance and thermal noise. Strictly speaking, the assumption that $\frac{g_m}{I_d}$ is the same constant for both PMOS and NMOS and in both differentialpair-based amplifier and inverter-based-amplifier is not very accurate. With a proper optimization, $V_{GS} - V_t$ can be slightly different among the NMOS M_1/M_2 in Figure 11.2a, and the NMOS M_2/M_4 and PMOS M_1/M_3 in Figure 11.2b. Considering this argument, this thesis concludes that the transconductance benefit and the thermal noise benefit of the inverter-based amplifier is about $2\times$, compared to the differential-pair-based amplifier.

• Power supply rejection: For the differential-pair-based amplifier in Figure 11.2a, the current source I_s offers a certain power supply rejection ratio (PSRR). For the inverter-based amplifier in Figure 11.2b, since it is voltage-biased, its power supply rejection is worse than the differential-pair-based amplifier. It is more sensitive to the noise on the supply. Thus, it is beneficial to use an on-chip LDO to regulate the supply for the inverter-based amplifier [9].

Based on the aforementioned comparison and considering the requirements of the loop filter amplifiers, a pseudo-differential multi-path Miller-compensated inverterbased amplifier is used for the loop filters, mainly because of its advantages regarding the voltage headroom, transconductance and noise. A source-degenerated differential-pair-based PMOS amplifier, is used to implement the Gm cell in the ELDcompensated 3-bit quantizers, for two reasons: (1) it offers high linearity with a large input swing; (2) its input and output common-mode voltages are more suitable for the application (discussed in Chapter 7). The circuit implementation of the multi-path Miller-compensated inverter-based amplifier is shown in Figure 11.3.



Figure 11.3: Inverter based amplifier.

The concept of the multi-path Miller-compensated structure proposed for a differentialpair-based amplifier in [107] is used in the design of the inverter-based amplifier. The differential path of the amplifier contains two paths. One path contains 3 inverters (inv 1 - inv 3), which is the high-gain low-speed path. Another path is composed of only one inverter (inv4), which is the high-speed low-gain path. A Miller capacitor C_M is inserted in parallel with inverter 3 (inv 3), to generate a dominant pole at the output node of inverter 2 (inv 2) [107]. A resistor R_z is inserted in series with C_M to generate a zero in the transfer function and to compensate the phase shift of the parasitic poles. Inverter 1, 2 and 4 use length of 80 nm, instead of the minimum length of 40 nm, to reduce their flicker noise and to achieve a higher g_m and gain. Inverter 3 uses the minimum length of 40 nm to achieve wide bandwidth, since its flicker noise is suppressed by the gain of inverter 1 and 2 when referred to the amplifier's input. All the inverters 1-6 have the same trip voltage, which is the input and output common-mode voltage of the amplifier. From AC noise simulation, the simulated flicker noise 3-dB-corner-frequency (when the thermal noise density equals to the flicker noise density) is about 16 MHz.

The circuit inside the dashed blue line is the common-mode feedback circuit [9]. R_{cmo} and C_{cmo} sense the output common-mode voltage. Inverters 5 and 6 make gain in the CMFB circuit. They use the minimum length of 40 nm for high-speed operation. Notice that the gain of the differential path contributes to the total gain of the CMFB loop as well. R_{cmi} and C_{cmi} form the actuator to correct the input common-mode voltage. In the CMFB circuit, C_{cmo} and C_{cmi} generate zeros in the transfer function of the CMFB loop, compensating the path shift caused by the parasitic poles. $C_{cm,p}$ is inserted in parallel with inverter 6, which creates a pole in the CMFB loop, reducing its bandwidth and increasing its gain margin. From post-layout simulation with the



Figure 11.4: Notch implementation can lead to a positive CMFB problem.

RC integrator layout, the CMFB loop has a worst-case phase margin of 70 degree, and a worst-case gain margin of 7 dB, over the considered process corners.

Chapter 5.1 mentions that the optimized zeros are not used on the ADC test chip because of implementation difficulties. Here, this limitation is described in detail. Figure 11.4 presents the considered circuit implementation of two RC integrators and a feedback path R_{fb} to realize the optimized zeros, with the inverter-based amplifiers shown in Figure 11.3. Some implementation details in Figure 11.3, such as R_z (in parallel with C_M), C_M and $C_{cm,p}$, are not shown here for simplicity. The implementation of the optimized zeros has two problems: (1) it generates a global positive CMFB loop; (2) it generates a "big latch". These two problems are elaborated in the following.

Since the OSR is very low (OSR = 7) in this design, after the system-level optimization shown in Chapter 5.3, the UGBW of the 1st integrator is only 0.48× signal BW. It means that the noise contributed by the 2nd integrator and DAC 2 in the frequency band (0.48×BW, BW) is amplified when referred to the ADC input. The notch frequency (result from the optimized zeros) is optimized to be 0.85× BW. Because of the low OSR, the notch frequency is higher referred to F_S compared to the case with high OSR. To implement the chosen SDM architecture and parameters, the ratio R_{in1}/R_{fb} is 0.97. Calculated from the SNDR and BW target and based on an optimized noise budgeting, the resistance values are chosen to be R_{in1} = 500 Ω , R_{in2} = 500 Ω , R_{fb} = 515 Ω .

The low resistance values of R_{in1} , R_{in2} and R_{fb} cause a problem in the common-mode stability of the circuit in Figure 11.4. First, the circuits in CMFB1 and CMFB2 are

for the moment ignored, and the reaction of the differential circuit on a common-mode disturbance is analyzed. Assume a common-mode disturbance $+\Delta V_{cm}$ at the SDM input nodes Vin+/-. Then, the 1^{st} integrator output nodes Vo1+/- see a commonmode disturbance $-G_1 \cdot \Delta V_{cm}$, where G_1 is the common-mode gain of the 1st RC integrator. A minus sign is in the formula since the differential path of the inverterbased amplifier has either 1 or 3 inverters. Similarly, the 2^{nd} integrator output nodes Vo2+/- see a common-mode disturbance $+G_1 \cdot G_2 \cdot \Delta V_{cm}$, where G_2 is the commonmode gain of the 2^{nd} RC integrator. The common-mode disturbance is fed back to the 1^{st} amplifier input nodes In1+/- with a value $+\frac{R_{in1}}{R_{in1}+R_{fb}}G_1 \cdot G_2 \cdot \Delta V_{cm}$. A global positive feedback loop for the common-mode disturbance is thus formed. The directions of the common-mode disturbances in this example are drawn as red arrows in Figure 11.4 for illustration purpose. Since $\frac{R_{in1}}{R_{in1}+R_{fb}}$ is 0.49 in this design, the global positive CMFB loop is very strong. The local negative CMFB loops through the CMFB circuits of the inverter-based amplifiers (CMFB1 and CMFB2 in the figure) must be stronger than the global positive feedback loop, to keep the common-mode voltage stable. It requires very low resistance value for the R_{cmi} in the local negative CMFB loops. However, when R_{cmi} is lower, it contributes more SDM input-referred noise. The PVT variation and device mismatch make the common-mode stability more fragile.

The notch implementation can cause another problem considering the transient behavior (large signal) and the start-up of the circuit. Figure 11.4 shows a big loop inv11 - inv12 - inv13 - R_{in2} - inv21 - inv22 - inv23 - R_{fb} - inv11 - inv12 - inv13 - R_{in2} - inv23 - R_{fb} . Since both R_{in2} and R_{fb} are very low, this loop tends to work as a "big latch" composed of 12 inverters. Actually there are more similar loops composed of 4 – 12 inverters. In the start-up of the chip, or because of a common-mode disturbance, or due to PVT variation and device mismatch, the nodes Vo1+/- and/or Vo2+/- can be trapped at Vdd or GND. In that case, because of the "big latch", it is very difficult to get out of this state.

Notice that the global positive CMFB loop exists in many designs as a result of the implementation of the notch (optimized zeros). The unique challenge faced in this design is caused by the low OSR, broad signal BW and high SNDR (or DR) target. The stability of the common-mode feedback loop for the inverter-based amplifier is more sensitive to PVT variation and device mismatch than the differential-pair-based amplifier because of its pseudo-differential structure. In this design, the optimized zeros is still high enough for the SNDR target.

Figure 11.5 shows the simulated feedback transfer function of the 1st RC integrator with its post-layout extracted model. The DC gain of the 1st integrator is 49.5 dB. At $F_S/2$ (3.5 GHz), the phase of the 1st RC integrator is -93°, which means -3° phase shift compared to the ideal transfer function. At around 5 GHz, some poles and zeros cause deviations of both magnitude and phase, which are due to the limited GBW of the inverter-based amplifier, routing resistance in the amplifier layout, and



Figure 11.5: AC simulation results on the feedback transfer function of the 1^{st} integrator: (a) magnitude (b) phase.

phase compensation techniques used in the amplifier (R_z in Figure 11.3) and in the integrator (R_{z1} in Figure 11.4). The frequency for which the phase drops to -135° (5 GHz in Figure 11.5) can be further pushed to a higher value at the cost of more power consumption, which is not needed for this design. In an open-loop testbench with the 1st inverter-based amplifier in post-layout extracted model (with load), the simulated GBW of the amplifier is about 10 GHz. The stability of the common-mode feedback loop is carefully verified for all four amplifiers. For the worst-case amplifier (in this design it is the amplifier in the 2nd integrator), over the considered process and temperature corners the worst-case phase margin of the CMFB loop is 70°, and the worst-case gain margin is 7.4 dB. All four inverter-based amplifiers in the loop filter have the same device size in the differential path for re-use purpose. The simulated power consumption of the 1st inverter-based amplifier is 18.4 mW.

11.1.3 Main DACs

In Chapter 4.3 three alternative circuit implementations of the main DAC unit element are shown in Figure 4.5. Their advantages and disadvantages (compared to the alternatives) are recapitulated here:

- The current-steering DAC with one-side switching (Figure 4.5a) has a few advantages [30]: (1) It requires simple DAC input logic to generate the switch control voltage $D^{+/-}$. (2) It can offer high output impedance, which means that it does not need an on-chip LDO to regulate its supply. (3) It does not require a negative supply. However, the main drawback of this implementation is that it has a noise penalty compared to the other two alternatives. The non-switching current sources $(0.5I_{DAC})$ do not contribute to the differential signal, but they contribute to the noise.
- The current-steering DAC with two-side switching (Figure 4.5b) has a few advantages [16, 63]: (1) It contributes less noise compared to the current-steering DAC with one-side switching. (2) It can offer high output impedance. However, it has some disadvantages as well: (1) It requires a more complicated DAC input logic to generate those two pairs of switch control voltages. (2) In some cases, it requires a negative supply [16].
- The resistive DAC (Figure 4.5c) has a few advantages [9]: (1) It has the best noise performance among all three alternatives, even if the noise on the supply (for the resistive DAC) and the noise on the DAC biasing circuit (for current-steering DACs) are included. (2) Since the matching of resistors is generally better than the matching of transistors, it is capable to achieve much better linearity [9]. (3) Its DAC input logic can be as simple as the current-steering DAC with one-side switching. However, its main drawback is that its output impedance (almost equal to R_{DAC}) is much lower than the current-steering DACs. Thus, a low-impedance low-noise wideband LDO must be integrated on-chip to provide its supply [9].

Since the resistive DAC requires a high-performance LDO to regulate its supply, which is not available in this design (and not in the scope of this thesis either), the resistive DAC was not chosen for this ADC test chip. Based on the aforementioned comparison between one-side switching and two-side switching, it is decided to implement the main DACs as current-steering DACs with one-side switching, with the unit DAC cell similar as in [30]. The schematic of a unit cell of the main DAC with a driver is presented in Figure 11.6.

In this design, the DAC driver is composed of an inverter and a pass-gate controlled by the DAC clock clkp/n_{DAC}. It has less delay than the DAC driver used in the work of [30], which is composed of a D-FF and two inverters, but it has less quantization gain as well. This DAC driver is designed for 7 GHz sampling rate, which means T_S is only 143 ps and the delay of the DAC driver must be minimized. Two alternative implementations of the data path are compared: (1) inverter – pass gate – DAC cell; (2) pass gate – inverter – DAC cell. The main drawback of (2) is that the mismatch of the inverters contributes to the different clock skew errors (timing errors) among the



Figure 11.6: Schematic of a unit cell of the main DAC with driver.

unit DAC slices. The advantage of (2) is that the DAC switch control signals V_{in} +/have a sharper edge than (1) with the same size of the inverter and pass gate. Based on the results of schematic Monte-Carlo (MC) simulations with device mismatch, (1) is chosen because of its better linearity.

In the layout design, the DAC switching core with cascode (M₂ - M₆), pass-gate (M₁₁ - M₁₂) and inverter (M₉ - M₁₀) are defined as the "DAC core" and it is placed between the quantizers and the loop filter (see Figure 11.14). R_{DAC} and M₁ compose a source-degenerated PMOS current source. M₇ and M₈ are NMOS current sources. Both source-degenerated PMOS current source and NMOS current source are placed a bit further away from the SDM's high-speed core, since they only offer static currents (see Figure 11.14). The detailed layout floorplanning is shown in Section 11.1.7. However, since the DAC core and DAC current source are not placed together, the routing between M₁ and M₂ is about 230 μ m long, which creates a parasitic capacitance of C_{rout} = 40 fF for DAC 1. The PMOS cascode transistor M₂ is designed in the deep saturation region and has a gain of 15 dB. It can suppress the effective parasitic capacitance seen by the V_t node caused by C_{rout}. However, since a large voltage headroom (238 mV) is occupied by M₂, the voltage drop on R_{DAC} is only 0.5 V. Reducing the voltage drop on R_{DAC} increases the DAC noise.

In the literature, common-centroid layout is often applied to the DAC current source which has critical matching requirement. However, in advanced CMOS technologies



Figure 11.7: Layout floorplan of the source-degenerated PMOS current source.

(such as 40 nm and 28 nm CMOS), the layout-dependent effects (LDE), including the well proximity effect, have a significant impact on the properties of the devices [108]. In this design, the source-degenerated PMOS current sources have stringent matching requirement, since their mismatch errors contribute to DAC and ADC nonlinearity. The layout floorplan of the source-degenerated PMOS current sources of DAC 1 is presented in Figure 11.7, considering both the common-centroid layout and LDE. Every unit current source I_{DAC} is divided into 4 equal pieces $\frac{1}{4}I_{DAC}$. Those 28 current source elements of $\frac{1}{4}I_{DAC}$ are placed into two rows. They are placed in a common-centroid way. The mismatch caused by LDE are canceled mutually at the first order. Dummy elements and dummy resistors are placed surrounding the active devices (in the current source element, the resistor R_{DAC} is placed close to the dummy resistor). The routing resistance and capacitance are equal among 7 unit current sources at the output nodes I < 1 >, ..., I < 7 >. Post-layout simulation with the LDE extracted shows that the systematic mismatch current among 7 unit current sources is only 23 nA, which is -83 dB referred to the nominal unit output current of 340 μ A.

11.1.4 Quantizers with ELD compensation

ELD-compensated 3-bit quantizers are designed for the ADC test chip with the techniques presented in Chapter 7.2, 7.3, 8.1 and 8.2. In the layout design of the current-mode locally-TI quantizer, a limitation caused by the coupling capacitance between the input/output nodes of two TI latches is detected. Figure 11.8 shows the layout of the TI-latches, samplers and buffers of Figure 8.6 ($M_7 - M_{30}$), in an intermediate phase during the design. The routing of the TI latches' input/output nodes Vo1^{+/-} and Vo2^{+/-} are highlighted in yellow, blue, white and red colors. They are crossed at the samplers (the design has a pair of main FB loop samplers and a pair of fast



Figure 11.8: Layout of TI comparators, samplers and buffers in an intermediate design phase.

Table 11.2: Extracted coupling capacitance between TI latches' input/output nodes

	$Vo2^+$	Vo2 ⁻
Vo1 ⁺	201 aF	$112 \mathrm{~aF}$
Vo1 ⁻	150 aF	166 aF

FB loop samplers as shown in Figure 8.12).

The coupling capacitance between the TI latches' input/output nodes from postlayout extraction is presented in Table 11.2. It shows that the coupling capacitance is not symmetrical. Thus, one latch output directly couples to the other latch's input. Although the unsymmetrical coupling capacitance seems very small (around 100 aF), its impact is significant. The reason is that from the loop filter output to the sampled initial condition of the comparator latch, the signal swing is attenuated by 18 dB (discussed in Chapter 8.2). Thus, the initial condition has a very small swing. On the other hand, since the latch output has a rail-to-rail swing, the error caused by the unsymmetrical coupling capacitance on the initial condition is significant. Another drawback of this layout is that the routing of Vo1^{+/-} and Vo2^{+/-} nodes is relatively long, which adds a relatively large routing capacitance and degrades the regeneration time constant of the latch.

Although the coupling capacitance between latch 1 and latch 2 can be reduced by revising the layout, it sacrifices the local floorplanning and it adds more routing capacitance to the sampler's output nodes. The solution used in this thesis is to add



Figure 11.9: Schematic of TI comparators, an additional buffer, and samplers.



Figure 11.10: Final layout of TI comparators, the additional buffer, samplers and buffers.

a buffer between the latches' input/output nodes and the samplers. A NMOS buffer with PMOS load is used to minimize the parasitic capacitance on the latch output nodes. The schematic of the TI comparators, the additional buffer and samplers is presented in Figure 11.9.

Figure 11.10 shows the final layout of the TI-latches, the additional buffer, samplers and buffers. It shows that the routing of the input/output nodes of latch 1 (highlighted in yellow and red color) is separated more from the routing of the input/output nodes of latch 2 (highlighted in green and magenta color), compared to Figure 11.8. Post-layout extraction and simulation show that the coupling capacitance between



Figure 11.11: Clock signals for the MASH SDM.

 $Vo1^{+/-}$ and $Vo2^{+/-}$ nodes is negligible.

11.1.5 Clock

The 2-1-1 MASH $\Sigma\Delta$ modulator needs a few clock signals. As shown in Chapter 8.1, the current-mode locally-time-interleaved quantizers need three pairs of clock signals – $\text{clkp}_{ltc}/\text{clkn}_{ltc}$, $\text{clkp}_{SH}/\text{clkn}_{SH}$, $\text{clkp}_{2SH}/\text{clkn}_{2SH}$. The clock phases clkn_{SH} and clkn_{2SH} are locally generated inside the quantizers from their opposite phases. Section 11.1.3 shows that the main DACs require clock signals $\text{clkp}_{dac}/\text{clkn}_{dac}$. Figure 11.11 presents an overview of the clock signals for the MASH SDM. In this figure, clkn_{dac} is not shown, which is the opposite phase of clkp_{dac} .

The TI latch clocks $clkp_{ltc}/clkn_{ltc}$ have a frequency of $F_S/2$ and a duty cycle of 50%. Quantizer sampling clocks $clkp_{1SH}$ and $clkp_{2SH}$ have a frequency of $F_S/2$ and a duty cycle of about 20%. The sampling clocks $clkp_{1SH}$ and $clkp_{2SH}$ become low ~15 ps earlier than the starting of the reset phase of the corresponding latch. The DAC clock $clkp_{dac}/clkn_{dac}$ have a frequency of F_S and a duty cycle of 50%. The rising edge of $clkp_{dac}$ is ~15 ps earlier than the falling edge of $clkp_{ltc}/clkn_{ltc}$, which equals to the DAC clock to DAC output (current) delay.

Figure 11.12 shows part of the clock circuitry on the 2-1-1 MASH $\Sigma\Delta$ modulator test chip. It shows the clock distribution and phase generation for the clock signals required by the main DACs and the quantizers. The clock signals needed by the demux, the calibration logic for the comparator offset calibration, the digital top-level control and memory are not shown in this figure; they are generated with some clock dividers and buffers. The clock circuitry shown in Figure 11.12 can be divided into two groups – a "clock receiver" and a "clock buffer". The clock receiver is designed to receive a 3.5 GHz – 7 GHz differential sine wave generated off-chip, and translate it to a close-to-square-wave differential clock signal. The clock input signals clkp_{in}/clkn_{in} are AC coupled to the 1st inverter's input through a capacitor C_{AC}. At



Figure 11.12: Clock circuitry.

the input of the 1^{st} inverter, the common-mode voltage is determined by the inverter itself through a resistor R_{cm} . Cross-coupled inverters (weaker than the 1^{st} and 2^{nd} inverter) are added at the 1^{st} and 2^{nd} inverter's outputs for clock-skew correction. At the output nodes of the clock receiver, the rising and falling edges (10% to 90%) are ~40 ps from post-layout simulation.

In the "clock buffer" the circuits for the generation and distribution of DAC clocks and quantizer clocks are separated. The DAC clock path only contains a few buffers, such that low clock jitter is achieved. The quantizer clock path contains a tunable delay block, a phase generator, and some buffers. With the tunable delay block, the relative delay between the DAC clocks and the quantizer clocks can be tuned in a range of $0 \sim 2 \times T_S$ with a step of 5 ps (for $F_S = 7$ GHz, simulated at TT corner, 65° C). The phase generator generates four clock signals/phases – clkp_{ltc}, clkn_{ltc}, clkp1_{SH} and clkp2_{SH}. The buffers drive three quantizers as load. From post-layout simulations, the rising/falling edges (10% to 90%) of the clock signals seen by the main DACs and quantizers are about 20 ps.

11.1.6 Measurement-assistance blocks

The MASH SDM test chip has digital outputs from three SDM stages (D₁, D₂, D₃ in Figure 11.1). Totally, 21 bit of unary-coded output at F_S is generated. It is very challenging to capture and synchronize the output data at F_S (7 GHz) directly in the measurement. To facilitate the data capture, the MASH SDM output is demultiplexed $32\times$ and stored in an on-chip memory. Figure 11.13 shows a block diagram of the MASH SDM, demux and on-chip memory. Data rates are indicated in the figure for $F_S = 7$ GHz. The main DAC switch control signals (V_{in} +/- in Figure 11.6) are used as the SDM digital output signals. They are buffered and then subsequently driving a local demux (×4). After the local demux, the frequency of the data is reduced to 1.75 GHz (F_S /4), and it can be routed for a few hundred μ m to the remote demux (×8).



Figure 11.13: Block diagram of SDM, demux and on-chip memory.

After the remote demux, the frequency of the data is further reduced to 218.75 MHz $(F_S/32)$. At this frequency, the 7-bit unary-coded quantizer output is converted to a 3-bit binary-coded format, to save memory space. This data is saved in an on-chip memory (1.5 Mbit). In the measurement, the data in the memory is read out via an I2C interface at 27.3 MHz $(F_S/256)$.

11.1.7 Layout floorplanning

Layout floorplanning is critical for broadband GHz-sampling ADCs. Figure 11.14 shows the micrograph of the ADC test chip. The digital blocks and analog blocks are separated in the floorplanning. The digital blocks include the digital control and memory block, and the quantizer calibration block. The remaining blocks are analog (or mixed-signal) blocks.

The key strategy in this layout design is to make the ADC high-speed core layout as compact as possible. The ADC high-speed core is defined as the loop filter, the ELD-compensated quantizer cores, the main DAC cores, local clock buffers, and the local demux.

The ELD-compensated quantizer's high-speed core contains the high-speed circuitry of the quantizer, including local clock buffers, Gm cells, ELD summation nodes, reference DAC core, ELD DAC core, TI latches, samplers and buffers, coarse calibration circuit, calibration DAC core, coarse calibration control circuit, etc. The ELD DAC, reference DAC and calibration DAC are implemented similarly (see Figure 8.6 and 8.12). The current sources $I_{dac,u}$, $I_{ref,u}$ and $I_{cali,u}$ are implemented as cascaded NMOS current sources. For the ELD DAC, reference DAC and calibration DAC,



Figure 11.14: Chip micrograph.

their switches and NMOS cascode transistors are defined as their high-speed core (DAC core), e.g. M_3 , M_4 and M_{31} for the ELD DAC in Figure 8.6. The NMOS current source transistors (e.g. M_{32} in Figure 8.6) are placed further away in the layout floorplan, since they do not have high-speed switching. They are matching critical and they have large gate area.

As shown in Figure 11.14, the quantizer's high-speed core is placed close to the main DAC cores and the loop filter. The other auxiliary circuitry in the quantizers, e.g. the fine calibration control circuit in Figure 8.12, is placed between the quantizers' high-speed core and the quantizer calibration.

The static circuitry, including the main DACs' current source and the quantizers' current source, is placed around the ADC's high-speed core. As shown in Figure 11.14, the ADC's high-speed core has an area of about 400 μ m × 400 μ m.

The ADC architecture presented in Figure 11.1 shows an additional challenge in the floorplanning of this design – it has two additional ELD DACs (DAC 11 and DAC 12) compared to other MASH architectures, e.g. the work of [16]. Figure 11.15 shows the local floorplanning of the 1^{st} and 2^{nd} quantizer core and the ELD DAC cores (DAC 8, DAC 9 and DAC 11) in an intermediate design phase. The layout of the 1^{st} quantizer is divided into 7 slices, in the same way as done with its schematic. Every


Figure 11.15: Local floorplanning of the 1^{st} and 2^{nd} quantizer cores and the ELD DAC cores (DAC 8, DAC 9 and DAC 11) in an intermediate design phase. "7" indicates that the 1^{st} quantizer has a 7-bit unary-coded output. The actual implementation is differential.



Figure 11.16: Final floorplanning of quantizer cores, ELD DAC cores and main DAC cores. "7" indicates 7 single-ended analog routing lines. The actual implementation is differential.

slice contains the TI latches, an ELD summation node, the ELD DAC core, etc. The ELD DAC core unit cells, whose output currents are added to the summation node of the slice, are placed in that slice. The x-dimension of the 1st quantizer core is 105 μ m, and its y-dimension is 66 μ m. The routing of the 1st quantizer's output to DAC 8 and DAC 11 is shown explicitly (D^{+/-} nodes in Figure 8.6). Layout design shows that the total routing length of D^{+/-} nodes is about 420 μ m (for every bit), and the extracted routing capacitance is about 20 fF. The buffers between the fast FB loop sampler and D^{+/-} nodes cannot drive this extra load. Increasing the size of the buffers adds load to the latch, and increases its regeneration time constant.

This problem is solved by changing the layout floorplanning. The final floorplanning of the quantizer cores, ELD DAC cores and main DAC cores is presented in Figure 11.16. In the final floorplanning, the ELD DAC core unit cells who are driven by the 1-bit quantizer output of the slice, are placed in that slice. The routing of $D^{+/-}$ nodes is significantly reduced, and the routing capacitance is reduced from 20 fF to 4 fF from post-layout extraction. This is achieved at a cost that the routing length

of the ELD DAC output current is significantly increased. Since the routing of the ELD DAC output current is part of the routing of the ELD summation node, the routing capacitance on the summation node is increased by about 20 fF compared to the old layout floorplanning in Figure 11.15. Fortunately, the problems caused by the additional routing capacitance on the summation node can be overcome by the techniques proposed in Chapter 7.3. From Figure 11.15 to Figure 11.16, the trade-off is that the long routing is moved from the D^{+/-} nodes to the ELD DAC output current nodes. A large routing capacitance of about 20 fF is also moved accordingly. This routing capacitance is a load to a voltage signal D^{+/-} in Figure 11.15, and it becomes a load to a current signal of the ELD DAC output current in Figure 11.16. In Figure 11.16 the placement of the main DAC cores (DAC 1 – DAC 7) is presented as well.

11.1.8 Challenges in the layout design phase

In this thesis, a traditional design flow for the MASH SDM test chip design is followed:

- 1. $\Sigma\Delta$ ADC architecture design, system-level modeling and simulations in Matlab/Simulink.
- 2. Schematic design and verification. The transistor layout parasitic model ('presimu') provided by the foundry is used. TT corner 65°C is considered. Monte-Carlo simulations with local mismatch are performed.
- 3. Layout design and verification. Post-layout extraction is done with the option 'RC coupled'. TT corner 65° C is considered.
- 4. Chip measurement.

In the layout design phase, a lot of challenges were encountered, mainly due to the following reasons:

- 1. The parasitics estimated by the 'presimu' are too optimistic. Post-layout extraction of a layout of the transistors gives more parasitics than 'presimu'.
- 2. Routing capacitance, coupling capacitance, and routing resistance cause problems in high-speed loops.

To overcome those challenges in the layout design phase, both schematic design and layout design of multiple blocks have several iterations. The maximal applicable sampling rate is reduced from 8.4 GHz in schematic simulations, to 7 GHz in post-layout simulations. Since the circuit of multiple blocks has to be sized up $2\times$, $4\times$ or even more to achieve the target sampling rate and signal bandwidth, the power consumption of the SDM and clock increases by about $4\times$ from schematic simulations

(147 mW) to post-layout simulations (619 mW). This section gives a few examples to elaborate it in more details.

In this design, the ELD-compensated quantizers consume the majority of the total power (see Figure 11.23), since the ELD-compensation loop is implemented in the quantizers. In the schematic design phase, the current source used in the input Gm cell of the 1st quantizer has a value $I_S = 300 \ \mu A$ (see Figure 8.6). In the layout design phase, the total routing capacitance on the summation node is about 50 fF. To be able to drive this additional load, the current source of the Gm cell is increased by $4 \times$ to $I_S = 1.2 \ \text{mA}$.

In the layout design phase, the routing capacitance on the buffered 1^{st} quantizer output nodes $(D^{+/-}$ nodes in Figure 8.6) is about 15 fF in the main feedback loop and 28 fF in the fast feedback loop $(D^{+/-})$ nodes are separated for main feedback loop and fast feedback loop, see Figure 8.12). The additional load from routing capacitance can be easily solved by adding some buffers with increased driving ability in lower speed ADCs. However, in this design, the target sampling rate in the layout design phase is 7 GHz, which means $T_S \approx 143$ ps. One additional inverter in the data path from the quantizer to the main DAC adds about 15 ps delay in post-layout simulations. This design allows only three buffers (one NMOS buffer with PMOS load, two CMOS buffers) and two pass-gates between the comparator core and the main DAC core, without any room to insert even one additional buffer. Thus, to drive the additional routing capacitance, the buffers, pass-gates, comparator core, Gm cell, ELD DAC, reference DAC, etc. have to be sized up and re-optimized. As a final result of this re-optimization, the transistor size of the comparator core $(M_9 - M_{12})$ in Figure 8.6) is increased by about $4\times$, and the transistor size of the CMOS buffer driving $D^{+/-}$ nodes (M_{29} - M_{30} in Figure 8.6) is increased by about 6×. The simulated regeneration time constant of the comparator core in schematic design phase after optimization (in schematic simulation) is about $\tau = 5.5$ ps. The simulated regeneration time constant of the comparator core in layout design phase after optimization (in post-layout simulation) is about $\tau = 6.5$ ps. The regeneration time constant is degraded, since the self-loading of the overall ELD compensation loop (comparator core - sampler buffers - ELD DAC - comparator core) is dominant. Even if the transistor size of the comparator core would be doubled again, after re-optimization of the whole ELD compensation loop, the improvement on the regeneration time constant is negligible.

In the layout design phase, because of the additional device parasitic capacitance, routing capacitance, routing resistance (including supply and ground routing), and increased load (from quantizers), the GBW of the loop filter amplifier with load is reduced, and the phase shift of the loop filter transfer functions at $F_S/2$ increases. To keep the stability and resolution of the MASH SDM, the transistor size of the loop filter amplifiers (which are inverter-based amplifiers) is increased by about $2\times$. A similar trend is also observed in the main DACs and clock circuitry.

Based on the aforementioned elaboration, this thesis concludes that, to achieve the target of 500 MHz BW, the MASH ADC test chip finally needs to be designed very



Figure 11.17: Simulated output spectrum with MASH SDM and clock at schematic level (TT corner, 65°C, no mismatch, no noise).

close to the speed boundary of the chosen 40 nm CMOS technology. To achieve the target sampling rate and signal bandwidth, the power consumption increases not linearly, but exponentially. It can explain why the figure-of-merit of this design is worse than the designs whose signal BW and peak SNDR (or DR) performance are not bounded by technology performance (see Chapter 11.4).

11.2 Simulation results

This section presents transistor-level simulation results of the 2-1-1 MASH $\Sigma\Delta$ ADC test chip. First, schematic simulation results are presented. Afterwards, post-layout simulation results are shown.

11.2.1 Schematic simulation results

For the following schematic simulation results, the 2-1-1 MASH $\Sigma\Delta$ modulator (including loop filter, ELD-compensated quantizers, main DACs) and clock distribution circuit are modeled at schematic level. The transistor layout parasitic model ('presimu') provided by the foundry is used. TT corner 65°C is considered. In the schematic simulations, ADC sampling frequency is $F_S = 8.4$ GHz, OSR = 7, and signal BW = 600 MHz. Input signal is a sine wave with frequency $f_{in} = 190.75$ MHz, and amplitude $A_{in} = 0.7$ V.

Figure 11.17 shows the simulated output spectrum with the MASH SDM and clock



Figure 11.18: Monte-Carlo simulation results with MASH SDM and clock at schematic level, with device mismatch only on the main DACs: (a) SNDR, SNR and THD of 19 independent simulations; (b) typical output spectrum.

at schematic level, without device mismatch and noise. The simulated SQNR is 69 dB for a signal BW of 600 MHz.

Monte-Carlo simulation with MASH SDM and clock at schematic level, with device mismatch only on the main DACs, are performed. Figure 11.18a presents the simu-



Figure 11.19: Monte-Carlo simulation results with MASH SDM and clock at schematic level with noise, with device mismatch only on the main DACs: (a) SNDR, SNR and THD of 19 independent simulations; (b) typical output spectrum.

lated SNDR, SNR and THD of 19 independent MC runs after calibration of the NCF coefficients. The simulated SNDR = 64.2 ± 0.9 dB (mean $\pm \sigma$), SNR = 65.7 ± 1 dB (mean $\pm \sigma$), and THD = -69.9 ± 2.2 dB (mean $\pm \sigma$). Figure 11.18b shows one typical output spectrum from this MC simulation, where SNDR = 64.2 dB, SNR = 66.6 dB and THD = -67.8 dB.

	noise	and distortion power
Item	in $[V^2]$	referred to signal power of $A_{in} = 0.7$ V sine input
Quantization noise	3.08×10^{-8}	-69 dB
Thermal and flicker noise	5.11×10^{-8}	-66.8 dB
Clock-jitter-introduced error	6.7×10^{-9}	-75.6 dB
DAC-mismatch-introduced error	6.23×10^{-8}	-65.9 dB
Total in-band noise and distortion	1.51×10^{-7}	-62.1 dB

Table 11.3: Overview of noise and distortion contributions from schematic simulations

Monte-Carlo simulation with MASH SDM and clock at schematic level, with device mismatch only on the main DACs, are performed with noise as well. Figure 11.19a shows the simulated SNDR, SNR and THD of 19 independent MC runs after calibration of the NCF coefficients. The simulated SNDR = 62.3 ± 0.9 dB (mean $\pm \sigma$), SNR = 63 ± 0.9 dB (mean $\pm \sigma$), and THD = -71.2 ± 2.5 dB (mean $\pm \sigma$). Figure 11.19b shows one typical output spectrum from this MC simulation, where SNDR = 62.3 dB, SNR = 63.3 dB and THD = -69.3 dB.

An overview of the noise and distortion contributions from schematic simulations with an $A_{in} = 0.7$ V sine input is presented in Table 11.3. A peak SNDR of 62.1 dB is expected for a signal BW of 600 MHz ($F_S = 8.4$ GHz). The peak SNDR is mainly limited by the DAC mismatch and transient noise (due to thermal noise and flicker noise).

11.2.2 Post-layout simulation results

For the following post-layout simulation results, the 2-1-1 MASH $\Sigma\Delta$ modulator and clock distribution circuit are post-layout extracted models (RC coupled). TT corner 65°C is considered. In the post-layout simulations, ADC sampling frequency is $F_S = 7$ GHz, OSR = 7, and signal BW = 500 MHz. The sampling frequency has to be reduced from 8.4 GHz to 7 GHz, otherwise the main feedback loop cannot be closed in 1 T_S . Input signal is a sine wave with frequency $f_{in} = 158.95$ MHz, and amplitude $A_{in} = 0.7$ V.

Figure 11.20 shows the simulated output spectrum with the MASH SDM and clock post-layout extracted, without device mismatch and noise. The simulated SQNR is 60.1 dB for a signal BW of 500 MHz. Notice that the out-of-band NTF peaking in the blue spectrum is an artifact from the calibration of the NCF coefficients. In the red spectrum (before calibration of the NCF coefficients), the out-of-band NTF peaking



Figure 11.20: Simulated output spectrum with MASH SDM and clock post-layout extracted (TT corner, 65°C, no mismatch, no noise).

is much lower.

Assuming the transient noise, clock-jitter-introduced error and DAC-mismatch-introduced error are the same as in the schematic simulations, the expected SNDR with SDM and clock post-layout extracted is 58.3 dB for a signal BW of 500 MHz ($F_S = 7$ GHz). The simulated power consumption of MASH SDM and clock is 619 mW.

11.3 Test chip validation

The CT 2-1-1 MASH SDM test chip has been fabricated in a 40 nm CMOS technology. For thin-oxide transistors, only the sub-type of super-low-threshold-voltage and normal-threshold-voltage transistors are used. The total area of the SDM, clock circuitry and demux is 1.56 mm², in which the area of the SDM and clock buffer is 1.34 mm². This section presents the measurement results. Figure 11.21 shows a photo of the validation board with a test chip sample during the measurement.

A low-phase-noise signal generator (Keysight E8257D) is used to generate the required clock signal [109]. The calculated clock jitter (RMS value) for a 6 GHz output clock signal from this clock generator is about 29 fs. The simulated on-chip clock jitter contribution for the DAC clock is about 44 fs. Thus, the calculated total clock jitter for the DAC clock is about 53 fs, which is below the clock jitter specification of 100 fs.

The MASH SDM test chip has been measured with 4 GHz, 5 GHz and 6 GHz sampling rates. This chapter presents the measurement results of single-tone test with 4 GHz



Figure 11.21: Validation board with a test chip sample during the measurement.

and 6 GHz sampling rates, and the measurement results of two-tone test with 5 GHz sampling rate. The possible reasons why the maximal applicable sampling rate in the measurement is lower than the simulation is explained in Section 11.3.5.

11.3.1 Comparator offset calibration

The comparator offset calibration presented in Chapter 8.2 is implemented on chip. In the calibration mode, the main DAC input should be disconnected from the quantizer output. However, adding a switch in the data path from the quantizer to the main DAC as shown in Figure 8.12 increases the propagation delay from the quantizer to the main DAC. This switch is not implemented on chip. In the measurement, in the calibration mode, the supplies of the loop filter and the main DACs are OFF. The supplies of the quantizers, digital top-level control, and calibration logic are ON. The input common-mode voltage of the quantizer is generated by a diode-connected inverter (which is a replica of one inverter used in the inverter-based amplifier).

In the measurement, the following calibration procedure is used, which is defined by the digital top-level control:

1. Coarse calibration: latch $1 \rightarrow$ latch $2 \rightarrow$ latch $1 \rightarrow$ latch $2 \rightarrow$ latch 1.



Figure 11.22: Histogram of the calibration codes in 1000 independent measurements: (a) coarse calibration code for latch 1 of the 4^{th} comparator in the 1^{st} quantizer; (b) fine calibration code for latch 1 of the 4^{th} comparator in the 1^{st} quantizer, when its coarse code = -3; (c) coarse calibration code for latch 2 of the 4^{th} comparator in the 1^{st} quantizer (d) fine calibration code for latch 2 of the 4^{th} comparator in the 1^{st} quantizer, when its coarse code = 1.

2. Fine calibration: latch $1 \rightarrow$ latch $2 \rightarrow$ latch $1 \rightarrow$ latch $2 \rightarrow$ latch 1.

In the measurement, it is found that the calibration codes are not always the same when the calibration is repeated (for the same test chip sample, the same temperature and voltage). For the same test chip sample, 1000 independent comparator offset calibrations are performed (each time the test chip is reset at the beginning). The histograms of the coarse and fine calibration codes for latch 1 and latch 2 of the 4^{th} comparator in the 1^{st} quantizer are presented in Figure 11.22, which is a typical case among the comparators of three quantizers. Figure 11.22a and 11.22c show that even the coarse calibration codes have a spread over 3 codes (the spread is 2 codes for some of the other latches). Typically one coarse calibration code is dominant,



Figure 11.23: Power breakdown when $F_S = 6$ GHz.

e.g. the dominant coarse code is -3 in Figure 11.22a. Figure 11.22b and 11.22d show the histograms of the fine calibration codes, when their coarse calibration code is the dominant code. With the same coarse calibration code, the fine calibration code typically has a spread of $6 \sim 7$ codes. The distribution of the fine calibration code is close to a Gaussian distribution, with one or two dominant codes. Thermal noise causes the spread of coarse and fine calibration codes. Generally, the performance of the $\Sigma\Delta$ ADCs is not sensitive to the thermal noise of the quantizer, since its thermal noise power is typically much lower than the quantization noise power. For this reason, the quantizer thermal noise is relatively large compared to the fine calibration step, and it directly influences the calibration code found by the calibration procedure. The error in the comparator offset calibration caused by the thermal noise can be averaged out by performing a large amount of independent calibrations, and choosing the calibration codes (the combination of a coarse code and a fine code) which appear most frequently. Since redundancy is built in between the fine calibration range and the coarse calibration step, even if the coarse calibration code has a +1 or -1 error due to thermal noise, the fine calibration range is large enough to find a correct fine calibration code to compensate the error.

11.3.2 Power consumption

The MASH SDM uses 1.1 V supply, except for DAC 1 ~ DAC 7 and quantizer input Gm cells that use 1.8 V supply. The power consumption of the $\Sigma\Delta$ modulator and the clock buffer is 574 mW when operating at $F_S = 6$ GHz, and 506 mW when operating at $F_S = 4$ GHz. The clock receiver has a power consumption of 17 mW when operating at $F_S = 6$ GHz, and 11 mW when operating at $F_S = 4$ GHz. The power and area of the clock receiver is excluded in the total SDM power and area in the benchmarking (Section 11.4), since it is not required when the $\Sigma\Delta$ ADC is

integrated in a receiver/transceiver product (the ADC clock is typically generated by an on-chip phase-locked loop (PLL) in a receiver/transceiver (SoC) product).

Figure 11.23 shows the power breakdown of the $\Sigma\Delta$ modulator and the clock buffer when $F_S = 6$ GHz. Since the ELD compensation is implemented in the quantizers, three quantizers consume 62% of the total power of the SDM and clock.

11.3.3 Single-tone test

The ADC has been measured with 4 GHz and 6 GHz sampling rates. Figure 11.24 presents the measured SNDR/SNR versus the input level. At 4 GHz sampling rate the measured DR and peak SNDR of the 2-1-1 MASH SDM in 300 MHz bandwidth are 65 dB and 60 dB respectively. If the output of the 3^{rd} stage is not used, the 2-1 MASH SDM has a peak SNDR of 55 dB. The 3^{rd} stage increases the peak SNDR by 5 dB, demonstrating the need of the 3^{rd} stage. At 6 GHz sampling rate the DR and peak SNDR of the 2-1-1 MASH SDM in 500 MHz bandwidth are 58 dB and 54 dB respectively.

Figure 11.25 shows the output spectra with 4 GHz and 6 GHz sampling rates before and after the comparator offset calibration. Before the comparator offset calibration, the 2^{nd} and 3^{rd} loops are partially overloaded, since the comparator offset of the proceeding loops is amplified by the inter-stage gain and causes a large offset error on the 2^{nd} and 3^{rd} loop's input signals. The comparator offset calibration improves the SNDR by 17 dB, and it improves the SFDR by 18 dB in 300 MHz BW at 4 GHz sampling rate. When the MASH SDM is sampled at 6 GHz, the comparator offset calibration improves the SNDR by 24 dB, and it improves the SFDR by 22 dB in 500 MHz BW. From the out-of-band noise it can be observed that the modulator is very stable and no NTF peaking is observed, thanks to the current-mode ELD compensation. When F_S is increased from 4 GHz to 6 GHz, the in-band noise floor increases by 5 dB due to increased metastability errors.

11.3.4 Two-tone test

A two-tone test has been performed with 5 GHz sampling rate. Two input tones of $A_{in} = -9$ dBFS at $f_{in1} = 255$ MHz and $f_{in2} = 305$ MHz are applied. The choice of f_{in1} and f_{in2} is limited by the available filters for the measurement. The measured spectrum (2¹⁷-point) is presented in Figure 11.26. For a signal BW of 375 MHz, the measured 2^{nd} order inter-modulation-distortion (IMD₂) is -70.1 dBc, and the 3^{rd} order inter-modulation-distortion (IMD₃) is -68.2 dBc.



Figure 11.24: SNDR/SNR vs. input level for (a) $F_S = 4$ GHz, BW = 300 MHz ($f_{in} = 80$ MHz); (b) $F_S = 6$ GHz, BW = 500 MHz ($f_{in} = 140$ MHz). 0 dBFS = 0.75 V_{peak} .

11.3.5 Measurement results compared to simulation results

This MASH $\Sigma\Delta$ ADC is designed to maximize the signal BW and sampling rate for the chosen 40 nm CMOS technology, for the peak SNDR and DR target given in Chapter 2.2. Post-layout simulation (TT corner, 65 °C, RC coupled post-layout extraction, no device mismatch, no transient noise) shows that a peak SNDR of 60.1 dB is achieved for signal BW of 500 MHz with $F_S = 7$ GHz. In the measurement,



Figure 11.25: Measured spectra (16384-point) before and after the comparator offset calibration: (a) with an $A_{in} = -1.5$ dBFS and $f_{in} = 80$ MHz input and $F_S = 4$ GHz in a linear frequency axis; (b) with an $A_{in} = -1.5$ dBFS and $f_{in} = 80$ MHz input and $F_S = 4$ GHz in a logarithmic frequency axis; (c) with a $A_{in} = -1.8$ dBFS and $f_{in} = 140$ MHz input and $F_S = 6$ GHz in a linear frequency axis; (d) with a $A_{in} = -1.8$ dBFS and $f_{in} = -$

the peak SNDR and DR of the ADC with $F_S = 6$ GHz are degraded compared to the post-layout simulation results because of increased metastability errors. Without input signal, when the sampling rate is increased to 6.5 GHz, the noise shaping of the SDM is still visible, although the noise floor further increases. The $\Sigma\Delta$ modulator is not stable when the sampling rate is 7 GHz in the measurement (no noise shaping observed).

An overview on the comparison among the schematic simulation, post-layout simulation and measurement results is presented in Table 11.4. From schematic simulations, the achievable sampling rate is $F_S = 8.4$ GHz. A peak SNDR of 62.1 dB is expected



Figure 11.26: Measured spectra (2¹⁷-point) with two input tones of $A_{in} = -9$ dBFS at $f_{in1} = 255$ MHz and $f_{in2} = 305$ MHz, $F_S = 5$ GHz.

Items	Post-layout simulation	Measu	rement
F_S (GHz)	7	6	4
signal BW (MHz)	500	500	300
peak SNDR (dB)	58.3^{a}	54	60
DR (dB)	61.3^{a}	58	65
power consumption (mW)	619	574	506
FoM_W (fJ/conv. step)	921	1402	1032
$\overline{\text{FoM}_S}$ (dB)	150	147	153

Table 11.4: Comparison between measurement and simulation results

^a Including quantization noise, thermal noise and DAC mismatch.

for a signal BW of 600 MHz, and the total power consumption of SDM and clock is 147 mW. From post-layout simulations, the maximal achievable sampling rate is $F_S = 7$ GHz. A peak SNDR of 58.3 dB is expected for a signal BW of 500 MHz, and the total power consumption of SDM and clock is 619 mW. In the measurement, the maximal applicable sampling rate is $F_S = 6$ GHz. At $F_S = 6$ GHz, the metastability error power is much higher than the post-layout simulations.

Compared to the post-layout simulation results, the measurement results have degra-

dation. The possible reasons are:

- 1. Device mismatch. Device mismatch can reduce the quantization gain of the quantizer and the data path from the quantizer to the main DACs, and increase the metastability errors. Figure 8.6 and Figure 11.6 show that the data path from the comparator core to the DAC cell contains one NMOS buffer with PMOS load, two CMOS buffers, and two pass-gates. There are two separated single-ended paths, which have a poor common-mode rejection (common-mode rejection is sacrificed to achieve high speed). Device mismatch changes the threshold voltages of those buffers, which can cause more metastability errors.
- 2. Supply and ground degradation. In the measurement, a large IR drop is observed from the supply and ground potentials measured at the LDOs on the validation board, to the supply and ground potentials measured on-chip through the analog test bus. The IR drop from the supply and ground together is up to 157 mV. From post-layout simulations, the maximal on-chip IR drop from the supply and ground together is up to 70 mV. No on-chip LDOs are used in this test chip. Simulations with the bonding wires modeled as inductance and resistance show an increased in-band noise floor, since the main DACs' output currents are modulated by the data-dependent supply and ground bouncing. The degradation of the supply and ground reduces the quantization gain in the loop as well, and increases the metastability errors.
- 3. The timing (especially the quantizer regeneration time) cannot be accurately tuned in the measurement. The design centering for the timing alignment is done according to post-layout simulation results for $F_S = 7$ GHz. In the measurement, only the relative delay between the DAC clock signals and the quantizer clock signals can be tuned. The quantizer regeneration time cannot be tuned separately, except for changing the ADC sampling rate. Thus, the timing alignment and the quantizer regeneration time can be not as optimal as in the post-layout simulations, e.g. due to PVT variations and device mismatch.
- 4. It is possible that the device models for simulations and post-layout extraction are not accurate enough for the target performance.

11.4 Benchmarking

The measured performance is summarized and compared to the state-of-the-art CT $\Sigma\Delta$ ADCs with signal BW ≥ 100 MHz in Figure 11.27. The state-of-the-art broadband CT $\Sigma\Delta$ ADCs can be divided into two categories. The works of [16, 88, 34, 30] maximize signal BW and peak SNDR (or DR) for given technologies. To maximize the signal BW and the sampling rate, they use flash sub-ADCs as the quantizers. Their Schreier FoMs are between 156 dB and 159 dB. The works of [63, 27, 52] optimize for the power efficiency of the ADC. For their technologies (16 nm and 28 nm CMOS), their signal BW and peak SNDR (or DR) performances are not bounded by technology performance. Their Schreier FoMs are between 168 dB and 174 dB.

For the MASH ADC test chip at $F_S = 4$ GHz, the probability of the metastability error is very low, and its influence on the peak SNDR and DR is not dominant. The static current consumption is dominant in the overall current consumption. It is mainly due to the static current in the ELD-compensated quantizers, loop filter and main DACs. The major part of the static current consumption, which is in the ELD-compensated quantizers and the loop filter, is required to achieve the target sampling rate (7 GHz) in the chosen 40 nm CMOS technology. The static current consumption in the main DACs is needed to achieve low thermal noise. Since the static current does not scale with the sampling rate, the Schreier FoM of this ADC at $F_S = 4$ GHz (153 dB) is a few dB less than the other $\Sigma\Delta$ ADCs which maximize the signal BW and peak SNDR (or DR). The aim of this work is to maximize the signal BW and sampling rate in a given technology, which is achieved at the cost of the static current consumption in the building blocks. At $F_S = 6$ GHz, the Schreier FoM is degraded by the increased metastability errors.

This ADC achieves the broadest signal bandwidth of 500 MHz for CT $\Sigma\Delta$ ADCs, while being implemented in a very mature 40 nm CMOS node. This ADC achieves less peak SNDR, DR and power efficiency with a similar signal BW compared to the work of [16], which is implemented at a more advanced technology node. The reported signal bandwidth of single-loop $\Sigma\Delta$ ADCs is limited to about 160 MHz [63]. With MASH SDM architecture, the signal bandwidth is increased to 500 MHz by about $3\times$. This signal bandwidth extension is achieved at the cost of more design complexity, for example:

- A MASH SDM architecture requires an additional digital noise cancellation filter. Matching is required between the analog transfer functions and digital noise cancellation filter, otherwise noise leakage can degrade the resolution of the MASH $\Sigma\Delta$ ADCs [73]. Calibrations are typically required to achieve the matching requirements.
- MASH $\Sigma\Delta$ ADCs have more stringent requirements on the loop filter amplifier GBW, parasitic poles in the loop and coefficient accuracy. These non-idealities can cause signal leakage and noise leakage, which can overload the back-end stages and degrade the resolution of the MASH $\Sigma\Delta$ ADCs [110].

The measured performance is further compared to the state-of-the-art CT LP $\Sigma\Delta$ ADCs with $F_S \geq 1$ GHz in 40 - 45 nm CMOS in Figure 11.28 [88, 30, 111, 86, 112, 113, 31]. The signal BW (500 MHz) of this work is more than $3\times$ higher than any other reported CT $\Sigma\Delta$ ADCs in 40 - 45 nm CMOS. This work demonstrates the highest sampling frequency of 6 GHz for a 3-bit $\Sigma\Delta$ modulator in 40 nm CMOS.

	This	drow	[16] JSSC 16	[88] VLSIC 17	[34] JSSC 12	[30] JSSC 11	[63] ISSCC 16	[27] ISSCC 17	[52] ISSCC 19
		210	Dong et al.	Dayanik et al.	Shibata et al.	Bolatkale et al.	Wu et al.	Huang et al.	Wang et al.
Technology (nm)	4	0	28	40	65	45	16	16	28
Architecture	2-1-1	MASH	1-2 MASH	3 rd order	6 th order	3 rd order	4 th order	4 th order	4 th order
Quantizer resolution (bit)		~	4	3.7	4.1	4	5	7	7
(V) bbV	1.8/	1.1	1.8 / 1.0 / -1.0	1.95 / 1.1 / -0.55	2.5 / 1.0 / -2.5	1.8/1.1	1.5 / 1.4 / 0.8	1.5 / 1.35 / 1	1.5/1.1
Area (mm²)	1.5	56	1.4	0.45	5.5	0.9	0.155	0.217	0.019
Fs (GHz)	4	9	8	5	4	4	2.88	2.15	2
OSR	6.67	9	8.6	16	13.3	16	6	8.6	10
Bandwidth (MHz)	300	500	465	156	150	125	160	125	100
Peak SNDR (dB)	60	54	67	66.5	n.a.	65	65.3	71.9	72.6
DR (dB)	65	58	72	70	73	70	72	74.8	76.3
Noise spectral density (dBc/Hz)	-150	-145	-159	-152	-155	-151	-154	-156	-156
THD (dBc)	-73	-67	n.a.	n.a.	n.a.	-74	-71	-80	n.a.
SFDR (dB)	75	89	80	83	n.a.	76	n.a.	n.a.	84
Power (mW)	506	574	890	233	750	256	40	54	16.3
FoM _S (dB)	153	147	159	158	156	157	168	168	174
Categories			Maximize	signal BW and r	esolution		Optin	nize power effici	ency
$FoM_{S} = DR+10^{*}log10$	(BW/P)								

Figure 11.27: Comparison table with state-of-the-art CT $\Sigma\Delta$ ADCs with signal BW \geq 100 MHz.

	This	work	[88] VLSIC 17 Dayanik et al.	[30] JSSC 11 Bolatkale et al.	[111] JSSC 17 Briseno-Vidrios et al.	[86] ISSCC 12 Srinivasan et al.	[112] JSSC 17 Edward et al.	[113] JSSC 15 Xing et al.	[31] JSSC 16 Loeda et al.
Technology (nm)	4	0	40	45	40	45	40	40	40
Architecture	2-1-1	MASH	3 rd order	3 rd order	3 rd order	3 rd order	2-2 MASH	1-1 MASH	4 th order
Quantizer resolution (bit)		~	3.7	4	4	-	4	5	1
Vdd (V)	1.8 /	1.1	1.95 / 1.1 / -0.55	1.8/1.1	1.1	1.8 / 1.4	2.5 / 1.15 / 1.1	0.9	n.a.
Area (mm²)	1.5	56	0.45	0.9	0.072	0.49	0.265	0.017	0.0194
Fs (GHz)	4	9	5	7	3.2	6	1	1.6	2.4
OSR	6.67	9	16	16	21.3	50	9.9	20	30
Bandwidth (MHz)	300	500	156	125	75	60	50.3	40	40
Peak SNDR (dB)	60	54	66.5	65	65.5	60.6	74.4	59.5	60.9
DR (dB)	65	58	20	20	n.a.	n.a.	76.8	n.a.	67.8
Noise spectral density (dBc/Hz)	-150	-145	-152	-151	n.a.	n.a.	-154	n.a.	-144
THD (dBc)	-73	-67	n.a.	-74	n.a.	n.a.	n.a.	n.a.	n.a.
SFDR (dB)	75	68	83	76	n.a.	75	84	67.6	n.a.
Power (mW)	506	574	233	256	22.8	20	43	2.57	5.25
FoM _S (dB)	153	147	158	157	163	n.a.	167.5	n.a.	167
$FoM_{S} = DR+10*log10$	(BW/P)								

Figure 11.28: Comparison table with state-of-the-art CT LP $\Sigma\Delta$ ADCs with $F_S \ge 1$ GHz in 40 - 45 nm CMOS.



Figure 11.29: Benchmarking plot of quantizer resolution (bit) vs. sampling rate (F_S) for state-of-the-art CT $\Sigma\Delta$ ADCs with $F_S \geq 1$ GHz, fabricated in 14 – 65 nm CMOS.

Figure 11.29 shows a benchmarking plot of the quantizer resolution vs. sampling rate for state-of-the-art CT $\Sigma\Delta$ ADCs with $F_S \geq 1$ GHz, fabricated in 14 – 65 nm CMOS. The blue/green/red dashed lines show the state-of-the-art envelopes for published 28 nm / 40 - 45 nm / 65 nm ADCs (3 best designs are considered for the envelopes). From these lines it can be extracted that for the same number of bits, 28 nm CMOS achieves ~1.6 – 2× higher sampling rate compared to 40 nm thanks to the technology advantage. Based on the technology benchmark it is expected that the bandwidth of $\Sigma\Delta$ ADCs can be extended further into the GHz range for more advanced technology nodes like 28 nm CMOS and beyond, employing the techniques of this work.

11.5 Conclusions

This chapter presents the circuit-level implementation and validation results of a 6-GHz-sampling 500-MHz-BW 2-1-1 MASH CT $\Sigma\Delta$ ADC in 40 nm CMOS technology. A multi-path Miller-compensated inverter-based amplifier is designed as the loop filter amplifier for its high GBW and power efficiency. The layout floorplan of the current source of the main DACs is not only common-centroid, but it also averages out the errors caused by the layout-dependent effects and the routing resistance. A buffer is added between the comparator core and the samplers to reduce the coupling

capacitance between the time-interleaved comparators.

Measurement results show that the calibration codes from the on-chip comparator offset calibration have a close-to-Gaussian-distributed spread caused by the quantizer thermal noise, and this error can be averaged out by performing multiple independent calibrations. At 4 GHz sampling rate the measured DR and peak SNDR in 300 MHz bandwidth are 65 dB and 60 dB respectively. At 6 GHz sampling rate the DR and peak SNDR in 500 MHz bandwidth are 58 dB and 54 dB respectively. The MASH ADC test chip is designed very close to the speed boundary of the chosen 40 nm CMOS technology. To maximize the sampling rate and signal bandwidth, the figure-of-merit is sacrificed since the power consumption increases exponentially with the sampling rate in the chosen 40 nm CMOS technology. In this design, the maximal sampling rate of the MASH SDM is limited by the main feedback loop, which means T_S is bounded by the regeneration time needed to generate the required quantization gain, plus the propagation delay (from quantizer to main DACs) and DAC delay to close the loop.

The target signal bandwidth of 500 MHz is indeed achievable in the chosen 40 nm CMOS technology; however, the power consumption increases exponentially with the sampling rate to achieve this signal bandwidth. If the power efficiency is also considered, 40 nm CMOS is not the best choice for this target specification if a more advanced technology node like 28 nm CMOS is also available. If the MASH ADC is ported from 40 nm CMOS into 28 nm CMOS, a broader signal BW and/or better power efficiency is expected for the following reasons: (1) 28 nm CMOS has faster transistors than 40 nm CMOS. In this design, the comparator core $(M_9 - M_{12})$ and M_{17} - M_{20} in Figure 8.6) achieves a regeneration time constant of $\tau = 6.5$ ps and it consumes 6 mA in post-layout simulations. In a state-of-the-art broadband $CT \Sigma\Delta$ ADC designed in 28 nm CMOS technology, a comparator core achieves a regeneration time constant of $\tau = 5.5$ ps and it consumes 4 mA [16]. (2) Since 28 nm CMOS technology has better matching than 40 nm CMOS technology, the area of the matching-critical blocks can be reduced, which reduces the length of the interconnections. (3) Since the minimum length in 28 nm CMOS is smaller, the devices are shrunk compared to 40 nm CMOS. Thus, the length of the interconnection is shorter in 28 nm CMOS, which reduces the parasitics of the interconnection. Notice that for the same length, the parasitics of the interconnection in 28 nm CMOS are not smaller than in 40 nm CMOS. For the same type of metal layers, the routing resistance per square in 28 nm CMOS is larger than in 40 nm CMOS, since the metal layers are thinner. The routing capacitance per unit area (e.g. per 100 nm \times 100 nm) is also larger in 28 nm CMOS, since the distance (dielectric) between metal layers is smaller.

The MASH SDM test chip design and validation prove that MASH SDM architectures can achieve a broader signal bandwidth than single-loop SDM architectures, for the same peak SNDR (or DR) target in the same technology nodes, at the cost of more design complexity.

The effectiveness of current-mode multi-path ELD compensation, current-mode locallytime-interleaved quantizer, and comparator offset calibration are proven with measurements. Thanks to the proposed techniques, the highest sampling rate of 6 GHz is demonstrated for a 3-bit SDM in 40 nm CMOS, which enables the broadest BW of 500 MHz for CT $\Sigma\Delta$ ADCs. Based on the technology benchmark it is expected that the bandwidth of $\Sigma\Delta$ ADCs can be extended further into the GHz range for more advanced technology nodes like 28 nm CMOS and beyond, employing the techniques of this work.

Chapter 12

Conclusions and recommendations

This chapter presents the conclusions of this thesis and recommendations for future work. First, Section 12.1 draws conclusions of this thesis. Afterwards, Section 12.2 provides recommendations for future work in broadband $CT \Sigma \Delta$ ADCs.

12.1 Conclusions

In a given technology, for a given DR target, MASH $\Sigma\Delta$ ADCs can achieve broader signal BW than single-loop $\Sigma\Delta$ ADCs, at the cost of more complexity. Single-loop $\Sigma\Delta$ ADCs have a simpler architecture than MASH $\Sigma\Delta$ ADCs. However, the out-of-band gain of the overall noise transfer function of a MASH $\Sigma\Delta$ ADC can be higher than the OBG of a single-loop $\Sigma\Delta$ ADC, for the same NTF order and quantizer resolution. Moreover, for MASH $\Sigma\Delta$ ADCs with multi-bit quantization, an inter-stage gain of higher than 1 can be applied to further improve the signal to quantization noise ratio. So MASH $\Sigma\Delta$ ADCs can achieve a certain BW and dynamic range target with lower OSR than single-loop $\Sigma\Delta$ ADCs for the same quantizer resolution.

The generic formulas proposed in this thesis to calculate the SQNR limits for both single-loop and MASH SDMs reduce the complexity of system level design and comparison significantly. The formulas can be used to calculate the SQNR limits for any single-loop or MASH SDM architectures with any NTF order, quantizer resolution, and OSR, and any NTF order for the sub-stages for MASH structures.

The proposed digital calibration of the errors due to the integrators' limited DC gain and 2^{nd} pole for CT MASH $\Sigma\Delta$ ADCs can relax the DC gain requirements of the 1^{st} – 3^{rd} integrators by about 15 dB, and the 2^{nd} pole requirement of the 1^{st} integrator by $3\times$.

The proposed current-mode multi-path ELD compensation technique can solve the problem due to the parasitic delay caused by the poles of the summing amplifier

and the pre-amplifier for multi-GHz sampling multi-bit $\Sigma\Delta$ ADCs. The test chip measurement results prove its effectiveness.

The limitations caused by the parasitic capacitance on the ELD summation node can be overcome by adding a cross-coupled feedforward capacitor C_{ff} in parallel with the ELD DAC, and by creating a zero in the transfer function of the V/I converter (Gm cell). The test chip measurement results prove its effectiveness.

The proposed current-mode locally-time-interleaved multi-bit quantizer can move the reset time and the time to track the next initial condition of the comparator out of the 1 T_S ELD time budget, and enable $1.33 \times$ higher ADC sampling rate and signal BW. The test chip measurement results prove its effectiveness.

The proposed two-step coarse-fine comparator offset calibration can calibrate the offset of the locally-time-interleaved comparators and restore the ADC resolution. The test chip measurement results prove its effectiveness.

The proposed implementation of the DAC linearization techniques in broadband CT $\Sigma\Delta$ ADCs with the mapping engine in the digital domain can ensure that the nonidealities of the mapping engine do not interfere with the analog purity of the reference signal. It is a general-purpose architecture and it offers the flexibility to implement different static and dynamic DAC linearization techniques.

The metastability errors in broadband CT $\Sigma\Delta$ ADCs can be counteracted by the proposed two system-level solutions – the metastability error compensation technique, and the metastability shaping technique.

The 3-bit 2-1-1 MASH CT $\Sigma\Delta$ modulator test chip in 40 nm CMOS technology achieves 58 dB DR and 54 dB peak SNDR in 500 MHz bandwidth when sampled at 6 GHz and consumes 574 mW power. It achieves 65 dB DR and 60 dB peak SNDR in 300 MHz bandwidth when sampled at 4 GHz, while consuming 506 mW power. Thanks to the proposed techniques, the highest sampling rate of 6 GHz is demonstrated for a 3-bit SDM in 40 nm CMOS, which enables the broadest BW of 500 MHz for CT $\Sigma\Delta$ ADCs.

To achieve the target of 500 MHz BW, the MASH ADC test chip is designed very close to the speed boundary of the chosen 40 nm CMOS technology. The maximal achievable sampling rate and the signal bandwidth are limited by the total time needed by the main feedback loop, including the regeneration time to generate the required quantization gain, and the propagation delay and DAC delay to close the main feedback loop and to maintain the stability of the SDM. The main limitations for the performance (resolution, signal BW, power efficiency) of the test chip are metastability errors and layout parasitics (especially interconnections).

The main conclusions of the thesis are:

• In a given technology, for a given DR target and a maximum power boundary, the achievable signal bandwidth can be maximized by applying multi-stage

noise-shaping (MASH), multi-bit quantization, and by maximizing the sampling frequency.

• This thesis demonstrates the broadest signal BW of 500 MHz for CT $\Sigma\Delta$ ADCs, and the highest sampling rate of 6 GHz for 3-bit $\Sigma\Delta$ ADCs in 40 nm CMOS.

Based on the technology benchmark it is expected that the bandwidth of $\Sigma\Delta$ ADCs can be extended further toward the GHz range for more advanced technology nodes like 28 nm CMOS and beyond, employing the techniques of this thesis.

12.2 Recommendations for future work

From the design and validation of the MASH ADC test chip, some circuit-implementation limitations in the test chip were discovered. They are not fundamental limitations of the CT MASH $\Sigma\Delta$ ADC or the technology, and give rise to recommendations for future research. The limitations are first listed here and then followed by the recommendations:

- The approach of using multiple paths to extract the quantization noise of the preceding stages leads to challenges in the circuit implementation.
- The mismatch of the quantizer current sources is not calibrated, which leads to large area.
- No on-chip LDOs are used for the MASH ADC test chip, which might cause performance degradation.
- Too high signal leakage is observed in the measurement, which limits the peak SNDR and DR of the MASH $\Sigma\Delta$ ADC by a few dB.
- Not enough tunability is designed for the clock signals, and the regeneration time of the comparator core cannot be optimized in the measurement.
- Current-steering DACs with one-side switching are used as the main feedback DACs, which give noise penalty compared to resistive DACs.

Investigation on alternative approaches which can simplify the extraction of the quantization noise of the preceding stages is recommended for future research of broadband MASH $\Sigma\Delta$ ADCs. For the 2-1-1 MASH $\Sigma\Delta$ ADC architecture synthesized in this thesis, multiple paths ($k_1 \sim k_{10}$ in Figure 5.6) are used to extract the quantization noise of the preceding SDM stage as the input of a next SDM stage. This approach has two limitations in the circuit implementation (especially in the layout design phase): (1) matching is required among multiple paths for an accurate extraction of the quantization noise. Due to coefficients mismatch, limited GBW of loop filter amplifiers, parasitic poles in the loops, etc., the transfer functions of those paths deviate from the ideal transfer functions. One consequence is that it results in undesired signal and quantization noise leakage, which occupies signal swing in the back-end SDM stages. This effect causes the back-end stages to reach overload with a smaller maximal input signal power than the ideal MASH SDM, which reduces the peak SNDR and DR of the MASH $\Sigma\Delta$ ADC. The 2nd consequence is that the real transfer function of the analog loop filter does not match the digital noise cancellation filter, which causes noise leakage in the MASH output and requires calibration of the NCF coefficients. (2) The additional ELD DACs (DAC 11 and DAC 12 in Figure 11.1) add extra load to the quantizer and add complexity in the layout floorplanning.

Investigation on alternative comparator offset calibration techniques is recommended to reduce the total area of the SDM. The comparator offset calibration technique proposed in Chapter 8.2 does not calibrate the amplitude error of the current sources of the reference DAC, ELD DAC and calibration DAC. Since the accuracy of the current sources for the quantizers relies on intrinsic matching, the quantizer current sources occupy a relatively large area (see Figure 11.14). System-level simulations show that the reference DAC's current sources have a more stringent matching requirement than the ELD DAC's and calibration DAC's current sources. For future research in comparator offset calibration, it is recommended to include the calibration of the relevant current sources. Thus, the total SDM area can be reduced.

On-chip LDOs are recommended to generate all sensitive supplies. Better ground strategy is recommended as well. For this MASH $\Sigma\Delta$ ADC test chip, no on-chip LDOs are used, and all the supplies are provided from outside the chip (generated on PCB). Moreover, many supply and ground domains are separated on chip. This supply and ground strategy has two limitations: (1) it is sensitive to bonding wire inductance, which can cause supply and ground bouncing. (2) it leads to electrostatic discharge (ESD) risk. Cross-domain protection circuits are not used on this test chip since they add delay in the high-speed loops. For future work in the design of broadband $\Sigma\Delta$ ADCs (especially for products), this thesis gives the following suggestions:

- All of the performance sensitive supplies should be generated on chip with LDOs, including all of the 1.1 V and 1.8 V supplies for the loop filter, main DACs, quantizers, clock circuitry. Proper decoupling capacitors should be added on-chip to regulate the supplies. Thus, the performance of the ADC is less sensitive to bonding wire inductance.
- Do not split ground domains unless it is necessary. Loop filter and main DACs are recommended to share the same analog ground on chip.

For future work in the design of broadband MASH $\Sigma\Delta$ ADCs, it is suggested to add tunability for the signal component in the back-end stages. In this MASH $\Sigma\Delta$ ADC test chip, too high signal leakage is observed in the measurement (the signal power in the 2nd and 3rd stage is close to the signal power in the 1st stage), which limits the peak SNDR and DR (by roughly 2 ~ 3 dB). For future work in design of broadband $\Sigma\Delta$ ADCs, it is suggested to add extra tunability on the clock signals. E.g., if the regeneration time of the comparator core can be independently tuned, an optimal quantization gain can be achieved in the measurement [16]. However, this additional tuning (in this paragraph and the previous paragraph) adds extra costs, e.g. design complexity, power consumption, calibration/trimming time (which add cost to the products).

For future work, it is suggested to consider using a resistive DAC as the main DAC, for its low noise and high linearity. The supply (or reference voltage) for the resistive DAC should be generated on chip with a low-noise high-performance LDO [9].

This thesis has a suggestion on simulations/verifications for future work in design of broadband (MASH) $\Sigma\Delta$ ADCs. Monte-Carlo simulations with device mismatch should be performed with the post-layout extracted model of the $\Sigma\Delta$ modulator. PVT corner simulations and a combination of PVT corners and device mismatch should also be performed with the post-layout extracted model for product design. Thus, if the metastability error increases when device mismatch is enabled, or when PVT corner is considered, it will be captured in these simulations.

Moreover, for several novel concepts proposed in this thesis, either they are not implemented on the test chip, or the measurement results cannot prove their effectiveness. Further research is recommended on those techniques as well.

For the new implementation of the DAC linearization techniques proposed in Chapter 9, the registers (mapping engine) shown in Figure 9.6 are implemented on the MASH ADC test chip. In the measurement (both single-tone test and two-tone test), different permutations on the reference codes were tried. However, no distinguishable difference on the harmonic distortion (in single-tone test) or inter-modulation distortion (in two-tone test) were measured. Future research on the proposed implementation of the DAC linearization techniques is recommended, to show the effectiveness of this approach with measurement results.

In the measurement of the MASH $\Sigma\Delta$ ADC test chip, the peak SNDR and DR of the MASH ADC at 6 GHz sampling rate is mainly limited by the increased metastability error. It is a pity that the proposed system-level solutions to combat the metastability error - the metastability error compensation technique (Chapter 10.4), and the metastability shaping technique (Chapter 10.5) – were not implemented on the test chip. During this thesis, we have supervised two master student's projects to investigate the circuit implementation of the proposed metastability error compensation technique [101, 100]. The implementation of the analog delay and summation, and the matching requirement among DAC 1, DAC 3 and DAC 4 (in Figure 10.6) have been studied at transistor-level. Schematic simulations show that the MEC circuit is feasible to be implemented in modern CMOS technologies. Unfortunately, for this PhD work, no time was left to further investigate the circuit implementation of the metastability shaping technique. The comparator offset of quantizer 2 (in Figure 10.13) might be a concern due to its impact on the stability of the SDM, which

requests further investigation. For future research, it is definitely recommended to investigate the circuit implementation of these two techniques to combat the metastability errors, and to show the effectiveness of these two techniques with measurement results.

References

- [1] "Integrated, quad RF transceiver with observation path ADRV9026 data sheet," Analog devices, Tech. Rep., 2019. [Online]. Available: https://www.analog.com/media/radioverse-adrv9026/adrv9026.pdf
- [2] B. P. Ginsburg, K. Subburaj, S. Samala, K. Ramasubramanian, J. Singh, S. Bhatara, S. Murali, D. Breen, M. Moallem, K. Dandu, S. Jalan, N. Nayak, R. Sachdev, I. Prathapan, K. Bhatia, T. Davis, E. Seok, H. Parthasarathy, R. Chatterjee, V. Srinivasan, V. Giannini, A. Kumar, R. Kulak, S. Ram, P. Gupta, Z. Parkar, S. Bhardwaj, R. Y.C., R. K. A., A. Shrimali, and V. Rentala, "A multimode 76-to-81GHz automotive radar transceiver with autonomous monitoring," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2018, pp. 158–159.
- [3] D. Guermandi, Q. Shi, A. Dewilde, V. Derudder, U. Ahmad, A. Spagnolo, I. Ocket, A. Bourdoux, P. Wambacq, J. Craninckx, and W. V. Thillo, "A 79-GHz 2 × 2 MIMO PMCW radar SoC in 28-nm CMOS," *IEEE J. Solid-State Circuits*, vol. 52, no. 10, pp. 2613–2626, 2017.
- B. Murmann, "ADC performance survey 1997-2021," 2021. [Online]. Available: http://web.stanford.edu/~murmann/adcsurvey.html
- [5] G. E. Moore, "Cramming more components onto integrated circuits," *Electron*ics Magazine, vol. 38, no. 8, 1965.
- [6] 3GPP website, 2020. [Online]. Available: 3gpp.org
- [7] C. E. Shannon, "Communication in the presence of noise," Proceedings of the IEEE, vol. 86, no. 2, pp. 447–457, 1998.
- [8] E. A. Vittoz and Y. P. Tsividis, *Frequency-dynamic range-power*. Kluwer Academic Publishers, 2002.
- [9] L. Breems, M. Bolatkale, H. Brekelmans, S. Bajoria, J. Niehof, R. Rutten, B. Oude-Essink, F. Fritschij, J. Singh, and G. Lassche, "A 2.2 GHz continuoustime ΔΣ ADC with -102 dBc THD and 25 MHz bandwidth," *IEEE J. Solid-State Circuits*, vol. 51, no. 12, pp. 2906–2916, 2016.

- [10] C. Moreland, M. Elliott, F. Murden, J. Young, M. Hensley, and R. Stop, "A 14b 100MSample/s 3-stage A/D converter," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2000.
- [11] K. Gulati, C. Munoz, S. Cho, G. Manganaro, M. Lugin, M. Peng, A. Pulincherry, J. Li, A. Bugeja, A. Chandrakasan, and D. Shoemaker, "A highly integrated analog baseband transceiver featuring a 12-bit 180MSPS pipelined A/D converter for multi-channel wireless LAN," in *IEEE Symposium* on VLSI Circuits, 2004, pp. 428–431.
- [12] A. M. A. Ali, H. Dinc, P. Bhoraskar, C. Dillon, S. Puckett, B. Gray, C. Speir, J. Lanford, J. Brunsilius, P. R. Derounian, B. Jeffries, U. Mehta, M. McShea, and R. Stop, "A 14 bit 1 GS/s RF sampling pipelined ADC with background calibration," *IEEE J. Solid-State Circuits*, vol. 49, no. 12, pp. 2857–2867, 2014.
- [13] S. Devarajan, L. Singer, D. Kelly, S. Kosic, T. Pan, J. Silva, J. Brunsilius, D. Rey-Losada, F. Murden, C. Speir, J. Bray, E. Otte, N. Rakuljic, P. Brown, T. Weigandt, Q. Yu, D. Paterson, C. Petersen, and J. Gealow, "A 12b 10GS/s interleaved pipeline ADC in 28nm CMOS technology," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2017, pp. 288–289.
- [14] B. Hershberg, D. Dermit, B. van Liempd, E. Martens, N. Markulic, J. Lagos, and J. Craninckx, "A 3.2GS/s 10 ENOB 61mW ringamp ADC in 16nm with background monitoring of distortion," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2019, pp. 58–59.
- [15] Q. Gu, RF system design of transceivers for wireless communications. Springer, 2005.
- [16] Y. Dong, J. Zhao, W. Yang, T. Caldwell, H. Shibata, Z. Li, R. Schreier, Q. Meng, J. B. Silva, D. Paterson, and J. C. Gealow, "A 72 dB-DR 465 MHz-BW continuous-time 1-2 MASH ADC in 28 nm CMOS," *IEEE J. Solid-State Circuits*, vol. 51, no. 12, pp. 2917–2927, 2016.
- [17] M. Viot, "Automotive security: why UWB measures up," Embedded Systems Engineering, pp. 6–10, 2014.
- [18] R. Walden, "Analog-to-digital converter survey and analysis," *IEEE Journal on Selected Areas in Communications*, vol. 17, no. 4, pp. 539–550, 1999.
- [19] R. Schreier and G. C. Temes, Understanding Delta-Sigma data converters. IEEE Press Wiley, 2005.
- [20] D. C. Daly and A. P. Chandrakasan, "A 6b 0.2-to-0.9V highly digital flash ADC with comparator redundancy," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2008, pp. 554–555.

- [21] H. Shibata, V. Kozlov, Z. Ji, A. Ganesan, H. Zhu, D. Paterson, J. Zhao, S. Patil, and S. Pavan, "A 9-GS/s 1.125-GHz BW oversampling continuous-time pipeline ADC achieving -164-dBFS/Hz NSD," *IEEE J. Solid-State Circuits*, vol. 52, no. 12, pp. 3219–3234, 2017.
- [22] T.-C. Hung, J.-C. Wang, and T.-H. Kuo, "A calibration-free 71.7db SNDR 100MS/s 0.7mW weighted-averaging correlated level shifting pipelined SAR ADC with speed-enhancement scheme," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2020, pp. 256–257.
- [23] P. Harpe, E. Cantatore, and A. van Roermund, "A 2.2/2.7fJ/conversion-step 10/12b 40kS/s SAR ADC with data-driven noise reduction," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2013, pp. 270–271.
- [24] L. Kull, T. Toifl, M. Schmatz, P. A. Francese, C. Menolfi, M. Brändli, M. Kossel, T. Morf, T. M. Andersen, and Y. Leblebici, "A 3.1 mW 8b 1.2 GS/s singlechannel asynchronous SAR ADC with alternate comparators for enhanced speed in 32 nm digital SOI CMOS," *IEEE J. Solid-State Circuits*, vol. 48, no. 12, pp. 3049–3058, 2013.
- [25] E. Janssen, K. Doris, A. Zanikopoulos, A. Murroni, G. van der Weide, Y. Lin, L. Alvado, F. Darthenay, and Y. Fregeais, "An 11b 3.6GS/s time-interleaved SAR ADC in 65nm CMOS," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2013, pp. 464–465.
- [26] Y.-Z. Lin, C.-Y. Lin, S.-C. Tsou, C.-H. Tsai, and C.-H. Lu, "A 40MHz-BW 320MS/s passive noise-shaping SAR ADC with passive signal-residue summation in 14nm FinFET," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2019, pp. 330–331.
- [27] S.-J. Huang, N. Egan, D. Kesharwani, F. Opteynde, and M. Ashburn, "A 125MHz-BW 71.9dB-SNDR VCO-based CT ΔΣ ADC with segmented phasedomain ELD compensation in 16nm CMOS," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2017, pp. 470–471.
- [28] M. Steiner and N. Greer, "A 22.3b 1kHz 12.7mW switched-capacitor ΔΣ modulator with stacked split-steering amplifiers," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2016, pp. 284–285.
- [29] S. R. Norsworthy, R. Schreier, and G. C. Temes, *Delta-Sigma data converters: theory, design, and simulation*. IEEE Press, 1997.
- [30] M. Bolatkale, L. J. Breems, R. Rutten, and K. A. A. Makinwa, "A 4 GHz continuous-time ΔΣ ADC with 70 dB DR and -74 dBFS THD in 125 MHz BW," *IEEE J. Solid-State Circuits*, vol. 46, no. 12, pp. 2857–2868, 2011.
- [31] S. Loeda, J. Harrison, F. Pourchet, and A. Adams, "A 10/20/30/40 MHz feedforward FIR DAC continuous-time $\Delta\Sigma$ ADC with robust blocker performance

for radio receivers," *IEEE J. Solid-State Circuits*, vol. 51, no. 4, pp. 860–870, 2016.

- [32] P. Cenci, M. Bolatkale, R. Rutten, M. Ganzerli, G. Lassche, K. Makinwa, and L. Breems, "A 3.2mW SAR-assisted CTΔΣ ADC with 77.5dB SNDR and 40MHz BW in 28nm CMOS," in *IEEE Symposium on VLSI Circuits*, 2019, pp. C230–C231.
- [33] H. Hegt, A. van Roermund, and G. Radulov, "PERFECT SDM project description."
- [34] H. Shibata, R. Schreier, W. Yang, A. Shaikh, D. Paterson, T. C. Caldwell, D. Alldred, and P. W. Lai, "A DC-to-1 GHz tunable RF ΔΣ ADC achieving DR= 74 dB and BW= 150 MHz at f₀ = 450 MHz using 550 mW," *IEEE J. Solid-State Circuits*, vol. 47, no. 12, pp. 2888–2897, 2012.
- [35] M. Neofytou, M. Zhou, M. Bolatkale, Q. Liu, C. Zhang, G. Radulov, P. Baltus, and L. Breems, "A 1.9 mW 250 MHz bandwidth continuous-time ΣΔ modulator for ultra-wideband applications," in *IEEE International Symposium on Circuits* and Systems, 2018.
- [36] S. Loeda, H. Reekie, and B. Mulgrew, "On the design of high-performance wideband continuous-time sigma-delta converters using numerical optimization," *IEEE Transactions on Circuits and Systems I: Regular Papers*, vol. 53, no. 4, pp. 802–810, 2006.
- [37] R. Schreier, "Delta Sigma toolbox, MATLAB central file exchange," 2021.
- [38] R. van de Plassche, CMOS Integrated Analog-to-Digital and Digital-to-Analog Converters. Springer, 2003.
- [39] W. L. Lee, "A novel higher order interpolative modulator topology for high resolution oversampling A/D converters," Master's thesis, Massachusetts Institute of Technology (MIT), 1987.
- [40] C.-Y. Ho, C. Liu, C.-L. Lo, H.-C. Tsai, T.-C. Wang, and Y.-H. Lin, "A 4.5mW CT self-coupled ΔΣ modulator with 2.2MHz BW and 90.4dB SNDR using residual ELD compensation," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2015, pp. 274–275.
- [41] A. M. Thurston and M. J. Hawksford, "Dynamic overload recover mechanism for Sigma Delta modulators," in Second International Conference on Advanced A-D and D-A Conversion Techniques and their Applications, 1994, pp. 124–129.
- [42] L. J. Breems, R. Rutten, and G. Wetzker, "A cascaded continuous-time ΣΔ modulator with 67-dB dynamic range in 10-MHz bandwidth," *IEEE J. Solid-State Circuits*, vol. 39, no. 12, pp. 2152–2160, 2004.

- [43] N. Maghari, S. Kwon, G. C. Temes, and U. Moon, "Sturdy MASH ΔΣ modulator," *Electronics Letters*, vol. 42, no. 22, pp. 1269–1270, 2006.
- [44] N. Maghari, S. Kwon, and U.-K. Moon, "74 dB SNDR multi-loop sturdy-MASH delta-sigma modulator using 35 dB open-loop opamp gain," *IEEE J. Solid-State Circuits*, vol. 44, no. 8, pp. 2212–2221, 2009.
- [45] D.-Y. Yoon, S. Ho, and H.-S. Lee, "A continuous-time sturdy-mash ΣΔmodulator in 28 nm cmos," *IEEE J. Solid-State Circuits*, vol. 50, no. 12, pp. 2880 – 2890, 2015.
- [46] L. Qi, A. Jain, D. Jiang, S.-W. Sin, R. P. Martins, and M. Ortmanns, "A 76.6-dB-SNDR 50-MHz-BW 29.2-mW multi-bit CT sturdy MASH with DAC non-linearity tolerance," *IEEE J. Solid-State Circuits*, vol. 55, no. 2, pp. 344– 355, 2020.
- [47] J. Harrison, M. Nesselroth, R. Mamuad, A. Behzad, A. Adams, and S. Avery, "An LC bandpass ΔΣ ADC with 70db SNDR over 20MHz bandwidth using CMOS DACs," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2012, pp. 146–147.
- [48] H. Chae and M. P. Flynn, "A 69dB SNDR, 25MHz BW, 800MS/s continuoustime bandpass $\Delta\Sigma$ ADC using DAC duty cycle control for low power and reconfigurability," in *IEEE Symposium on VLSI Circuits*, 2013, pp. C62–C63.
- [49] C.-Y. Wang, J.-H. Tsai, S.-Y. Su, J.-C. Tsai, J.-R. Chen, and C.-H. Lou, "An 80MHz-BW 31.9fJ/conv-step filtering ΔΣ ADC with a built-in DACsegmentation/ELD-compensation 6b 960MS/s SAR-quantizer in 28nm LP for 802.11ax applications," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Pa*pers, 2019, pp. 338–339.
- [50] T. Kim, C. Han, and N. Maghari, "An 11.4mW 80.4dB-SNDR 15MHz-BW CT Delta-Sigma modulator using 6b double-noise-shaped quantizer," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2017, pp. 468–469.
- [51] V. Dhanasekaran, M. Gambhir, M. Elsayed, E. Sanchez-Sinencio, J. Silva-Martinez, C. Mishra, L. Chen, and E. Pankratz, "A 20MHz BW 68dB DR CT ΔΣ ADC based on a multi-bit time-domain quantizer and feedback element," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2009, pp. 174–175.
- [52] W. Wang, C.-H. Chan, Y. Zhu, and R. P. Martins, "A 72.6dB-SNDR 100MHz-BW 16.36mW CTDSM with preliminary sampling and quantization scheme in backend subranging QTZ," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2019, pp. 340–341.

- [53] K. Matsukawa, Y. Mitani, M. Takayama, K. Obata, S. Dosho, and A. Matsuzawa, "A fifth-order continuous-time delta-sigma modulator with singleopamp resonator," *IEEE J. Solid-State Circuits*, vol. 45, no. 4, pp. 697–706, 2010.
- [54] H. Chae, J. Jeong, G. Manganaro, and M. Flynn, "A 12mw low-power continuous-time bandpass ΔΣ modulator with 58dB SNDR and 24MHz bandwidth at 200MHz IF," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2012, pp. 148–149.
- [55] H. Chae, J. Jeong, G. Manganaro, and M. P. Flynn, "A 12 mW low power continuous-time bandpass ΔΣ modulator with 58 dB SNDR and 24 MHz bandwidth at 200 MHz IF," *IEEE J. Solid-State Circuits*, vol. 49, no. 2, pp. 405–415, 2014.
- [56] M. Ortmanns, F. Gerfers, and Y. Manoli, "Compensation of finite gainbandwidth induced errors in continuous-time sigma-delta modulators," *IEEE Trans. Circuits Syst. I: Reg. Papers*, vol. 51, no. 6, pp. 1088–1099, 2004.
- [57] S. Pavan, "Systematic design centering of continuous time oversampling converters," *IEEE Trans. Circuits Syst. II: Express Briefs*, vol. 57, no. 3, pp. 158– 162, 2010.
- [58] M. Bolatkale, L. J. Breems, and K. A. A. Makinwa, *High Speed and Wide Bandwidth Delta-Sigma ADCs*. Springer International Publishing, 2014.
- [59] B. Razavi, Design of analog CMOS integrated circuits. New York, NY: McGraw-Hill Education, 2017.
- [60] K. H. Lundberg, "Noise sources in bulk CMOS," 2002. [Online]. Available: http://www.mit.edu/people/klund/papers/UNP_noise.pdf
- [61] J. A. Cherry and W. M. Snelgrove, "Clock jitter and quantizer metastability in continuous-time Delta-Sigma modulators," *IEEE Trans. Circuits Syst. II: Analog Digit. Signal Process.*, vol. 46, no. 6, pp. 661–676, 1999.
- [62] S. Hashemi and B. Razavi, "Analysis of metastability in pipelined ADCs," *IEEE J. Solid-State Circuits*, vol. 49, no. 5, pp. 1198–1209, 2014.
- [63] S.-H. Wu, T.-K. Kao, Z.-M. Lee, P. Chen, and J.-Y. Tsai, "A 160MHz-BW 72dB-DR 40mW continuous-time ΔΣ modulator in 16nm CMOS with analog ISI-reduction technique," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2016, pp. 280–281.
- [64] Y. Tang, J. Briaire, K. Doris, R. van Veldhoven, P. C. W. van Beek, H. J. A. Hegt, and A. H. M. van Roermund, "A 14 bit 200 MS/s DAC with SFDR >78 dBc, IM3 < -83 dBc and NSD <-163 dBm/Hz across the whole nyquist band enabled by dynamic-mismatch mapping," *IEEE J. Solid-State Circuits*, vol. 46, no. 6, pp. 1371–1381, 2011.

- [65] J. G. Kauffman, P. Witte, J. Becker, and M. Ortmanns, "An 8.5 mW continuous-time ΔΣ modulator with 25 MHz bandwidth using digital background DAC linearization to achieve 63.5 dB SNDR and 81 dB SFDR," *IEEE J. Solid-State Circuits*, vol. 46, no. 12, pp. 2869–2881, 2011.
- [66] J. G. Kauffman, P. Witte, M. Lehmann, J. Becker, Y. Manoli, and M. Ortmanns, "A 72 dB DR, CT ΔΣ modulator using digitally estimated, auxiliary DAC linearization achieving 88 fJ/conv-step in a 25 MHz BW," *IEEE J. Solid-State Circuits*, vol. 49, no. 2, pp. 392–404, 2014.
- [67] Y.-S. Shu, J.-Y. Tsai, P. Chen, T.-Y. Lo, and P.-C. Chiu, "A 28fJ/conv-step CT ΔΣ modulator with 78dB DR and 18MHz BW in 28nm CMOS using a highly digital multibit quantizer," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2013, pp. 268–269.
- [68] P. Shettigar and S. Pavan, "Design techniques for wideband single-bit continuous-time $\Delta\Sigma$ modulators with FIR feedback DACs," *IEEE J. Solid-State Circuits*, vol. 47, no. 12, pp. 2865–2879, 2012.
- [69] J. A. Cherry and W. M. Snelgrove, "Excess loop delay in continuous-time Delta-Sigma modulators," *IEEE Trans. Circuits Syst. II: Analog Digit. Signal Pro*cess., vol. 46, no. 4, pp. 376–389, 1999.
- [70] R. H. M. van Veldhoven and A. H. M. van Roermund, Robust Sigma Delta Converters. Springer-Verlag GmbH, 2011.
- [71] M. Pelgrom, A. Duinmaijer, and A. Welbers, "Matching properties of MOS transistors," *IEEE J. Solid-State Circuits*, vol. 24, no. 5, pp. 1433–1439, 1989.
- [72] M. Ortmanns, F. Gerfers, and Y. Manoli, "A case study on a 2-1-1 cascaded continuous-time sigma-delta modulator," *IEEE Trans. Circuits Syst. I: Reg. Papers*, vol. 52, no. 8, pp. 1515–1525, 2005.
- [73] Y.-S. Shu, J. Kamiishi, K. Tomioka, K. Hamashita, and B.-S. Song, "LMSbased noise leakage calibration of cascaded continuous-time ΔΣ modulators," *IEEE J. Solid-State Circuits*, vol. 45, no. 2, pp. 368–379, 2010.
- [74] H. Banba, H. Shiga, A. Umezawa, T. Miyaba, T. Tanzawa, S. Atsumi, and K. Sakui, "A CMOS bandgap reference circuit with sub-1-V operation," *IEEE J. Solid-State Circuits*, vol. 34, no. 5, pp. 670–674, 1999.
- [75] N. Sarhangnejad, "High-precision $\Sigma\Delta$ modulator with improved linearity," Master's thesis, Delft University of Technology, 2010.
- [76] M. Gustavsson, J. J. Wikner, and N. N. Tan, CMOS Data Converters for Communications. Springer, New York, NY, 2002.
- [77] M. Keller, A. Buhmann, J. Sauerbrey, M. Ortmanns, and Y. Manoli, "A comparative study on excess-loop-delay compensation techniques for continuoustime Sigma–Delta modulators," *IEEE Trans. Circuits Syst. I: Reg. Papers*, vol. 55, no. 11, pp. 3480–3487, 2008.
- [78] L. Risbo, " $\Sigma\Delta$ modulators stability and design optimization," Ph.D. dissertation, Technical University Denmark, 1994.
- [79] C. Zhang, L. J. Breems, G. Radulov, M. Bolatkale, H. Hegt, and A. van Roermund, "A digital calibration technique for wide-band CT MASH ΣΔ ADCs with relaxed filter requirements," in *Proc. IEEE Int. Symp. Circuits and Systems (ISCAS)*, 2016, pp. 1486–1489.
- [80] S. Kanagaraj and B. J. Farahani, "An online fully-digital calibration of leakage noise in MASH continuous time ΔΣ modulators," in *Proc. IEEE Int. Symp. Circuits and Systems (ISCAS)*, 2011, pp. 1632–1635.
- [81] C. Zhang, L. J. Breems, G. Radulov, M. Bolatkale, Q. Liu, H. Hegt, and A. van Roermund, "Current-mode multi-path excess loop delay compensation for GHz sampling CT ΣΔ ADCs," in *Proc. IEEE Int. Symp. Circuits and Systems (IS-CAS)*, 2017, pp. 547–550.
- [82] L. J. Breems, M. Bolatkale, and C. Zhang, "Sigma Delta modulator, integrated circuit and method therefor," U.S. Patent 10 439 633, Oct., 2019, granted.
- [83] Y. Zhang, C.-H. Chen, T. He, and G. C. Temes, "A continuous-time Delta-Sigma modulator for biomedical ultrasound beamformer using digital ELD compensation and FIR feedback," *IEEE Trans. Circuits Syst. I: Reg. Papers*, vol. 62, no. 7, pp. 1689–1698, 2015.
- [84] A. Jain and S. Pavan, "Continuous-time Delta-Sigma modulators with timeinterleaved FIR feedback," *IEEE Trans. Circuits Syst. I: Reg. Papers*, 2017.
- [85] Y. Dong, W. Yang, R. Schreier, A. Sheikholeslami, and S. Korrapati, "A continuous-time 0–3 MASH ADC achieving 88 dB DR with 53 MHz BW in 28 nm CMOS," *IEEE J. Solid-State Circuits*, vol. 49, no. 12, pp. 2868–2877, 2014.
- [86] V. Srinivasan, V. Wang, P. Satarzadeh, B. Haroun, and M. Corsi, "A 20mW 61dB SNDR (60MHz BW) 1b 3rd-order continuous-time Delta-Sigma modulator clocked at 6GHz in 45nm CMOS," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2012, pp. 158–159.
- [87] S. Patón, A. Di Giandomenico, L. Hernández, A. Wiesbauer, T. Pötscher, and M. Clara, "A 70-mW 300-MHz CMOS continuous-time ΣΔ ADC with 15-MHz bandwidth and 11 bits of resolution," *IEEE J. Solid-State Circuits*, vol. 39, no. 7, pp. 1056–1063, 2004.

- [88] M. B. Dayanik, D. Weyer, and M. P. Flynn, "A 5GS/s 156MHz BW 70dB DR continuous-time sigma-delta modulator with time-interleaved reference dataweighted averaging," in *Proc. IEEE Symp. VLSI Circuits*, 2017, pp. C38–C39.
- [89] F. Jiang, C. Chen, Y. Xiao, J. Xu, and J. Ren, "Beyond-one-cycle loop delay CT ΔΣ modulators with proper rational NTF synthesis and time-interleaved quantizers," in *Proc. IEEE Int. Midwest Symp. Circuits and Systems (MWSCAS)*, 2013, pp. 558–561.
- [90] G. Radulov, P. J. Quinn, H. Hegt, and A. H. van Roermund, Smart and flexible digital-to-analog converters. Springer, 2011.
- [91] W. Schofield, D. Mercer, and L. S. Onge, "A 16b 400MS/s DAC with <-80dBc IMD to 300MHz and <-160dBm/Hz noise power spectral density," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2003.
- [92] K. Doris, A. H. M. van Roermund, C. Lin, and D. M. Leenaerts, "Error optimization in digital to analog conversion," World Intellectual Property Organisation (WIPO) Patent WO 03/021 790, 2003.
- [93] A. Yasuda, H. Tanimoto, and T. Iida, "A third-order Δ-Σ modulator using second-order noise-shaping dynamic element matching," *IEEE J. Solid-State Circuits*, vol. 33, no. 12, pp. 1879–1886, 1998.
- [94] M. Vadipour, "Techniques for preventing tonal behavior of data weighted averaging algorithm in Σ-Δ modulators," *IEEE Trans. Circuits Syst. II: Analog Digit. Signal Process.*, vol. 47, no. 11, pp. 1137–1144, 2000.
- [95] R. Radke, A. Eshraghi, and T. Fiez, "A 14-bit current-mode ΣΔ DAC based upon rotated data weighted averaging," *IEEE J. Solid-State Circuits*, vol. 35, no. 8, pp. 1074–1084, 2000.
- [96] Y. Geerts, M. Steyaert, and W. Sansen, "A high-performance multibit ΔΣ CMOS ADC," *IEEE J. Solid-State Circuits*, vol. 35, no. 12, pp. 1829–1840, 2000.
- [97] W. Yang, W. Schofield, H. Shibata, S. Korrapati, A. Shaikh, N. Abaskharoun, and D. Ribner, "A 100mW 10MHz-BW CT ΔΣ modulator with 87dB DR and 91dBc IMD," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2008, pp. 498–499.
- [98] C. Zhang, L. J. Breems, and M. Bolatkale, "Metastability shaping technique for continuous-time Sigma-Delta analog-to-digital converters," U.S. Patent 10763 888, Sep., 2020, granted.
- [99] J. de la Rosa, Sigma-Delta Converters: Practical Design Guide. Wiley-IEEE Press, 2018.

- [100] S. Xue, "Metastability error compensation system for high-speed continuous time $\Sigma\Delta$ ADCs," Master's thesis, Eindhoven university of technology, 2017.
- [101] Y. Zhang, "Metastability error compensation system for high-speed continuous time $\Sigma\Delta$ ADCs," Master's thesis, Eindhoven university of technology, 2016.
- [102] Y. Chae and G. Han, "Low voltage, low power, inverter-based switchedcapacitor Delta-Sigma modulator," *IEEE J. Solid-State Circuits*, vol. 44, no. 2, pp. 458–472, 2009.
- [103] M. Figueiredo, R. Santos-Tavares, E. Santin, J. Ferreira, G. Evans, and J. Goes, "A two-stage fully differential inverter-based self-biased CMOS amplifier with high efficiency," *IEEE Trans. Circuits Syst. I: Reg. Papers*, vol. 58, no. 7, pp. 1591–1603, 2011.
- [104] S. Zeller, C. Muenker, R. Weigel, and T. Ussmueller, "A 0.039 mm² inverterbased 1.82 mW 68.6 dB-SNDR 10 MHz-BW CT-ΣΔ-ADC in 65 nm CMOS using power- and area-efficient design techniques," *IEEE J. Solid-State Circuits*, vol. 49, no. 7, pp. 1548–1560, 2014.
- [105] L. J. Breems, J. van Sinderen, T. Fric, H. Stoffels, F. Fritschij, H. Brekelmans, H. van der Ploeg, U. Moehlmann, R. Rutten, M. Bolatkale, S. Bajoria, J. Niehof, B. Oude-Essink, and G. Lassche, "A full-band multi-standard global analog & digital car radio SoC with a single fixed-frequency PLL," in *IEEE Radio Frequency Integrated Circuits Symposium (RFIC)*, 2019, pp. 315–318.
- [106] J. van Sinderen, G. W. de Jong, F. Leong, X. He, M. Apostolidou, H. K. Subramaniyan, R. Rutten, J. Niehof, J. Verlinden, H. Wang, A. Hoogstraate, K. C. Kwok, R. Verlinden, R. Hoogendoorn, D. Jeurissen, A. Salfelner, E. Bergler, J. M. V. Torres, C. J. Haji-Michael, T. Unterweger, E. Tarvainen, M. Posch, R. Schmidt, M. Stattmann, J. Tyminski, P. Jean, S. Darfeuille, O. Aymard, A. le Grontec, C. Boucey, C. Kelma, and G. Monnerie, "Wideband UHF ISMband transceiver supporting multichannel reception and DSSS modulation," in *IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers*, 2013, pp. 454–455.
- [107] R. Eschauzier, L. Kerklaan, and J. Huijsing, "A 100-MHz 100-dB operational amplifier with multipath nested Miller compensation structure," *IEEE J. Solid-State Circuits*, vol. 27, no. 12, pp. 1709–1717, 1992.
- [108] C. Ndiaye, V. Huard, R. Bertholon, M. Rafik, X. Federspiel, and A. Bravaix, "Layout dependent effect: Impact on device performance and reliability in recent CMOS nodes," in *IEEE International Integrated Reliability Workshop* (*IIRW*), 2016, pp. 24–28.
- [109] "E8257D PSG sheet," microwave analog signal generator data Keysight Technologies, Tech. Rep., 2021.[Online]. Available: https://www.keysight.com/us/en/assets/7018-01211/data-sheets/5989-0698.pdf

- [110] Q. Liu, L. J. Breems, S. Bajoria, M. Bolatkale, C. Zhang, and G. Radulov, "Analysis of the inter-stage signal leakage in wide BW low OSR and high DR CT MASH $\Delta\Sigma$ M," in *Proc. IEEE Int. Symp. Circuits and Systems (ISCAS)*, 2020.
- [111] C. Briseno-Vidrios, A. Edward, A. Shafik, S. Palermo, and J. Silva-Martinez, "A 75-MHz continuous-time Sigma-Delta modulator employing a broadband low-power highly efficient common-gate summing stage," *IEEE J. Solid-State Circuits*, vol. 52, no. 3, pp. 657–668, 2017.
- [112] A. Edward, Q. Liu, C. Briseno-Vidrios, M. Kinyua, E. G. Soenen, A. I. Karşılayan, and J. Silva-Martinez, "A 43-mW MASH 2-2 CT ΣΔ modulator attaining 74.4/75.8/76.8 dB of SNDR/SNR/DR and 50 MHz of BW in 40-nm CMOS," *IEEE J. Solid-State Circuits*, vol. 52, no. 2, pp. 448–459, 2017.
- [113] X. Xing and G. G. E. Gielen, "A 42 fJ/step-FoM two-step VCO-based Delta-Sigma ADC in 40 nm CMOS," *IEEE J. Solid-State Circuits*, vol. 50, no. 3, pp. 714–723, 2015.

List of publications

Conference proceedings

- C. Zhang, L.J. Breems, G.I. Radulov, M. Bolatkale, J.A. Hegt and A.H.M. van Roermund, "A digital calibration technique for wide-band CT MASH ΣΔ ADCs with relaxed filter requirements," in *IEEE International Symposium on Circuits* and Systems, Montreal, Canada, May 22 – 25, 2016, pp. 1486 – 1489.
- C. Zhang, L.J. Breems, G.I. Radulov, M. Bolatkale, Q. Liu, J.A. Hegt and A.H.M. van Roermund, "Current-mode multi-path excess loop delay compensation for GHz sampling CT ΣΔ ADCs," in *IEEE International Symposium on Circuits and Systems*, Baltimore, MD, USA, May 28 – 31, 2017.
- M. Zhou, M. Neofytou, M. Bolatkale, Q. Liu, C. Zhang, P. Cenci, G.I. Radulov, P. Baltus and L.J. Breems, "A 2 GHz 0.98 mW 4-bit SAR-based quantizer with ELD compensation in an UWB CT ΣΔ modulator," in *IEEE International* Symposium on Circuits and Systems, Florence, Italy, May 27 – 30, 2018.
- 4. M. Neofytou, M. Zhou, M. Bolatkale, Q. Liu, C. Zhang, G.I. Radulov, P. Baltus and L.J. Breems, "A 1.9 mW 250 MHz bandwidth continuous-time $\Sigma\Delta$ modulator for ultra-wideband applications," in *IEEE International Symposium* on Circuits and Systems, Florence, Italy, May 27 – 30, 2018.
- S. Ringeling, L. Steinebach, Q. Liu, C. Zhang, S. Bajoria, M. Bolatkale, L. Breems, G. Radulov, "Novel baseband analog beamforming through resistive DACs and Sigma Delta modulators," in *IEEE European Conference on Circuit Theory and Design*, Sofia, Bulgaria, September 7 – 10, 2020.
- 6. Q. Liu, L.J. Breems, S. Bajoria, M. Bolatkale, C. Zhang and G.I. Radulov, "Analysis of the inter-stage signal leakage in wide BW low OSR and high DR CT MASH $\Delta\Sigma$ M," in *IEEE International Symposium on Circuits and Systems*, Seville, Spain, October 11 – 14, 2020.
- 7. C. Zhang, L.J. Breems, Q. Liu, G.I. Radulov, M. Bolatkale, S. Bajoria, R. Rutten and A.H.M. van Roermund, "A 6GS/s 0.5GHz BW continuous-time 2-1-1 MASH $\Delta\Sigma$ modulator with phase-boosted current-mode ELD compensation

in 40nm CMOS," accepted at *IEEE European Solid-State Circuits Conference*, Grenoble, France, September 6 - 9, 2021.

Patents

- L.J. Breems, M. Bolatkale and C. Zhang, "Sigma Delta modulator, integrated circuit and method therefor" *patent US10439633* (granted on 08-10-2019) and *patent EP3407500* (published on 28-11-2018).
- C. Zhang, L.J. Breems and M. Bolatkale, "Metastability shaping technique for continuous-time Sigma-Delta analog-to-digital converters" *patent US10763888* (granted on 01-09-2020), *patent EP3736988* (published on 11-11-2020), and *patent CN111917417* (published on 10-11-2020).
- C. Zhang, L.J. Breems and M. Bolatkale, "A Sigma Delta modulator, integrated circuit and method therefor" *patent US2021/0126648* (published on 29-04-2021), *patent EP3817234* (published on 05-05-2021), and *patent CN112737594* (published on 30-04-2021).
- 11. C. Zhang, L.J. Breems and S. Bajoria, "Techniques for linearizing digital-toanalog converters in Sigma-Delta analog-to-digital converters" US patent filed.
- 12. C. Zhang, L.J. Breems and M. Bolatkale, "Techniques for high-speed excess loop delay compensation in Sigma-Delta analog-to-digital converters" *US patent* filed.
- 13. C. Zhang, M. Ganzerli, P. Cenci and L.J. Breems, "A Sigma Delta modulator and method therefor" *CN patent* filed.

Summary Broadband Continuous-Time MASH Sigma-Delta ADCs

This PhD dissertation describes a study to the design of broadband continuous-time MASH $\Sigma\Delta$ ADCs. Architecture choices and implementation challenges are addressed, and based on that a test chip is designed for a CT 2-1-1 MASH $\Sigma\Delta$ modulator with 3-bit quantization. Several new concepts are proposed in this dissertation, including a digital calibration of the mismatch between the analog loop filter and the digital noise cancellation filter; a current-mode multi-path ELD compensation; a current-mode locally-time-interleaved quantizer; a comparator offset calibration; a metastability error compensation technique; and a metastability shaping technique. Several of these new concepts have been validated with a 6 GS/s 0.5 GHz bandwidth continuous-time 2-1-1 MASH 3b $\Sigma\Delta$ modulator test chip in 40 nm CMOS. The MASH ADC test chip achieves 65 dB DR in 300 MHz BW when sampled at 4 GHz, and 58 dB DR in 500 MHz BW when sampled at 6 GHz.

Samenvatting

Deze thesis beschrijft een onderzoek naar het ontwerpen van breedbandige tijdcontinue MASH $\Sigma\Delta$ ADC's. Architectuurkeuzes en implementatieuitdagingen worden behandeld en uitgaande daarvan wordt een testchip ontworpen voor een tijdcontinue 2-1-1 MASH $\Sigma\Delta$ -modulator. Diverse nieuwe concepten worden in deze dissertatie voorgesteld, waaronder digitale calibratie van de afwijkingen tussen het analoge lusfilter en het digitale ruisonderdrukkingsfilter; een stroom-gebaseerde meerpads ELDcompensatie; een kwantisator die op stroomsturing en lokale tijd-multiplexing is gebaseerd; een offset-calibratie voor de comparator; een techniek om metastabiliteitsfouten te compenseren; en een methode om metastabiliteitsfouten buiten de signaalband te schuiven. Een aantal van die concepten zijn gevalideerd met een 6 GS/s 0.5 GHz bandbreedte tijdcontinue 2-1-1 MASH 3b $\Sigma\Delta$ -modulator-testchip in 40 nm CMOS. De MASH ADC testchip realiseert 65 dB DR in 300 MHz bandbreedte wanneer hij met 4GHz geklokt wordt, en 58 dB DR in 500 MHz BW wanneer hij met 6GHz geklokt wordt.

Acknowledgements

It has been a long journey since I started my PhD project in May 2013, which is longer than most of the other PhDs. Now when I look back, I am very grateful to all the people who helped me in this journey.

Firstly I would like to thank my promotors, supervisors and co-workers. Without their help and support, this work cannot be achieved. Thank you, Lucien! It is you who taught me how to design a Sigma Delta ADC. Thank you, Arthur! From you I have learned not only about how to do a good research work, but also about the altitude to science and technology, and the high-level view and insight. Thank you, Georgi! You have given me detailed and unreserved suggestions and support during my project. Thank you, Muhammed and Shagun! When I had technical problems in the project, you shared hands-on experience and helped me in finding the solutions. Thank you, Qilong! You helped me with some design work of the test chip, and we spent days, evenings, and weekends together to discuss, debug, brainstorm, and try solutions. I would like to thank Robert Rutten for his help in digital circuit design and test chip validation. I would like to thank Hans Hegt for his help and support in the beginning of my project and for his encouragement every time he met me even after his retirement. I also would like to thank Gijs Meuleman for his cooperation in the beginning of my project.

Next, I would like to thank my colleagues from the Integrated Circuits group (previously Mixed-signal Microelectronics group) for their help and support in this work. I would like to thank Peter Baltus for his technical and administrative support. I would like to thank Margot van den Heuvel for her help and support as the secretary of the group. I would like to thank Pieter Harpe for the valuable technical discussions and for his suggestions. I would like to thank Eugenio Cantatore for his efforts as a member in my doctorate committee. I would like to thank Marion Matters-Kammerer for her willingness to be a back-up member in my doctorate committee. I would like to thank all the other professors and (former) colleagues for their help and support in this work: Hao Gao, Pavlos Athanasiadis, Zhe Chen, Domine Leenaerts, Hanyue Li, Xiao Liu, Carlos Mendes da Costa Junior, Dusan Milosevic, Debasish Mitra, Marios Neofytou, Kevin Pelzers, Daan Rosenmuller, Yuting Shen, Zhe Song, Loek Steinebach, Vojkan Vidojkovic, Bindi Wang, Heinrich Wörtche, Haoming Xin, Kuangyuan Ying, Yijing Zhang, Meiyi Zhou, Maoqiang Liu, Shuang Song, Qiang Ma, Xi Zhang, Chuang Lv, Xiangrong Zhang, Jun Wang, Song Ma, Lammert Duipmans, Juan Camilo Castellanos Rodriguez, Elbert Bechthum, Sahel Abdinia, Juan Alfaro, Jing Li, Jingjing Hu.

Moreover, I would like to thank my managers and (former) colleagues from NXP Semiconductors for their help and support in this work. I would like to thank Marcel Arkesteijn, Kostas Doris, Willem Jan Wagenaar, Paul Swinkels, and Andrew Turley for their management and administrative support. I would like to thank Ericson James De Los Santos, Cristine Badiao and Win Love Pili for their layout support. I would like to thank Pieter Oude Egberink for his administrative support in the test chip tape-out. I would like to thank Peter Kersten for his help in the design of the validation board. I would like to thank Con Verheven for his support in the digital back-end design. I would like to thank Wei Kong for his help in the test chip validation. I would like to thank Yu Lin, Marcello Ganzerli, Hans Brekelmans, Pierluigi Cenci, Jan Niehof, Erwin E. Janssen, and Vladislav Dyachenko for the technical discussions and support. I would like to thank all the other (former) colleagues who helped me during my PhD project: Peter Prieshof, Yihan Gao, Athon Zanikopoulos, Dolphin Abessolo Bidzo, Anton de Graauw, Erwin Janssen, Maarten Lont, Waqas Syed, Yuan Gao, Zhirui Zong, Hao Wang, Yuanhao Bi, Wendi Zhang, Liang Li. I also would like to thank all of the master students (or graduates) whom I supervised during my PhD project (not yet mentioned above) for their contribution to this work: Shuang Xue, Qizhi Wu, Xiangyu Fang, Fatima Abidine, Patrick van Mourik, Remco Schalk.

I would like to thank Dutch Technology Foundation STW for funding my PhD project. I would like to thank Prof. Maurits Ortmanns and Prof. Georges Gielen for their efforts as my doctorate committee members. I would like to thank Prof. Sonia Heemstra de Groot for chairing my PhD defense ceremony.

I also would like to thank my friends (not yet mentioned above) who spent this enjoyable time with me in the past years: Yuxiang Yin, Jiangying Chai, Shaojun Wang, Chaojie Xu, Hua Ma, Meng Li, Jiao Pan, Hang Liu, Yuyang Wang, Yang Li, Ya Zhang, Chuan Nie, Yue Fan, Weilu Li, Lu Liu, Cheng Li, Chao Wang, Justin Verkuijl, Guus van Kollenburg, Lut van Kollenburg.

In the end, I would like to thank my family who love and support me. I would like to thank my parents who raised me up and encouraged me to pursue the highest academic degree since I was a primary school student. I would like to thank my wife who sacrifices her time and spends much more efforts to take care of our daughter and family, such that I can spend more time on my PhD work. I would like to thank my daughter for all the happy, moving, and special moments and time together. I would like to thank my wife's parents for their care, trust and support. And I would like to thank my wife's younger brother Shaoling and his wife Yang especially for taking care of my wife's dad when he had an accident. I LOVE YOU!

I feel very lucky and grateful for all the care, help and support I received. Thank you very much!

Biography

Chenming Zhang was born in Yongqing, Hebei Province, China in 1988. From 2006 to 2007, he studied at Tongji University, Shanghai, China. He received the Bachelor of Science degree (cum laude) in Electronic Engineering from Politecnico di Torino, Turin, Italy, in 2010, and the Master of Science degree in Electrical Engineering and Information Technology from ETH Zürich, Zürich, Switzerland, in 2013. Since 2013, he started to pursue his PhD's degree in the integrated circuits group (named mixed-signal microelectronics group at that time) at Eindhoven University of Technology, Eindhoven, The Netherlands. His PhD project is on broadband continuous-time MASH Sigma-Delta ADCs.

In 2010, he was with National Institute of Metrological Research, Turin, Italy, as an intern. From 2011 to 2012, he was with the ABB Switzerland Ltd., Semiconductors, Lenzburg, Switzerland, as an intern. Since 2018, he has been working as a wideband ADC designer in the group "Radar & Wideband Frontends", business line Radio Frequency Processing, NXP Semiconductors, Eindhoven, The Netherlands. His current research interests include high-speed broadband high-linearity continuous-time Sigma-Delta ADC design and ADC calibration techniques.

